Validation of Methane and Carbon Monoxide from Sentinel-5 Precursor using TCCON and NDACC-IRWG stations

Mahesh Kumar Sha¹, Bavo Langerock¹, Jean-François L. Blavier², Thomas Blumenstock³, Tobias Borsdorff⁴, Matthias Buschmann⁵, Angelika Dehn⁶, Martine De Mazière¹, Nicholas M. Deutscher⁷, Dietrich G. Feist^{8,9,10}, Omaira E. García¹¹, David W. T. Griffith⁷, Michel Grutter¹², James W. Hannigan¹³, Frank Hase³, Pauli Heikkinen¹⁴, Christian Hermans¹, Laura T. Iraci¹⁵, Pascal Jeseck¹⁶, Nicholas Jones⁷, Rigel Kivi¹⁴, Nicolas Kumps¹, Jochen Landgraf⁴, Alba Lorente⁴, Emmanuel Mahieu¹⁷, Maria V. Makarova¹⁸, Johan Mellqvist¹⁹, Jean-Marc Metzger²⁰, Isamu Morino²¹, Tomoo Nagahama²², Justus Notholt⁵, Hirofumi Ohyama²¹, Ivan Ortega¹³, Mathias Palm⁵, Christof Petri⁵, David F. Pollard²³, Markus Rettinger²⁴, John Robinson²³, Sébastien Roche²⁵, Coleen M. Roehl²⁶, Amelie N. Röhling³, Constantina Rousogenous²⁷, Matthias Schneider³, Kei Shiomi²⁸, Dan Smale²³, Wolfgang Stremme¹², Kimberly Strong²⁵, Ralf Sussmann²⁴, Yao Té¹⁶, Osamu Uchino²¹, Voltaire A. Velazco⁷, Corinne Vigouroux¹, Mihalis Vrekoussis^{27,5}, Pucai Wang^{29,30}, Thorsten Warneke⁵, Tyler Wizenberg²⁵, Debra Wunch²⁵, Shoma Yamanouchi²⁵, Yang Yang^{31,29,1}, and Minqiang Zhou¹

¹Royal Belgian Institute for Space Aeronomy (BIRA-IASB), Brussels, Belgium

²Jet Propulsion Laboratory, California Institute of Technology, Pasadena, California, USA

³Karlsruhe Institute of Technology, IMK-ASF, Karlsruhe, Germany

⁴SRON Netherlands Institute for Space Research, Utrecht, the Netherlands

⁵Institute of Environmental Physics, University of Bremen, Bremen, Germany

⁶European Space Agency, ESA/ESRIN

⁷Centre for Atmospheric Chemistry, School of Earth, Atmospheric and Life Sciences, University of Wollongong, Wollongong, Australia

⁸Ludwig-Maximilians-Universität München, Lehrstuhl für Physik der Atmosphäre, Munich, Germany

⁹Deutsches Zentrum für Luft- und Raumfahrt, Institut für Physik der Atmosphäre, Oberpfaffenhofen, Germany

¹⁰Max Planck Institute for Biogeochemistry, Jena, Germany

¹¹Izaña Atmospheric Research Centre (IARC), State Meteorological Agency of Spain (AEMET), Santa Cruz de Tenerife, Spain

¹²Centro de Ciencias de la Atmósfera, Universidad Nacional Autonoma de Mexico, UNAM, Mexico

¹³National Center for Atmospheric Research, Boulder, CO, USA

¹⁴Finnish Meteorological Institute, FMI, Sodankylä, Finland

¹⁵NASA Ames Research Center, Moffett Field, CA, USA

¹⁶LERMA-IPSL, Sorbonne Université, CNRS, Observatoire de Paris, PSL Université, Paris, France

¹⁷Institut d'Astrophysique et de Géophysique, Université de Liège, Liège, Belgium

¹⁸Department of Atmospheric Physics, Faculty of Physics, St. Petersburg State University, Saint Petersburg, Russia

¹⁹Earth and Space Sciences, Chalmers University of Technology, Gothenburg, Sweden

²⁰UAR 3365 – OSU Réunion, Université de La Réunion, Saint-Denis, Réunion, France

²¹National Institute for Environmental Studies (NIES), Tsukuba, Japan

²²Institute for Space-Earth Environmental Research (ISEE), Nagoya University, Japan

²³National Institute of Water and Atmospheric Research Ltd (NIWA), Lauder, New Zealand

²⁴Karlsruhe Institute of Technology, IMK-IFU, Garmisch-Partenkirchen, Germany

²⁵Department of Physics, University of Toronto, Toronto, Ontario, Canada

²⁶Division of Geological and Planetary Sciences, California Institute of Technology, Pasadena, CA, USA

²⁷Climate and Atmosphere Research Center (CARE-C), The Cyprus Institute, Nicosia, Cyprus
 ²⁸Earth Observation Research Center (EORC), Japan Aerospace Exploration Agency (JAXA), Japan
 ²⁹LAGEO, the Institute of Atmospheric Physics, Chinese Academy of Sciences, Beijing, China
 ³⁰University of Chinese Academy of Sciences, Beijing, China
 ³¹Shanghai Ecological Forecasting and Remote Sensing Center, Shanghai, China

Correspondence: Mahesh Kumar Sha (mahesh.sha@aeronomie.be)

Abstract. The Sentinel-5 Precursor (S5P) mission with the TROPOspheric Monitoring Instrument (TROPOMI) onboard has been measuring solar radiation backscattered by the Earth's atmosphere and its surface since its launch on 13 October 2017. Methane (CH₄) and carbon monoxide (CO) data with a spatial resolution (initially $7 \times 7 \text{ km}^2$, upgraded to $5.5 \times 7 \text{ km}^2$ on 6^{th} of August 2019) have been retrieved from shortwave infrared (SWIR) and near-infrared (NIR) measurements since the

- 5 end of November 2017 and made available to the experts for early validation and quality checks before the official product release. In this paper, we present for the first time the S5P operational methane (CH₄and CO-) and carbon monoxide (CO) products validation results (covering a period from November 2017 to September 2020, see Table 1 for version details) of about 3 years using global Total Carbon Column Observing Network (TCCON) and Infrared Working Group of the Network for the Detection of Atmospheric Composition Change (NDACC-IRWG) network data, accounting for a priori alignment and
- 10 smoothing uncertainties in the validation, and testing the sensitivity of validation results towards the application of advanced co-location criteria.

We found that the S5P standard and bias-corrected methane CH_4 data over land surface for the recommended quality filtering fulfil the mission requirements of a bias (systematic error) less than 1.5% and a random error less than 1%. The systematic difference between the S5P standard of the bias-corrected XCH₄ and data with respect to TCCON data is on average

-0.68-0.26±0.740.56%. The systematic difference changes to a value of -0.26in comparison to -0.68±0.560.74% for the S5P bias-corrected standard XCH₄ data. We found, with a correlation of above 0.6 for most stations, which is mostly dominated by the seasonal cycle. The contribution of uncertainty due to smoothing at the individual stations was estimated and found to be dependent on the location. The highest contribution of the smoothing uncertainty is observed for mid-latitude TCCON stations and high latitude stations for NDACC. A seasonal dependency of the relative bias is seen. We observe a high bias during the springtime measurements at high SZA and a decreasing bias with increasing SZA for the rest of the year.

. The bias shows a seasonal dependence. We found that the S5P earbon monoxide CO data over all surfaces for the recommended quality filtering in general fulfil the mission requirements of a bias (systematic error) less than 15% and a random error of <10%. There are a few stations where this is not the case, generally fulfil the missions requirements, with a few exceptions, which are mostly due to co-location mismatches and the limited availability of eo-located data. We compared the S5P XCO

25 data with respect to standard TCCON XCO and unscaled TCCON XCO (without application of the empirical scaling factor) data sets. data. The systematic difference between between the S5P XCO and the TCCON data is on average 9.22±3.45 % (standard TCCON XCOdata) and 2.45±3.38 % (unscaled TCCON XCOdata). We found that the systematic difference between the S5P CO column and NDACC CO column data (excluding two stations that were obvious outliersoutlier stations) is on average 6.5±3.54 %. We found a correlation of above 0.9 for most TCCON and NDACC stationsindicating that the temporal

- 30 variations in CO column captured by the ground-based instruments are reproduced very similarly by the S5P CO column. The contribution of smoothing uncertainty at the individual stations is estimated and found to be significant. They are found to be dependent on the location with large changes seen for stations located in the Southern Hemisphere as compared to the Northern Hemisphere and at highly polluted stations. A cone co-location criterion, which gives a better match between the ground-based instrument's line-of-sight and satellite pixels, seems to give better results for high latitude stations and stations located close
- 35 to emission sources. The validation results for the clear-sky and cloud cases of S5P pixels are in general comparable to the validation results including all pixels with recommended quality filtering. We observe that the relative bias increases with increasing SZA. We estimated this increase is about 10% over the complete range of measurement SZAs.

 \therefore The study shows the high quality of S5P CH₄ and CO data by validating the products against reference global TCCON and NDACC stations covering a wide range of latitudinal bands, atmospheric conditions, and surface conditions.

40 1 Introduction

The Sentinel-5 Precursor (S5P) mission with the TROPOspheric Monitoring Instrument (TROPOMI) onboard was launched on 13 October 2017. The S5P is orbiting in a Sun-synchronous polar orbit with an equator crossing at 13:30 local solar time. The TROPOMI instrument is a nadir-viewing hyperspectral spectrometer measuring solar radiation reflected by the Earth's atmosphere and its surface from the ultraviolet-visible (270–495 nm), near-infrared (675–775 nm) and shortwave-infrared (2305-2385 nm) with daily global coverage for monitoring atmospheric trace gases and aerosol (Veefkind et al., 2012). Methane (CH₄) and carbon monoxide (CO) are retrieved from shortwave-infrared (SWIR) and near-infrared (NIR) measurements.

Methane is the second most important anthropogenic greenhouse gas (GHG) after carbon dioxide (CO₂). It has a global warming potential of about 28 times larger than CO₂ over a 100 year time period. It is less abundant in the atmosphere and has a significantly shorter lifetime than CO₂ (Stocker et al., 2013). Reduction in CH₄ will affect the Earth's radiation budget on a short time scale. CH₄ is also relevant in atmospheric chemistry, where it reacts with hydroxyl radicals (OH), thereby reducing

50

45

the oxidation capacity of the atmosphere and producing ozone (Kirschke et al., 2013). Carbon monoxide is a poisonous reactive gas considered principally an anthropogenic atmospheric pollutant. Volatile organic

compounds (VOCs) are emitted to the atmosphere by incomplete combustion (e.g., vehicles, industry and biomass burning) and have an important role in the production of CO. The lifetime of CO is relatively short and ranges from weeks to months

55 (Novelli et al., 1998). CO reacts with atmospheric oxidants, ozone (O_3), hydroperoxy (HO_2), and hydroxyl radicals (OH). It is the largest direct sink of OH affecting the self-cleansing capacity of the atmosphere. An increase in CO would imply a higher OH loss through chemical reaction and therefore less availability of OH for the depletion of other atmospheric constituents such as CH₄. CO is therefore affecting the concentrations of primary greenhouse gases and has an indirect but important influence in determining the chemical composition and radiative properties of the atmosphere. It is therefore considered as an indirect

60 greenhouse gas (Stocker et al., 2013).

Continuous precise and accurate global measurements of these gases are very important for long-term monitoring and their use by the inverse models such that the inferred surface fluxes can be better constrained. This paper focuses on the quality assessment of the operational S5P CH_4 and CO products by performing validation of the total columns of these two products with the reference data from all stations in the ground-based Total Carbon Column Observing Network (TCCON) and

65 Infrared Working Group of the Network for the Detection of Atmospheric Composition Change (NDACC-IRWG) networks. The systematic and random error requirements of the CH_4 and CO products are checked based on 2.8 years of S5P data and possible reasons are given where large deviations are observed.

The paper is organised as follows: Section 2 describes the satellite and ground-based reference data used in this study. Section 3 gives the details of the validation methodology. Section 4 gives the validation results for CH_4 and section 5 gives the validation results for CO. Section 6 summarizes our results and conclusions.

2 Data

80

In this section, we present an overview of the input data from the S5P and the reference ground-based data from the TCCON and NDACC-IRWG, herewith referred to as NDACC, which are used for the validation of the S5P operational CH_4 and CO products.

75 2.1 S5P Methane and Carbon monoxide data sets

TROPOMI is the unique payload of the ESA/Copernicus Sentinel-5 Precursor mission orbiting in a low-Earth Sun-synchronous polar orbit with a wide swath of 2600 km across track resulting in daily global coverage. The TROPOMI radiometric measurements of the Earth's radiance and solar irradiance are processed using on-ground data processor to retrieve the atmospheric abundances of ozone (O_3), nitrogen dioxide (NO_2), sulphur dioxide (SO_2), formaldehyde (HCHO), methane (CH_4), carbon monoxide (CO), as well as cloud and aerosol properties. The spatial resolution of the operational Level 2 (L2) CH_4 and CO products was originally 7 × 7 km² and was increased to 5.5 × 7 km² on 6th of August 2019.

The operational processing to retrieve the total column-averaged dry air mole fraction of methane (XCH_4) is performed by the RemoTeC-S5P algorithm. The details describing the theoretical baseline of the algorithm, the input and ancillary data needed, averaging kernel, and the output generated are described in detail in Hu et al. (2016) and Hasekamp et al. (2019). The

- use of satellite measurements for estimating sources and sinks of CH_4 strongly depends on the precision and accuracy achieved. Systematic biases or lower precision on regional or seasonal scales can jeopardise the usefulness of the satellite measurements for the estimation of source and sink estimates (Bergamaschi et al., 2007). The bias requirement for S5P XCH₄ is 1.5 % and the random error requirement is 1 % (as reported in the official ESA document ESA-EOPG-CSCOP-PL, 2017, Table 1, page 14). The current S5P CH₄ data are only processed for cloud-free measurements over land. Along with the standard CH₄ product, a
- bias-corrected CH_4 product is also made operationally available. We provide a brief summary of the CH_4 bias correction here and the details of the bias correction can be found in section 5.6 of the Algorithm Theoretical Baseline Document (ATBD) for S5P methane retrieval (Hasekamp et al., 2019). The operational S5P CH_4 product has been compared to co-located GOSAT proxy measurements. The S5P-GOSAT XCH₄ ratio shows a high correlation to the retrieved surface albedo in the SWIR. The highest correlation is for low surface albedo scenes. A posteriori bias correction has been applied to the S5P CH_4 product using

- 95 a second-order polynomial fit. The effect of the bias correction is an increase of the retrieved CH_4 for scenes with relatively low albedo conditions (e.g., forest scenes) and a decrease of CH_4 for scenes with high albedo conditions (e.g., desert scenes). In the paper, we will show the validation results of both standard and bias-corrected S5P CH_4 products. The latest product versions of S5P CH_4 data for the reprocessed (RPRO) + offline (OFFL) data from the start of the mission to 30 September 2020 are used in this work. The version numbers and the respective dates are listed in Table 1 and further details on the relevant improvements
- are given in the Product Readme File (PRF; https://sentinel.esa.int/documents/247904/3541451/Sentinel-5P-Methane-Product-Readme-File, last access 14 July 2020). The quality assurance (QA) value is provided as part of the CH₄ data product. QA >0.5 is used as recommended by the PRF to filter out the S5P CH₄ data to be used for the validation studies. This selection filters out measurements performed with surface albedo <0.02, solar zenith angle (SZA) >70°, viewing zenith angle >60°, and some other criteria as mentioned in the PRF.
- 105 The operational processing to retrieve the total column density of carbon monoxide (CO) simultaneously with interfering trace gases and effective cloud parameters (cloud height and optical thickness) is performed by the Shortwave Infrared Carbon Monoxide Retrieval (SICOR) algorithm (Landgraf et al., 2016). The details describing the theoretical baseline of the algorithm, the input and ancillary data needed, example plots of averaging kernel, and the output generated are described in details in Landgraf et al. (2018). The bias requirement for total column-averaged dry air mole fraction of carbon monoxide (XCO) is
- 110 15 % and the random error requirement is <10 % (as reported in the official ESA document ESA-EOPG-CSCOP-PL, 2017, Table 1, page 14). The CO total column L2 data products are available as the Offline (OFFL) and Near Real Time (NRTI) timeliness data products. The version numbers and the respective dates are listed in Table 1 and further details on the relevant improvements are given in the Product Readme File (PRF; https://sentinel.esa.int/documents/247904/3541451/Sentinel-5P-Carbon-Monoxide-Level-2-Product-Readme-File, last access 14 July 2020). The latest product versions of S5P CO data for</p>
- 115 the reprocessed (RPRO) + offline (OFFL) data from the start of the mission to 30 September 2020 are used in this work. The NRTI data stream delivers the CO data product within 3 hours after sensing, whereas the OFFL data are available a few days after sensing. Due to the different timeliness, the NRTI product are given in 5 min data granules, whereas the OFFL data product are given per satellite orbit. The consecutive data granules of the NRTI product show an overlap of about 12 scan lines. The NRTI processing chains employ the same algorithm as the OFFL since processor version 01.03.02 starting from orbit
- 120 number 8906 on 3rd of July 2019 (see section 9.4 of Lambert et al. (2020) for validation results showing the equivalence of S5P NRTI and OFFL CO products). More details on the two processing streams of the two data sets are given in the Algorithm Theoretical Basis Document (ATBD) (Landgraf et al., 2018). In this paper, we show the detailed validation results of the S5P OFFL CO product. Data with QA values >0.5 is used as recommended by the PRF. This selection filters out measurements performed with SZA \geq 80°, sensor zenith angle \geq 80°, two most westward pixels due to unresolved calibration issues and
- some other criteria as mentioned in the PRF. Furthermore, we also separated retrievals performed for measurements under clear-sky (CLSKY; cloud optical thickness <0.5 & cloud height <500 m, over land) and cloudy conditions (CLOUD; cloud optical thickness ≥ 0.5 & cloud height <5000 m, over land and ocean) as suggested by Borsdorff et al. (2018b). The clear-sky observations over the ocean have too low signal intensities in the SWIR and therefore cannot be used for the data interpretation. Unlike the S5P CH₄ a priori profiles which are available in the L2 files, the S5P a priori profiles for CO were downloaded from

- 130 ftp://ftp.sron.nl/pub/jochen/TROPOMI_apriori/ (last access 01 December 2020). Among the known data quality issues of the CO product, single overpasses of S5P show stripes of erroneous CO values <5% in the flight direction, probably due to calibration issues of S5P. We did not do any correction of this stripe pattern as we show the operational validation of the CO product and also because of small number of pixels <5% are affected by it. The striping effect is analysed in detail by Borsdorff et al. (2019). The effect on the TCCON validation was small. The destriping approach suggested by this work is planned to be</p>
- 135 implemented by the operational TROPOMI CO processing in the near future. Furthermore, the effect of updating the spectral cross-sections in the TROPOMI CO processing for clear-sky and cloudy conditions was analysed with ground-based FTIR measurements from 12 stations of the TCCON network (Borsdorff et al., 2019).

In addition to the operational S5P CH_4 and CO products, a scientific version of the products (TROPOMI/WFMD) using the Weighting Function Modified Differential Optical Absorption Spectroscopy (WFM-DOAS) has been developed by the Uni-

140 versity of Bremen (http://www.iup.uni-bremen.de/carbon_ghg/products/tropomi_wfmd/, last access 1 June 2021). The details of the TROPOMI/WFMD CH₄ and CO products, their validation against reference ground-based measurements and operational TROPOMI products, as well as use cases of the products to address important scientific applications can be found in (Schneising et al., 2019, 2020a, b; Vellalassery et al., 2021).

2.2 Ground-based TCCON reference data set

- 145 The Total Carbon Column Observing Network (TCCON) represents a network of ground-based Fourier transform spectrometers (FTS), of the type Bruker IFS 125HR (some long-existing sites also use Bruker 120/5HR), that records direct solar absorption spectra in the near-infrared (NIR) spectral range to retrieve accurate and precise column-averaged abundances of atmospheric constituents including CO₂, CH₄ and CO amongst other species (Wunch et al., 2011, 2015). It is the current state-of-the-art validation system for total column measurements of important greenhouse gases (GHGs) by remote sensing.
- 150 TCCON data from several stations have been used in previous studies for the validation of trace gas data products from satellite platforms such as OCO-2 (O'Dell et al., 2018; Wunch et al., 2016), GOSAT (Iwasaki et al., 2017; Kulawik et al., 2016), S5P (Sha et al., 2018a; Borsdorff et al., 2018a, 2019), MOPITT (Hedelius et al., 2019), SCIAMACHY (Borsdorff et al., 2016; Hochstaffl et al., 2018). Data from all stations (23 in the Northern Hemisphere and 5 in the Southern Hemisphere) are used in this study and are listed in Table 2. The stations cover various atmospheric conditions (humid, dry, polluted, presence of
- aerosol), various surface conditions (range of albedo, flat terrain, high altitude locations), latitudinal distribution from 80° N to 45° S. The stations at Nicosia and Xianghe are not yet officially part of TCCON but performs observations and data analysis fully compatible with TCCON guidelines. GGG2014 (the current standard TCCON retrieval code) XCH₄ systematic errors for TCCON are below 0.5 % for SZAs below 85°. The XCO errors are below 4 % and decrease with SZA (Wunch et al., 2015). The uncertainty in the scaling slope for XCO is 6 % (2 σ) (Hedelius et al., 2019). Previous studies have shown that the scaling factor
- 160 of ~7% used in GGG2014 to the the TCCON XCO measurements to the World Meteorological Organization (WMO) in situ scale is large compared to the current uncertainty in spectroscopy (Sha et al., 2018b; Hedelius et al., 2019; Zhou et al., 2019). A scaling factor of 7% provided the best scaling to the in situ data available when the scaling for GGG2014 was calculated. There is currently an ongoing effort within the TCCON community to determine whether the scaling factor is appropriate.

These results are very important to decide on the choice of spectroscopic cross-sections that should be implemented for the

165 future improved S5P CO product (Borsdorff et al., 2019). In this work, we use the official TCCON XCO product as well as an XCO product without the application of the empirical scaling factor, herewith referred to as unscaled XCO. The unscaled XCO was calculated following Eq. 2 of Wunch et al. (2015), where the TCCON data without the scaling to the WMO scale were obtained from the site PIs. The validation work is done using the standard and rapid delivery of TCCON data from the whole network. The publicly available TCCON data can be accessed via https://tccondata.org/.

170 2.3 Ground-based NDACC-IRWG reference data set

The Infrared Working Group (IRWG) of the Network for the Detection of Atmospheric Composition Change (NDACC) represents a network of high-resolution Fourier transform spectrometers that records solar absorption spectra in the mid-infrared (MIR) spectral range. It is a multi-national collection of over twenty stations distributed from pole to pole (Eureka 80° N to Arrival Heights 77.8° S). The solar absorption spectra are used to retrieve the atmospheric concentrations of a number of gaseous atmospheric components, including ozone (O₃), nitric acid (HNO₃), hydrogen chloride (HCl), hydrogen fluoride (HF), carbon monoxide (CO), nitrous oxide (N₂O), methane (CH₄), hydrogen cyanide (HCN), ethane (C₂H₆) and chlorine nitrate (ClONO₂) (https://www2.acom.ucar.edu/irwg). NDACC CH₄ and CO data from several stations have been used in previous studies for satellites validation (Borsdorff et al., 2020; Hedelius et al., 2019; Hochstaffl et al., 2018; Sha et al., 2018b; Buchholz et al., 2017; Olsen et al., 2017). In this study, data from all stations (19 in the Northern Hemisphere and 5 in the Southern

- 180 Hemisphere) are used and are listed in Table 3. Several of the stations are located in high latitude regions and many stations are located at high altitudes to reduce the interference of water vapour in the measurements. Some of these stations (e.g., Karlsruhe, Garmisch, Sodankylä, Porto Velho) are not officially part of NDACC but performs observations and data analysis fully compatible with NDACC guidelines. The co-located NDACC and TCCON stations often share one FTIR instrument, applying the respective detector and filter settings. The spectra are analysed either with the SFIT4 algorithm, an evolution
- of SFIT2 (Pougatchev et al., 1995) or with the PROFFIT9 algorithm (Hase et al., 2004) to retrieve vertical profiles of CH_4 and CO. The retrieval allows the derivation of a tropospheric and a stratospheric column of the target gases (Sepúlveda et al., 2012, 2014). The NDACC CO column values can be used directly to validate the S5P CO column values. However, for the S5P XCH₄ validation, the NDACC XCH₄ values need to be calculated. Due to the NDACC measurements being performed in the MIR range, the oxygen (O₂) total column is not available from the spectrum for calculating the column-averaged dry air
- 190 mole fractions of the target gas (Xgas), similar to what is done for TCCON (see Eq. A9 of Wunch et al. (2011)). Therefore, the total column of dry air is computed as described in Eq. 1 of Deutscher et al. (2010). The surface pressure (P_s) is recorded at the local weather station of the FTS stations and H₂O total column (TC_{H₂O}) is derived from the National Centers for Environmental Prediction (NCEP) reanalysis data set. In case if there is no surface pressure available, then we extrapolate the pressure grid to the surface. The XCH₄ calculated values for NDACC measurements are then used for the validation of the S5P
- 195 XCH₄ data. Unlike TCCON data, where a species specific scaling factor is applied to tie the measurements to the WMO in-situ scale, the NDACC data do not apply any scaling of the retrieved results. The typical accuracy and precision of the NDACC CH₄ data is about 3 % and 1.5 %, respectively. The typical accuracy and precision of the NDACC CO data is about 3 % and

1%, respectively. High systematic uncertainty is mainly due to the too conservative spectroscopic uncertainty component. Both the consolidated data available via http://www.ndacc.org and the rapid delivery data supported by the CAMS27 project

200

205

3 Validation methodology

(https://cams27.aeronomie.be/) have been used in this study.

S5P provides the total column density of CO, which can be directly validated against the NDACC CO total column density product. However, we need to calculate the corresponding XCO values in order to compare to the TCCON XCO products. The S5P XCO is calculated by taking the ratio of the total column of CO (TC_{CO}) divided by the total column of the dry air (TC_{drv.air}) (following Eq. 1 in Deutscher et al. (2010)).

$$XCO = \frac{TC_{CO}}{TC_{dry,air}} = \frac{TC_{CO}}{P_s/(g \times m_{dry,air}) - TC_{H_2O} \times (m_{H_2O}/m_{dry,air})},$$
(1)

where P_S is the surface pressure, TC_{H_2O} is the total column of H_2O , g is the column-averaged acceleration due to gravity, m_{drv.air}, and m_{H₂O} are the molecular masses of dry air and H₂O, respectively. P_s and TC_{H₂O} are taken from the S5P files.

The validation of the S5P methane and carbon monoxide data is performed based on the reference data sets from the ground-

210

based TCCON and NDACC networks. We present the results for both the networks with different co-location criteria applied to the data sets. The differences in the validation results are also based on whether or not a common prior has been used for the satellite and ground-based Fourier transform infrared (FTIR) data sets, details are discussed in Appendix A.

S5P provides daily global coverage with a huge data set having a wide swath at a high spatial resolution for every overpass. Therefore, the selection of good co-location criteria is a crucial task in finding the best strict criteria while ensuring sufficient

co-located data for a statistically significant validation. We tried several co-location criteria to test the sensitivity of the method 215 in relation to the choice of the parameter (e.g., time, distance, line-of-sight, ...). The best co-location criteria will be such that the bias is robust and not sensitive to small changes in the co-location criteria. In the next sections, the results of the application of these criteria are shown for the case with the reduction of uncertainty due to smoothing and in relation to direct comparisons.

4 Validation of S5P methane products

- 220 The validation of the S5P methane products with the ground-based FTIR data is discussed in this section. The TCCON stations cover a wide range of varying ground conditions and topography. The high latitude stations (e.g., Eureka, Ny-Ålesund, Sodankylä, East Trout Lake, ...) challenge the satellite algorithm for measurements at very high SZAs, high air masses and scenes with snow or ice coverage. The Edward site is adjacent to a very bright playa. The Park Falls and Lamont stations have relatively uniform surface properties but the ground cover can vary seasonally. The TCCON stations at Izaña and Zugspitze
- 225 are located at high altitude. Izaña along with Ascension, Réunion, and Burgos are located on small islands, remote from large landmasses but with significant topography. Several stations are located near or in urban regions with a large population (e.g., Pasadena, Paris, Tsukuba, ...). The Darwin site has the ocean to the north. The Wollongong site has the ocean on one side and

sharp escarpment on the other. The Lauder site is surrounded by hills. Nicosia is a new site, operational since August 2019, using a FTIR which was moved from the Białystock TCCON station after its closure in October 2018. The TCCON observa-

- 230 tory at Nicosia has been calibrated by vertical aircraft profiling at its former location Białystok, but not at its current location. Xianghe site in China, located in a heavily populated region, is a new site, which is operated following the recommendations of TCCON, but is not yet affiliated as a TCCON station. The NDACC stations are often located at high altitude (e.g., Altzomoni, Jungfraujoch, Mauna Loa, Zugspitze, Izaña, Maïdo, Boulder). Several of the NDACC stations are located at high latitudes (e.g., Eureka, Ny-Ålesund, Thule, Kiruna, Sodankylä). Several of the NDACC stations are located near or in urban areas (e.g.,
- 235 Bremen, St.Petersburg, Toronto, Boulder, Altzomoni close to Mexico City). The NDACC station at Arrival Heights is the only site on the Antarctic continent. TCCON provides dry-air column-averaged mole fractions of methane similar to the S5P product, whereas NDACC provides concentration profiles of methane with sensitivities up to about 20 km. As the characteristics of the two reference ground-based data sets are different, two slightly different comparison methods were applied for the validation study which are discussed in this section.

240 4.1 Validation of S5P bias-corrected vs. standard methane data using TCCON and NDACC data sets

The validation results of the S5P bias-corrected and standard methane products with reference TCCON and NDACC data are discussed in this section. The S5P observations co-located with the ground-based reference measurements are found by selecting all filtered S5P pixels within a radius of 100 km around each site and with a maximal time difference of 1 h for TCCON and 3 h for NDACC observations. The 1 h time difference for TCCON can be justified by noting that TCCON

- 245 instruments acquire only one type of spectra and from each good spectrum methane is retrieved, while NDACC instruments are required to measure different types of spectra with different optical filter configurations, making the number of methane observations more sparse. An effective location of the FTIR measurement on the line-of-sight (i.e. at a 5 km altitude) is used to do the co-location. The co-located pixels can therefore differ from measurement to measurement. For each of the ground-based measurements which are co-located with the S5P measurements, an average of all S5P pixels is done. Co-located pixels
- are created between ground-based and averaged S5P only if a minimum of five pixels is found in applying the coincidence criteria. In the comparison, the a priori in the TCCON and NDACC retrievals have been substituted with the S5P methane a priori following Eq. A1. The a priori alignment, i.e. aligning the a priori to a common one, is done to compensate/correct its contribution to the smoothing equation (Rodgers and Connor, 2003). The TCCON results with the S5P prior substituted are then compared directly to the S5P XCH₄ data. However, the NDACC CH₄ concentration profile with the S5P prior substituted
- is additionally smoothed with the S5P column averaging kernel following Eq. A2. The NDACC XCH₄ is derived as discussed in section 2.3 and then compared to the S5P XCH₄ data. Furthermore, each validation run also includes the adaptation of the S5P columns to the altitude of the ground-based FTIR instruments for cases where satellite averaging kernel is not applied or when column boundaries may differ (see Appendix B for details).

Table 4 provides the validation results for the S5P bias-corrected and standard XCH₄ data with the a priori aligned TCCON data at each TCCON station. The systematic difference (the mean of all relative differences) between the S5P and TCCON data is on average $-0.68\pm0.74\%$ (S5P standard XCH₄ product) and $-0.26\pm0.56\%$ (S5P bias-corrected XCH₄ product). Only at a few TCCON stations (Sodankylä, East Trout Lake, Park Falls and Wollongong) the bias is slightly higher than 1.5 % for the S5P standard XCH₄ product. The albedo dependence correction of the S5P XCH₄ product show a reduced bias relative to the TCCON data and are within the 1.5 %. The standard deviation of the relative bias, which is a measure of the random error, is well below 1 % for both standard ($0.59 \pm 0.17 \%$) and bias-corrected ($0.57 \pm 0.18 \%$) S5P XCH₄ products.

265

- 270 the different TCCON stations. Figure 2 shows the relative difference of the bias for the standard (top panel) and bias-corrected (bottom panel) S5P XCH₄ products as a function of the retrieved S5P SWIR surface albedo at the TCCON stations. The bias correction of the S5P XCH₄ product brings the high negative relative differences closer to zero for low surface albedo conditions and the high positive relative differences closer to zero for high surface albedo conditions. The low surface albedo conditions also show a high scatter in the relative difference plots. The latter is mainly because the scenes with low surface
- 275 albedo are challenging for satellite retrieved products due to large measurement noise. The difference of the mean relative bias between the S5P bias-corrected and the standard XCH₄ product for each TCCON station is shown as magenta bar in the middle panel plot (labeled diff_bcvsstd) of Fig. 1. It shows the overall direction of change is positive for most stations (low surface albedo conditions) and negative for few stations like Edwards, JPL, Pasadena (high surface albedo conditions). The standard deviation of the relative bias for the S5P standard and bias-corrected XCH₄ products are comparable. Scenes with low and high
- albedos pose specific challenges for S5P CH_4 retrieval. Validation of S5P CH_4 data at additional sites with different conditions (e.g. high surface albedo, high humidity, regions not covered by TCCON and NDACC) using portable FTIR spectrometers (Sha et al., 2020) will give further insight into the S5P CH_4 product quality.

The relative biases are plotted as mosaic plots and shown in Fig. 3, where the top panel shows the bias for S5P standard XCH_4 product while the bottom panel shows the bias for S5P bias-corrected XCH_4 product relative to TCCON. Each bar in

- the mosaic plots represents the weekly averages of the relative bias values. The high latitude stations show a high positive bias during the spring, which is then reduced and even switched sign to show negative bias during the autumn. Lorente et al. (2021) while analysing the improvements of their scientific S5P XCH₄ product found similar seasonality in the bias at the high latitude sites of Sodankylä and East Trout Lake and indicated correlations of high bias during spring time with the presence of snow (low surface albedo in the SWIR but high surface albedo in the NIR). In addition, the high latitude sites are also
- influenced by the polar vortex, which is difficult to be represented by the a priori. The difference of the a priori from the true atmospheric profile will also add to the bias. This will be discussed further in the next section. Since measurements rely on direct line-of-sight of the sun, data are not available during the winter months for high latitude stations. The time series of the S5P bias-corrected XCH₄ product and TCCON data for each site are shown in Figs. 4 and 5. The ground-based TCCON XCH₄ data are represented in grey and the S5P data during that period is shown in light blue. The S5P data co-located with TCCON
- 295 data are shown in blue and the co-located TCCON data with a priori alignment are shown in black. The amplitude of the CH_4 seasonal cycle is different at the different sites. This is related to the variability of the CH_4 concentrations in the atmosphere.

Figure 1 shows the bar plots for the S5P XCH₄ mean relative bias with respect to the TCCON XCH₄ data at all stations (left panel) and the standard deviation of the relative bias (right panel). The comparisons relative to the S5P bias-corrected XCH₄ product (labeled – bcsm100k1hr) are the blue bars and those for the standard XCH₄ product (labeled – stdsm100k1hr) are the magenta bars. The bias-correction of the S5P XCH₄ product being a function of the surface albedo acts differently at the different TCCON stations. Figure 2 shows the relative difference of the bias for the standard (top panel) and bias-corrected

The CH_4 concentration profile decreases rapidly with increasing altitude above the tropopause height. The concentration of CH_4 in the stratosphere, along with the troposphere, plays a key role in determining the total column of CH_4 at the given location. The CH_4 seasonal cycle in the troposphere is driven by the seasonality of both, CH_4 sources and its sinks (mainly due

- to the reaction with OH), while the CH_4 seasonal cycle in the stratosphere is dominated by the vertical transport (Sepúlveda et al., 2012; Ostler et al., 2014; Bader et al., 2017; Zhou et al., 2018). The time series of the relative bias plots shown in Figs. 6 and 7 indicate a seasonal cycle, which is clearly seen for stations with a high density of reference data with a low scatter e.g., Park Falls, East Trout Lake, Lamont, Edwards, Pasadena.
- Taylor diagram for the S5P bias-corrected XCH₄ and TCCON XCH₄ data with a priori alignment are shown in Fig. 8. The 305 correlation, represented by the angular coordinate, is above 0.6 for most stations (see Table 4 for exact values), and the distance to the origin of the ground-based dot relative to the satellite dot (ratio of std of ground-based/std of S5P) is below 1 for most stations implying that the satellite data are more variable than the ground-based data. The correlation is mostly dominated by the seasonal cycle and low correlations are seen for high latitude sites where a bias jump is seen between spring and summer periods. Outliers such as Ny-Ålesund, JPL, and Białystok are due to the limited data sets available for the comparison. Ny-
- 310 Ålesund station is located on the shore of a bay on the west coast of the island of Spitsbergen in Svalbard, Norway. As a result only a few valid S5P XCH₄ pixels are found around the station resulting in limited co-located data available for comparison. The TCCON instrument from JPL and Białystok stations were moved to Edwards and Nicosia, respectively. Thus resulting in a limited data sets available from these sites. The very low correlation for Darwin and Wollongong is due to the low satellite values for some days (see Fig. 5) and for high latitude sites is due to the jump in the bias between spring and later months
- 315 (see Fig. 6). The altitude correction of the pixels works well as can be seen by the relatively good correlation for Zugspitze, however, the scatter in the data is high.

Table 5 provides the validation results for the S5P bias-corrected and standard XCH₄ data with the smoothed NDACC data at each NDACC station. The systematic difference (the mean of all relative differences) between the S5P and NDACC data is on average -0.11 \pm 1.19 % (S5P standard XCH₄ product) and 0.57 \pm 0.83 % (S5P bias-corrected XCH₄ product). The mean of all

- 320 stations is calculated by excluding outliers, which are stations with a low number of co-locations (Ny-Ålesund, Rikubetsu), high scatter in the ground-based data (Toronto), and unexpected high bias (Thule, Arrival Height). Thule is located on the western coastline of Greenland. The valid S5P XCH4 pixels within the co-location radius around Thule show several pixels with high XCH₄ values. These high XCH₄ values are in general found along the coastline and regions with altitude variability. Although a filter for the variability of the terrain roughness is applied in the QA filter options, these high values along the coastline of
- 325 Greenland need detailed investigation and possible optimisation of the filter settings to remove the unexpected high values. We also observe valid pixels with unexpected high XCH₄ around the coastline and terrains with altitude variability in Antarctica. This is also the reason for the high bias observed at the Arrival Heights station located along the west side of Hut Point Peninsula in Ross Island, Antarctica. The bias at Altzomoni is relatively high (2.44% for S5P XCH₄ bias-corrected product), while the random error is comparable to other sites and within 1%. Bezanilla et al. (2014) found large variability in CH₄ total
- 330 columns measured at Mexico City Basin, pointing to significant local emissions affecting the natural background levels. A co-location mismatch would contribute partly to the bias seen with respect to S5P (see section 4.3 on how using an advanced

co-location criterion reduces the bias at Altzomoni). The mean standard deviation of the relative bias which is a measure of the random error is about 1 % for both S5P standard (1.05 ± 0.51 %) and bias-corrected (1.04 ± 0.52 %) XCH₄ products. The high latitude stations in the Northern Hemisphere show values slightly higher than 1 %.

- The S5P XCH₄ mean relative bias and the standard deviation of the relative bias with respect to the NDACC stations as shown in Table 5 are shown as bar plots in Fig. 9. The comparisons relative to the S5P bias-corrected XCH₄ product (labeled – bcsm100k1hr) are the blue bars and those for the standard XCH₄ product (labeled – stdsm100k3hr) are the magenta bars. The standard deviation of the relative bias (right panel) for the S5P standard and bias-corrected XCH₄ products are comparable. Figure 10 shows the relative difference of the bias for the S5P standard (top panel) and bias-corrected (bottom panel) XCH₄
- 340 products as a function of the retrieved surface albedo at the NDACC stations. Similar to the TCCON comparison, we also see here that the bias correction of the S5P XCH₄ product brings the high negative relative differences closer to zero for low surface albedo conditions and the high positive relative differences closer to zero for high surface albedo conditions. The data at stations with low surface albedo conditions also show a high scatter in the relative difference plots. The difference of the mean relative bias between the S5P bias-corrected and the standard XCH₄ product for each NDACC station is shown as magenta bar
- 345 in the middle plot (labeled diff_bcvsstd) of Fig. 9. It shows the overall direction of change is positive for most stations (low surface albedo conditions) and negative for few stations like Boulder, Altzomoni (high surface albedo conditions).

The relative biases are plotted as mosaic plots and are shown in Fig. 11, where the top panel shows the bias for the S5P standard XCH_4 product while the bottom panel shows the bias for the S5P bias-corrected XCH_4 product relative to NDACC. Each bar in the mosaic plots represents the weekly averages of the relative bias values. The high latitude stations show a high

- 350 positive bias during the spring, which is then reduced and even switched sign to show negative bias during the autumn. This is the reason for the high standard deviation of the relative difference seen for the high latitude stations having measurements during the spring and summer or autumn. Since measurements rely on direct line-of-sight of the sun, the data are not available during the winter months for high latitude stations. The time series of the S5P bias-corrected XCH₄ product and the NDACC data for each site are shown in Figs. 12 and 13, and the respective relative bias are shown in Fig. 14 and 15. In the plots, the NDACC data are shown in grey and the S5P data are shown in light cyan. The S5P data co-located with NDACC data are
 - shown in cyan and the co-located NDACC data are shown in black. Taylor diagram for the S5P bias-corrected XCH₄ and NDACC smoothed XCH₄ data are shown in Fig. 16. The correlation,

represented by the angular coordinate, is above 0.5 for most stations (see Table 5 for exact values). No clear conclusion can be drawn as to if the satellite data are more variable than the ground-based NDACC data, as we find quite some stations where the

360 distance to the origin of the ground-based dot relative to the satellite dot is both below 1 and above 1. The correlation is mostly dominated by the seasonal cycle and low correlations are seen for high latitude sites where a bias jump is seen between spring and summer periods. Outliers such as Ny-Ålesund, Rikubetsu, Porto Velho are due to the limited data sets available for the comparison. The ground-based data set from Toronto show a high scatter, while a high unexpected bias for Thule and Arrival Heights indicates some problem with the S5P data set. The ground-based data set from Harestua show a high scatter for few

365 co-locations. The low correlation for the high latitude stations Sodankylä and Kiruna is due to the jump in bias between spring

and later months (see Figs. 12 and 14).

Eight ground-based stations contributed to the validation study by providing XCH_4 data from both TCCON and NDACC measurements performed at the sites. The differences in the relative bias of the S5P bias-corrected XCH_4 product with respect

- to the TCCON and NDACC (bias_{NDACC}-bias_{TCCON}) for these stations are the following: 0.15 % (~2.9 ppb) for Eureka, 0.99 % (~18.8 ppb) for Sodankylä, 1.59 % (~30.2 ppb) for Bremen, 0.69 % (~13.1 ppb) for Karlsruhe, 0.6 % (~11.4 ppb) for Garmisch, 0.62 % (~11.8 ppb) for Zugspitze, 0.84 % (~16.0 ppb) for Wollongong, and 0.26 % (~5.0 ppb) for Lauder. Ostler et al. (2014) in a multistation (five) intercomparison study of column-averaged methane from NDACC and TCCON showed that there is no overall bias between MIR (NDACC) and NIR (TCCON) XCH₄ retrievals in general. However, dynamical variability can
- 375 cause NDACC-TCCON differences in the XCH₄ values at the sites, with values up to 30 ppb. The high latitude stations are affected by the stratospheric subsidence induced by the polar vortex, whereas for other locations, a deep stratospheric intrusion event can be the cause for the difference. Our study also shows differences between the $bias_{NDACC}$ - $bias_{TCCON}$ of the same order (up to ~30 ppb) for the co-located stations. In the next section, we show detailed results of the a priori alignment and smoothing correction at the individual stations.

380 4.2 Smoothing effect in the validation of S5P methane data

The validation of the S5P bias-corrected XCH₄ data relative to the TCCON and NDACC XCH₄ data with and without (i.e., direct comparison) a priori alignment and smoothing correction are discussed in this section. S5P, TCCON and NDACC all have different vertical sensitivities and use different a priori profiles for their retrievals. In the case of similar vertical sensitivities, we can assume that the smoothing effects from satellite and ground-based retrievals are of nearly equal magnitude. However, the vertical sensitivities and the a priori used are different, which means that the a priori profiles and the averaging kernels should be taken into account. For the case of TCCON, only an a priori alignment is done. The S5P prior is used as the common prior in our validation study. Smoothing effects are most relevant for cases with strong dynamic variability in the atmosphere. TCCON performs a profile scaling retrieval on the measurements performed in the NIR spectral region, whereas NDACC performs a profile retrieval in the MIR spectral region. The altitude of perturbation of the CH₄ profile plays a significant role on smoothing correction and is different for NIR and MIR retrievals. Ostler et al. (2014) showed that TCCON retrievals are more accurate when perturbations are due to stratosphere–troposphere exchanges in the upper troposphere/lower stratosphere (UTLS) region, whereas NDACC retrievals are more accurate for cases of stratospheric subsidence. In order to ascertain the effect of a priori alignment and smoothing, the validation results of the direct comparison are compared against the validation

395

The validation results of the S5P bias-corrected XCH₄ data relative to the TCCON and NDACC data without a priori alignment and smoothing correction (direct comparison) are shown in columns 12–15 of Tables 4 and 5, respectively. The S5P XCH₄ mean relative bias and the standard deviation of the relative bias with respect to TCCON and NDACC are shown as grey bars in the left panel and right panel plots of Figs. 1 and 9, respectively. The standard deviation of the relative bias without smoothing correction is similar to the standard deviation of the relative bias for the case with smoothing correction. The

results with a priori alignment and smoothing as discussed in the previous section.

- 400 differences between the mean relative bias with and without smoothing correction for the S5P bias-corrected XCH₄ data for each TCCON and NDACC station are shown as grey bars in the middle panel plot (labeled – diff_smvsnosm) of Figs. 1 and 9, respectively. The difference plot relative to TCCON shows that the overall direction of change is negative for all stations, with high values for most stations in the Northern Hemisphere corresponding to regions with high dynamic variability. We observe a maximum difference of -0.25 % (~-4.8 ppb) and a mean difference of -0.14 \pm 0.07 % (~-2.7 \pm 1.3 ppb) across all TCCON sites
- 405 for the duration of available measurements used in this study. The a priori alignment correction for the Southern Hemisphere sites is low where we observe on average a difference of about -0.07 % (~-1.3 ppb). The difference plot relative to NDACC shows that the overall direction of change is positive for all stations. Ny-Ålesund, which has the lowest number of collocations, shows the highest difference of 2.2 % (~41.8 ppb). Thule, which has an unexpected high bias, shows the second highest difference of 1.86 % (~35.3 ppb), and Toronto, which has a high scatter in the ground-based data, shows a high difference of 1.05 % (~20 ppb). The difference at all other stations is below 1 %, with the high values seen for high latitude sites, the mean
- 410 1.05 % (~20 ppb). The difference at all other stations is below 1 %, with the high values seen for high latitude sites, the mean difference of the selected NDACC sites shown in Table 5 is 0.38±0.28 % (~7±5.3 ppb).

As pointed out in section 4.1, the difference of smoothing (only a priori alignment for TCCON) vs. no smoothing for the eight co-located stations is observed highest for mid-latitude TCCON stations and that for the NDACC stations, we observe the highest difference for the high latitude stations. It is therefore important to use a realistic a priori profile for scaling retrievals,

415 especially for cases of stratospheric subsidence or stratosphere-troposphere exchanges. For such cases, improved a priori profiles representing the realistic atmospheric state will reduce the difference.

4.3 Comparison of circular vs. cone co-location criterion for validation of S5P methane data

In our standard S5P CH₄ validation settings with or without smoothing, we have used a co-location radius of 100 km around each ground-based site. As the operational S5P CH₄ pixels are currently provided only over land, the circular co-location criterion may not be optimal to be applied for all sites. Ground-based sites located close to a sea/ocean coast will always lack S5P CH₄ pixels over water. Furthermore, for sites located close to regions with high emission sources, there are possible scenarios when the ground-based FTIR line-of-sight is not covering all pixels observed by the satellite using the circular co-location criterion. This is also relevant for high latitude sites where the ground-based FTIRs, mostly measuring at high solar zenith angles, are always looking south for Northern Hemispheric sites and are looking north for Southern Hemispheric sites.
425 We have implemented a cone selection criterion where we follow the ground-based FTIR line-of-sight with a 1° opening angle of the cone at the highest altitude. Using the cone co-location criterion, we have done the validation of the S5P bias-corrected CH₄ data with smoothing and compared to the validation results using circular co-location criterion using the same settings as discussed in section 4.1.

The validation results of the S5P bias-corrected XCH₄ data relative to the TCCON and NDACC data applying cone colocation criterion are shown in columns 16–20 of Tables 4 and 5, respectively. Using the cone co-location criterion reduces the number of S5P co-locations with ground-based FTIRs significantly (see column 16 in relation to column 3). The S5P XCH₄ mean relative bias and the standard deviation of the relative bias with respect to TCCON and NDACC using the cone co-location criterion are shown as orange bars in the left panel and right panel plots of Figs. 1 and 9, respectively. The standard deviation of the relative bias with the cone co-location criterion is smaller than the standard deviation of the relative bias for the

- 435 circular co-location criterion for sites with significantly reduced co-locations and is similar for other sites with small reduction in the number of co-locations. The difference between the mean relative bias with circular and cone co-location criterion for the S5P bias-corrected XCH₄ data for each TCCON and NDACC station is shown as orange bars in the middle panel plot (labeled – diff_circvscone) of Figs. 1 and 9, respectively. The difference plot relative to TCCON shows the magnitude of change in bias, with values for some stations being negative while for others stations being positive. We observe a maximum difference
- 440 of 0.3 % (~5.7 ppb) and a mean difference of -0.02±0.12 % (~-0.4±2.3 ppb) across all TCCON sites for the duration of available measurements used in this study. The high latitude sites in the Northern Hemisphere show a significantly low number of co-locations for the cone criterion. The relative bias for these sites, Eureka, Ny-Ålesund, Sodankylä, and East Trout Lake, shows a slight increase for the cone co-location criterion in comparison to the circular co-location criterion. Sites where the relative bias using the cone criterion as compared to the circular criterion is lower by at least 2 ppb are the following: JPL
- (-0.2%), Pasadena (-0.18%), Lamont (-0.11%), and Białystok (-0.11%). While the sites where the cone criterion as compared to the circular criterion is higher by at least 2 ppb are the following: Lauder (0.3%), Saga (-0.18%), and Orléans (0.1%). The difference plot relative to NDACC shows the magnitude of change in bias with values for some stations being negative while for other stations being positive. We observe a maximum difference of 0.49% (~9.3 ppb) and a mean difference of 0.01±0.2% (~0.2±3.8 ppb) across the selected NDACC sites (see Table 5) for the duration of available measurements used in
- 450 this study. Several sites have few co-locations left upon selecting the cone criterion with Ny-Ålesund showing no match at all. Amongst the sites where a significant number of co-locations remains, the sites where the relative bias using the cone criterion as compared to the circular criterion is lower by at least 2 ppb are the following: Altzomoni (0.49 %), Sodankylä (0.14 %), and Jungfraujoch (-0.11 %). The sites where the cone criterion as compared to the circular criterion is higher by at least 2 ppb are the following: Lauder (-0.30 %), Kiruna (0.25 %), Bremen (-0.15 %), and St. Petersburg (-0.12 %).
- We have observed that applying the cone co-location criterion reduces the number of co-locations for all sites and quite significantly for some sites. There are seven TCCON stations and seven NDACC stations where the magnitude of the difference is above 2 ppb. Amongst all the stations, the magnitude of change in the relative bias between the two settings is the highest for Altzomoni station (see section 5.3 for further discussion on the site).

4.4 Solar zenith angle dependence of the S5P methane bias relative to ground-based reference data

- 460 The remote sensing measurements made either from the ground or satellite are known to be affected by the solar zenith angle (SZA) of the measurements. In this section, we show the S5P CH₄ bias relative to the ground-based reference data as a function of the measurement SZA. Figure 17 shows the S5P relative bias for the a priori aligned and smoothed case as a function of the measurement SZA against some of the reference ground-based TCCON stations. As mentioned in section 2.1, the S5P CH₄ data are only available for SZA \leq 70°. The upper limits of the plots therefore show values only until 70°. The S5P relative bias
- 465 shows a high scatter for high SZAs. Stations like Sodankylä, East Trout Lake, and Park Falls show high values in the relative bias for measurements at high SZAs when measurements are performed during winter and spring months. These measurements are influenced by surface conditions with snow cover and polar vortex conditions. Whereas, the negative bias at high SZA is

from the summer and autumn measurements (e.g., see Figs. 6 and 7). At Lamont we observe a strong increase in bias with decreasing SZA for measurements performed during spring. This is seen particularly in the case where the bias correction due

470 to the SWIR surface albedo change occurred between 0.25 and 0.1 for measurements performed in this period at the site. The bias increase with decreasing SZA is also seen for other months at the different sites. Except for the spring measurements, which show a high bias, we observe a general decrease in relative bias with increasing SZA.

5 Validation of S5P carbon monoxide products

the ground-based FTIR instruments.

The validation of the S5P carbon monoxide data with the ground-based FTIR data from TCCON and NDACC stations is
discussed in this section. The official S5P CO products are available over land as well as over water. As a result, in addition to the stations mentioned in the S5P methane validation results, co-locations with ground-based stations located on islands (e.g., Ascension, Izaña, Réunion, and Mauna Loa) are found and discussed here. The NDACC station at Paramaribo and Porto Velho are the only stations in the South American continent currently contributing to the S5P CO validation study. As NDACC provides the CO column values, they are used directly to validate the S5P CO column values. Whereas for the validation using TCCON XCO data, the S5P CO columns are converted to XCO as described in section 3.

5.1 Validation of S5P XCO data using TCCON standard and unscaled XCO data and analysis of smoothing uncertainty

As mentioned in section 2.2, the validation of the S5P XCO offline data is performed with the TCCON standard XCO data as well as the TCCON unscaled XCO data and the results are discussed in this section. The density of the official S5P valid CO pixels is higher as compared to the valid CH₄ pixels. As a result, we found that using a co-location radius of 50 km around each ground-based station gave a sufficient number of pixels for robust statistics. We have used a maximal time difference of 1 h for TCCON observations, which is similar to the settings used for CH₄ validation. An effective location of the FTIR measurement on the line-of-sight is used to do the co-location. As a result, the co-located pixels can differ from measurement to measurement. For each of the ground-based measurements, which are co-located with the S5P measurements, an average of all S5P pixels is made. Co-located pairs are created between ground-based and averaged S5P pixels only if a minimum of five pixels is found in applying the coincidence criteria. In the comparison, the a priori in the TCCON retrievals have been substituted with the S5P CO a priori following Eq. A1. The TCCON results with the S5P prior substituted are then compared directly to the S5P XCO data. Furthermore, each validation run includes the adaptation of the S5P columns to the altitude of

Table 6 provides the validation results using the a priori aligned TCCON unscaled and standard XCO data at each TCCON station. The systematic difference (the mean of all relative differences) between the S5P and TCCON data is on average 9.22±3.45% (TCCON standard XCO data) and 2.45±3.38% (TCCON unscaled XCO data). The absolute maximum bias value of 8.27% is observed with respect to TCCON unscaled XCO data. While most stations show a positive relative bias of S5P XCO with respect to the TCCON unscaled XCO, there are few exceptions that show high negative values (e.g., Xianghe,

- 500 JPL, and Pasadena all urban sites). This will be further discussed in detail later in this section. The standard deviation of the relative bias, which is a measure of the random error, is well below <10 % for comparison against both TCCON standard and unscaled XCO data at all stations except at Wollongong where the value is 17.93 % (for TCCON unscaled XCO) and 19.37 % (for TCCON standard XCO). The high standard deviation of the relative bias at this station is due to the co-location mismatch during the period of fire event in that region producing enhanced CO plume passing over/nearby the ground-based station at
- 505 Wollongong. As a result for some of the days we found enhanced CO values in the S5P co-located pixels, which were not observed by the FTIR as the enhanced CO plume is not directly in the line-of-sight of the FTIR, while for other days we found enhanced CO values varying during the day as the fire plume passes by the station and in comparison the satellite measures for a shorter duration during the local noon and therefore misses the variability of CO during the co-location time selected for the validation. We tested with a reduced time co-location criterion of 30 min and found that, for the Wollongong station,
- 510 the standard deviation of the relative bias reduced marginally to 17.89 % and the relative bias reduced to 1.87 % (for TCCON unscaled XCO validation results). The CO plumes emitted from the Australian fire during the summer of 2019/2020 were also observed at the Lauder station in New Zealand. The CO was well dispersed by the time the fire plumes were measured there, resulting in a better match between the S5P and ground-based FTIR measured XCO (see Figs. 20 and 22).
- Figure 18 shows the bar plots for the S5P XCO mean relative bias (left panel) and the standard deviation of the relative bias
 (right panel) with respect to the TCCON XCO data at all stations. The comparisons relative to the TCCON unscaled XCO data (labeled unscsm50k1h) are the blue bars and those for the TCCON standard XCO data (labeled stdsm50k1hr) are the magenta bars. The mean relative bias of the S5P XCO data with respect to the TCCON unscaled XCO data is systematically lower than the mean relative bias with respect to the TCCON standard XCO data. The difference of the mean relative bias for S5P XCO data using the TCCON unscaled XCO and the standard XCO data for each station is shown as magenta bar in the middle panel plot (labeled diff unscvsstd) of Fig. 18. It shows the overall direction of change is negative with mean value
- 520 middle panel plot (labeled diff_unscvsstd) of Fig. 18. It shows the overall direction of change is negative with mean value of -6.77±0.57% for all stations. The result confirms the previously reported studies (Kiel et al., 2016; Sha et al., 2018b; Zhou et al., 2019) showing that the correction factor to tie the TCCON XCO data to WMO in situ scale is large and that TCCON XCO data is smaller than the uncorrected XCO data by about 7%. The standard deviation of the relative bias for the S5P XCO data relative to the TCCON unscaled and standard XCO data are comparable.
- The time series of the S5P XCO and TCCON unscaled XCO data for each site are shown in Figs. 19 and 20. The groundbased TCCON XCO data are represented in grey and the S5P XCO data during that period are shown in light red. The S5P data co-located with TCCON data are shown in red and the co-located TCCON data with a priori alignment are shown in black. The S5P and TCCON measurements observe the same seasonal cycle of CO. At the Northern Hemispheric sites, the high CO values are observed during winter and low values are observed during summer dominated by the OH variation (Té et al., 2016).
- 530 At Southern Hemispheric sites, the high CO values are observed during September November dominated by the influence of biomass burning (Duflot et al., 2010; Zeng et al., 2012). In addition to the seasonal cycle, we also see that at several of the ground-based sites, S5P and TCCON observe sometimes very high values of CO. These enhanced CO concentrations are due to the passing of the plumes with elevated CO concentrations over/nearby the station location (e.g., high CO seen at Wollongong during the Australian forest fires November 2019 February 2020). Yurganov et al. (2004) also reported enhanced CO buildup

- 535 measured at several sites with values much larger than the emission estimates. The time series of the relative bias plots shown in Figs. 21 and 22 indicate a seasonal cycle with a high bias seen during the high CO event and low bias seen during the low CO event. Sometimes very low S5P XCO values are observed in the validation plots at some stations, which pass the quality filter and find a match with the reference TCCON XCO data following our selection criterion. In these particular cases, we observe very low values in the relative bias plots. However, there are only a few occurrences of such low S5P XCO values.
- 540 The relative biases are plotted as mosaic plots and shown in Fig. 23, where the top panel shows the S5P bias with respect to the TCCON unscaled XCO data. Each bar in the mosaic plots represents the weekly averages of the relative bias values. We will focus on the comparison of the results using TCCON unscaled XCO data. As mentioned in the previous paragraph, we observe a high positive bias during the high CO event periods, which is then reduced and even switched sign to show a negative bias during the low CO event periods.
 545 As TCCON performs solar absorption measurements, data are not available during winter for high latitude stations.
- Taylor diagram for the S5P XCO and TCCON unscaled XCO data with a priori alignment are shown in Fig. 24. The correlation, represented by the angular coordinate, is above 0.9 for most stations (see Table 6 for exact values), and the distance to the origin of the ground-based dot relative to the satellite dot is below 1 for most stations implying that the satellite data are more variable than the ground-based data. The good correlation indicates that the short scale temporal variations in the XCO column captured by the ground-based instruments are moderately reproduced by S5P. Outliers such as Ascension, Zugspitze, and JPL are due to the limited data sets available for the comparison. The altitude correction of the pixels works well as can be seen by the relatively good correlation for Zugspitze, however, the scatter in the data is high.

In this section, we further show the results focusing on the effect of smoothing while doing the S5P XCO validation against TCCON unscaled XCO data. S5P and TCCON have different vertical sensitivities (averaging kernels) and use different a priori profiles for their retrievals. The different a priori and vertical sensitivities should be taken into account in the validation. In case of TCCON only an a priori alignment is done. Smoothing corrections are most relevant for cases with strong dynamic variability in the atmosphere. TCCON performs a profile scaling retrieval on the measurements performed in the NIR spectral range and provides XCO. In order to ascertain the effect of smoothing correction, the results of the S5P validation using TCCON unscaled XCO are compared to the S5P validation results using a priori aligned TCCON unscaled XCO data.

- The validation results of the S5P XCO data relative to the TCCON unscaled XCO data without smoothing correction (direct comparison) are shown in columns 12–15 of Table 6. The S5P XCO mean relative bias and the standard deviation of the relative bias with respect to the TCCON unscaled XCO data are shown as grey bars (labeled unsc50k1h) in the left panel and right panel plots of Fig. 18. It can be seen that there exists an apparent interhemispheric difference in the bias for the direct comparison case (grey bars) between the Southern Hemispheric and Northern Hemispheric sites. This difference is greatly
- 565 reduced when smoothing uncertainties are correctly accounted (blue bars) in the validation results (see left panel of Fig. 18). The difference between the mean relative bias with and without a priori alignment for the S5P XCO data for each TCCON station are shown as grey bars in the middle panel plot (labeled diff_smvsnosm) of Fig. 18. The magnitude of change between the smoothed and direct comparison is larger in the Southern Hemisphere than in the Northern Hemisphere with exception for sites located in high polluted regions. The change at some stations (e.g., the Southern Hemispheric sites and high polluted

- 570 sites) is significant as it is larger than the XCO error estimated in Wunch et al. (2015). Zhou et al. (2019) reported similar findings for a comparison between six co-located sites, where both NDACC and TCCON CO measurements were performed. The difference plot shows the highest value of -17.43 % for Xianghe, a station located in a polluted area, due to a very high a priori difference from the true atmospheric state. As a result, the CO volume mixing ratio (VMR) at the surface is relatively high but it is not represented by the TCCON a priori, leading to an underestimation from the smoothing uncertainty. The same
- 575 is true for other stations like Karlsruhe (change of -5.73 %), and Pasadena (change of -3.62 %). We observe a mean difference of 0.43 ± 4.44 % across all TCCON stations. Figure 18 shows the TCCON stations where the a priori alignment uncertainty plays an important role in the bias and needs to be accounted for in the CO validation studies.

5.2 Validation of S5P CO column data using NDACC CO column data and analysis of smoothing uncertainty

- In this section, the validation results of the S5P CO columns using NDACC CO columns are discussed. The S5P observations co-located with the NDACC measurements are found by selecting all filtered S5P pixels within a radius of 50 km around each site and with a maximal time difference of 3 h. An effective location of the measurement on the line-of-sight is used to do the co-location. The co-located pixels can therefore differ from measurement to measurement. For each of the NDACC measurements, co-located with the S5P measurements, an average of all S5P pixels is done. Co-located pairs are created between NDACC and averaged S5P only if a minimum of five pixels is found in applying the coincidence criteria. In addition to the direct comparison of the S5P and NDACC CO columns (referred to as NDACC CO un-smooth), the NDACC CO column values are additionally aligned with the S5P prior (referred to as NDACC CO ap-smooth) and used for the S5P validation, and in a further step the NDACC CO column values with the S5P prior substituted are additionally smoothed with the S5P column
- averaging kernel (referred to as NDACC CO smooth) following Eq. A2 and used for S5P validation. Each validation run also includes the adaptation of the S5P columns to the altitude of the ground-based FTIR instruments.
- Table 7 provides the validation results for the S5P CO columns using smooth, un-smooth, and ap-smooth NDACC CO column data at each NDACC station. The systematic difference (the mean of all relative differences) between the S5P and NDACC data is on average $6.76\pm4.65\%$ (NDACC CO un-smooth), $4.27\pm5.62\%$ (NDACC CO ap-smooth), and $7.62\pm5.04\%$ (NDACC CO smooth). However, the bias values are quite high at Altzomoni and Arrival Heights stations. Eliminating the results of these two stations from the statistics of the overall stations, we observe the systematic difference between the S5P and
- 595 NDACC data is on average 5.69±3.07 % (NDACC CO un-smooth) 3.14±4.19 % (NDACC CO ap-smooth), and 6.5±3.54 % (NDACC CO smooth). The NDACC station at Altzomoni is located at a high altitude in the south-west direction of the Mexico City (Plaza-Medina et al., 2017; Baylon et al., 2017). The station is located <60 km from the city center. As a result, the emission from the world's eighth-largest megacity, with >22 million population in its metropolitan area, plays a significant role in the satellite footprint (Stremme et al., 2013; Borsdorff et al., 2018a, 2020). In the example plot shown in Fig. 25, we can see
- 600 that the ground-based FTIR located at Altzomoni, with the line-of-sight to the south indicated by the yellow line, is not able to observe the high CO values located to the north-west of the station, which are selected for S5P using our co-location criterion. However, using the cone co-location criterion as described in section 4.3 we can eliminate the pixels with high CO values that are not in the line-of-sight of the FTIR instrument and thereby reduce the co-location mismatch. The bias at Arrival Heights,

high latitude background station located on the Antarctic continent showing very low values of CO, is slightly worse than the

- 605 requirement, while the random error is way below 10%. The mean standard deviation of the relative bias, which is a measure of the random error, is well below <10% for validation using both smoothed and direct NDACC CO data. However, there are few exceptions for stations like Altzomoni, Wollongong, and Boulder. The high values are due to the co-location mismatch during the high CO events (e.g., passage of a plume with a high CO concentration in the vicinity of the site) observed at these sites.
- Figure 26 shows the bar plots for the S5P CO mean relative bias (left panel) and the standard deviation of the relative bias (right panel) with respect to the NDACC CO column data at all stations. The comparisons relative to the NDACC smoothed CO data (labeled ALLsm50k3h) are the blue bars, those for the NDACC un-smooth CO data (labeled ALL50k3hr) are the magenta bars, and those for the NDACC ap-smooth CO data (labeled ALLap50k3h) are the grey bars. The high latitude stations show a high bias while some stations like Paramaribo, Izaña, Mauna Loa show a low bias. The difference of the mean
- 615 relative bias for S5P CO data for the NDACC smoothed CO (labeled diff_smvsnosm) and NDACC ap-smooth (labeled diff_apvsnosm) relative to the un-smooth CO data for each station are shown as magenta and grey bars in the middle panel plot of Fig. 26. It shows the magnitude of change in bias with values for some stations being positive while for other stations being negative. The effect of smoothing appears to be dependent on the station location. We observe a maximum difference of -6.89 % and a mean difference of 0.86±2.79 % for all stations for the diff_smvsnosm case. And we observe a maximum
- 620 difference of -11.26 % and a mean difference of -2.49±2.96 % for all stations for diff_apvsnosm case. The changes at some stations are significant as it is larger than the CO column error estimated in NDACC. The standard deviation of the relative bias for the S5P CO data relative to the NDACC CO data with and without smoothing is comparable.

The time series of the S5P CO column and NDACC smoothed CO column data for each site are shown in Figs. 27 and 28.
The ground-based NDACC CO data are represented in grey and the S5P data during that period are shown in light red. The
S5P data co-located with NDACC data are shown in red and the co-located NDACC smoothed data are shown in black. The implication of the altitude correction can easily be seen for stations located at high altitude (e.g., Zugspitze, Jungfraujoch, Izaña, Mauna Loa, Altzomoni, Maïdo). The S5P and NDACC measurements observe the same seasonal cycle of CO. Similar to the TCCON results, we also see that at several of the NDACC sites, S5P and NDACC sometimes observe very high values of CO columns due to the passing of the plumes with elevated CO concentrations over/nearby the station location (e.g., Wollongong, Boulder, St. Petersburg, Porto Velho). The time series of the relative bias plots shown in Figs. 29 and 30 indicate a seasonal

cycle with a high bias seen during the high CO event and low bias seen during the low CO event. The high scatter observed at the Toronto site is related to the scatter observed in the ground-based NDACC CO column data at the site.

The relative biases of the S5P CO column and NDACC smoothed CO column data for each site are shown as a mosaic plot in Fig. 31. Each bar in the mosaic plot represents the weekly averages of the relative bias values. The plot shows high positive

635

bias during the high CO event periods, which is then reduced and even switched sign to show negative bias during the low CO event periods. The bias at few stations like Toronto, Altzomoni, and Arrival Heights appear as outliers in the plot. As NDACC CO column data are retrieved from solar absorption measurements, the data are not available during few weeks in winter for high latitude stations when the sun is very low on the horizon.

- Taylor diagram for the S5P CO column and NDACC smoothed CO column data are shown in Fig. 32. The correlation, represented by the angular coordinate, is above 0.9 for most stations (see Table 7 for exact values), and the distance to the origin of the ground-based dot relative to the satellite dot is below 1 for most stations (except at Paramaribo and Rikubetsu which is due to the limited data sets available for the comparison) implying that the satellite data is more variable than the ground-based data. The good correlation indicates that the temporal variations in the CO column captured by the ground-based instruments are reproduced very similarly by S5P. Outliers such as Wollongong, Boulder, and Altzomoni are due to the co-location mismatch during the high CO events (e.g., passage of a plume with a high CO concentration in the vicinity of the site) observed at
- these sites. The altitude correction of the pixels works well as can be seen by the relatively good correlation at the high altitude stations.

Eleven ground-based stations (Eureka, Ny-Ålesund, Bremen, Karlsruhe, Garmisch, Zugspitze, Rikubetsu, Izaña, Réunion–
Maïdo, Wollongong, and Lauder) contributed to the validation study by providing CO data from both TCCON and NDACC measurements performed at the sites. The mean difference in the relative bias of the S5P CO data with respect to the smoothed NDACC and TCCON (bias_{S5PvsNDACC}-bias_{S5PvsTCCON}) for these eleven stations is -4.41±3.68%. This indirectly implies that the NDACC CO is 4.41±3.68% larger than TCCON CO data. The ground-based data available for these eleven stations do not always cover the same period. Therefore, this is only a qualitative estimate indicating the mean difference between NDACC
and TCCON CO data at these eleven sites. Zhou et al. (2019) showed that the bias between co-located and smoothed TCCON and NDACC XCO data products for six stations has a mean value of 6.8% (range 5.6%–8.6%). Our indirect comparison

results for more sites and not exactly co-located ground-based data for the TCCON and NDACC show similar differences.

5.3 Comparison of circular vs. cone co-location criterion for validation of S5P carbon monoxide data

In our standard S5P CO validation settings with or without smoothing, we have used a co-location radius of 50 km around each ground-based site. In this section, we will discuss the validation results of the S5P CO column data with the smoothed ground-based data following the cone co-location criterion as described in section 4.3. These results are further compared to the circular co-location criterion using the same settings.

The application of the cone co-location criterion is shown in Fig. 25 for one sample day. The top-left panel plot shows all available S5P pixels containing CO column number density data in the overpass file. The Altzomoni station is marked at the center of the plot. The high CO values to the north-west of the station are the footprint of the CO from Mexico City. Towards the northeast side of the station some missing pixels are filtered due to clouds. The top-right panel plot shows the co-located S5P pixels with circular co-location criterion with a radius of 50 km as used for the CO validation study. As seen in the plot, there are few pixels with high CO values in the north-west, which are included in the selected pixels. The yellow line in the plot represents the line-of-sight of the ground-based FTIR at Altzomoni. Therefore, the high CO values in the north-west will

670 not be observed by the FTIR measurement. This mismatch is a cause of the potential bias. The bottom panel plot shows the co-located S5P pixels with the cone co-location criterion with 1° opening angle of the cone at the highest altitude. The selected

S5P pixels using the cone co-location criterion are in the line-of-sight of the ground-based FTIR instrument and will potentially reduce a mismatch and therefore lowering the potential bias between the satellite and ground-based data.

675

The validation results of the S5P CO data relative to the TCCON and NDACC data with smoothing and applying cone co-location criterion are shown in columns 16–20 of Tables 6 and 7, respectively. Using the cone co-location criterion only marginally reduces the number of S5P co-locations with ground-based FTIRs (see column 16 in relation to column 3). This is due to the high density of the official S5P valid CO pixels availability. The S5P CO mean relative bias and the standard deviation of the relative bias with respect to TCCON and NDACC using the cone co-location criterion are shown as orange bars in the left panel and right panel plots of Figs. 18 and 26, respectively. The S5P CO mean relative bias is comparable 680 or slightly smaller for the cone co-location criterion as compared to the circular co-location criterion. The standard deviation of the relative bias with the cone co-location criterion is similar to the standard deviation of the relative bias for the circular co-location criterion. The difference between the mean relative bias with circular and cone co-location criterion for the S5P CO data for each TCCON and NDACC station is shown as orange bars in the middle panel plot (labeled - diff circvscone) of Figs. 18 and 26, respectively. The difference plot relative to TCCON shows the magnitude of change in bias, with values for

685 some stations being negative while for other stations being positive. We observe a maximum difference of 2.56 % and a mean difference of 0.09 ± 0.55 % across all TCCON sites for the duration of available measurements used in this study. Sites where the relative bias using the cone criterion as compared to the circular criterion is outside the 1 σ limit of the mean are Eureka (0.6%), Garmisch (0.81%), and Zugspitze (2.56%). The difference plot relative to NDACC shows the magnitude of change in bias, with values for some stations being negative while for other stations being positive. We observe a maximum difference of

690 -1.35 % and a mean difference of -0.07 ± 0.47 % across the selected NDACC sites for the duration of available measurements used in this study. The sites where the relative bias using the cone criterion as compared to the circular criterion is outside the 1σ limit of the mean are Eureka (0.78 %), Harestua (-0.53 %), Zugspitze (-0.75 %), Jungfraujoch (-0.7 %), Boulder (0.64 %), Arrival Heights (-1.35%). The high difference is observed mostly for the high latitude stations where the cone co-location criteria following the ground-based FTIR line-of-sight is the best choice.

695 5.4 Validation of S5P CO (CLSKY, CLOUD, and ALL) data using TCCON and NDACC data sets

As discussed in section 2.1, we separated S5P retrievals performed for measurements under clear-sky (CLSKY; cloud optical thickness <0.5 & cloud height <500 m, over land) and cloudy conditions (CLOUD; cloud optical thickness > 0.5 & cloud height <5000 m, over land and ocean) in addition to our standard all case (ALL; cloud height <5000 m over land and ocean). The validation results of S5P CO for ALL settings have been discussed in detail in sections 5.1 - 5.3. In this section, we show

700 the validation results of the S5P CO for CLSKY and CLOUD settings against TCCON unscaled XCO with a priori alignment and NDACC CO column data with smoothing and compare the results in relation to the results of the ALL settings. Each validation run includes the adaptation of the S5P columns to the altitude of the ground-based FTIR instruments.

Tables 8 and 9 provide the validation results for the S5P CO data for ALL case, CLSKY case, and CLOUD case at each TCCON and NDACC station. The systematic difference (the mean of all relative differences) between the S5P and unscaled TCCON data is on average $2.45\pm3.38\%$ (ALL case), $2.83\pm3.43\%$ (CLSKY case), and $1.89\pm3.11\%$ (CLOUD case). The 705

22

standard deviation of the relative bias, which is a measure of the random error, is well below <10% for all sites except at Wollongong (ALL and CLOUD cases) and Pasadena (CLOUD case).

Figure 33 shows the bar plots for the S5P XCO mean relative bias (left panel) and the standard deviation of the relative bias (right panel) with respect to the TCCON XCO data at all stations. The comparisons relative to the TCCON unscaled XCO data

- 710 for ALL case (labeled unscsm50k1hALL) are the blue bars, those for the CLSKY case (labeled unscsm50k1hCLSKY) are the red bars, and those for the CLOUD case (labeled – unscsm50k1hCLOUD) are the green bars. The middle panel plot of Fig. 33 shows for each TCCON station the difference of the mean relative bias for S5P XCO data using the TCCON unscaled XCO ALL case and the CLSKY case (labeled – diff_ALLvsCLSKY) as red bars, as well as the CLOUD case (labeled – diff_ALLvsCLOUD) as green bars. The overall direction of change for the CLSKY case is negative with few exception, the
- maximum value of change is 2.41 % and a mean value of -0.38 ± 1.05 % for all stations. The overall direction of change for the CLOUD case is positive with few exceptions, maximum value of change is 3.14 % and a mean value of 0.55 ± 0.79 % for all stations.

The systematic difference (the mean of all relative differences) between the S5P and NDACC data is on average $7.62\pm5.04\%$ (ALL case), $7.7\pm4.96\%$ (CLSKY case), and $7.74\pm4.97\%$ (CLOUD case). The validation results at the Altzomoni and Arrival

- 720 Heights stations show quite high bias also for the CLSKY and CLOUD cases, similar to that observed for the ALL case. Eliminating the results of these two stations from the statistics of the overall stations, we observe that the systematic difference between the S5P and NDACC data is on average 6.5±3.54% (ALL case), 6.49±3.11% (CLSKY case), and 6.68±3.69% (CLOUD case). The random error at Arrival Heights, a high latitude station located on the Antarctic continent, is well below 10%. The mean standard deviation of the relative bias, which is a measure of the random error, is well below <10% for all three cases of validation results with few exceptions for stations like Altzomoni, Wollongong, and Boulder. The high values</p>
- 725 three cases of validation results with few exceptions for stations like Altzomoni, Wollongong, and Boulder. The high values are due to the co-location mismatch during the high CO events (e.g., the passage of a plume with a high CO concentration in the vicinity of the site) observed at these sites.

Figure 34 shows the bar plots for the S5P CO mean relative bias (left panel) and the standard deviation of the relative bias (right panel) with respect to the NDACC CO column data at all stations. The comparisons relative to the NDACC CO column

- 730 data for ALL case (labeled ALLsm50k3h) are the blue bars, those for the CLSKY case (labeled ALLsm50k3hCLSKY) are the red bars, and those for the CLOUD case (labeled ALLsm50k3hCLOUD) are the green bars. The middle panel plot of Fig. 34 shows for each NDACC station the difference of the mean relative bias for S5P CO column data using the NDACC CO column ALL case and the CLSKY case (labeled diff_ALLvsCLSKY) as red bars, as well as the CLOUD case (labeled diff_ALLvsCLSKY) as green bars. The direction of change for the CLSKY and CLOUD cases is negative for some stations
- 735 while for other stations it is positive. The maximum value of change is 2.68 % and a mean value of 0.23 ± 1.11 % for CLSKY case for all stations. The maximum value of change is -1.73 % and a mean value of -0.11 ± 0.83 % for CLOUD case for all stations.

The CLSKY and CLOUD selection criteria can be useful in case of specific applications. For example, the CLSKY case helped to reduce the standard deviation of the relative bias for Wollongong's TCCON and NDACC validation results. This

740 is related to the significant filtering of the pixels over the ocean that are missing in the CLSKY case. The satellite clear-sky

observations made over ocean have too low signal in the SWIR spectral region and are therefore filtered out. However, the ALL case results are quite comparable to the CLSKY and CLOUD cases in general and are therefore used as the general S5P CO data set in our validation studies.

5.5 Solar zenith angle dependence of the S5P carbon monoxide bias relative to ground-based reference data

- 745 In this section, we show the S5P carbon monoxide bias relative to the ground-based reference data as a function of the measurement SZA. Figure 35 shows the S5P relative bias for the a priori aligned and smoothed case as a function of the measurement SZA against the reference ground-based TCCON stations at Sodankylä (left panel) and Lauder (right panel). As mentioned in section 2.1, the S5P carbon monoxide data are only available for SZA<80°. The upper limits of the plots therefore show values only till 80°. As explained in section 5.2, the high values of S5P relative bias are observed during winter (measurements per-</p>
- 750 formed mostly at high SZAs) and the low values during summer (measurements performed mostly at low SZAs). We observe that the relative bias increases with increasing SZA of the measurement. This increase is about 10% over the complete range of measurements SZAs.

6 Conclusions

In this study, we have done the geophysical validation of Sentinel-5 Precursor operational methane and carbon monoxide data sets (see Table 1 for version details) using reference ground-based TCCON and NDACC stations. A total of 28 TCCON stations and 24 NDACC stations covering a wide latitudinal range (Eureka 80° N to Arrival Heights 77.8° S), various atmospheric conditions (dry, humid, clean and polluted), various surface conditions (range of surface albedo), flat and high altitude terrains, oceanic terrain have been used in this study. Furthermore, the combined use of the near-infrared TCCON data and mid-infrared NDACC data, as a whole network and at co-located stations, with their benefits helped to evaluate the Sentinel-5 Precursor operational methane and carbon monoxide product's quality in our validation study.

We found that the systematic difference between the S5P standard XCH₄ and a priori aligned TCCON data is on average - 0.68 ± 0.74 %. The systematic difference changes to a value of -0.26 ± 0.56 % for the S5P bias-corrected XCH₄ data. The bias for both S5P standard and bias-corrected XCH₄ data is well within the mission requirements for bias (systematic error) of 1.5 %. We also found that the random error is well below 1 % for both standard (0.59 ± 0.17 %) and bias-corrected (0.57 ± 0.18 %) S5P

- 765 XCH₄ data. Most stations show a correlation above 0.6, the poor correlation at some sites are mostly dominated by the seasonal cycle or due to limited data sets available for the comparison. The systematic difference between the S5P standard and bias-corrected XCH₄ against smoothed NDACC data are on average -0.11 \pm 1.19% and 0.57 \pm 0.83%, respectively. As the accuracy and precision of NDACC CH₄ data are lower than TCCON, conclusions about the S5P systematic and random error are drawn based on TCCON validation results. The bias-correction of the S5P XCH₄ data being a function of the retrieved surface albedo
- acts differently at different locations. We observe high scatter in the relative bias for low surface albedo conditions. A seasonal dependency of the relative bias is seen. We observe a high bias during the springtime measurements at high SZAs for high latitude sites and a decreasing bias with increasing SZA for the rest of the year at all sites. The SZA dependence of the

bias includes albedo correction and a priori difference from the true atmospheric state. We estimated the contribution of the a priori alignment uncertainty at the ground-based stations and found values up to ~4.8 ppb at a TCCON station with mean

- value of $\sim -2.7 \pm 1.3$ ppb. The mean value of the smoothing uncertainty contribution at the NDACC stations is $\sim 7\pm 5.3$ ppb with some stations showing high values of up to ~ 41.4 ppb. At the co-located TCCON and NDACC stations, we observed the highest contribution of the a priori alignment and smoothing uncertainty for mid-latitude TCCON stations, whereas for the NDACC stations we observe the highest contribution for the high latitude stations. The comparison with a priori alignment and taking smoothing effects into account is recommended as the preferred method. However, the direct comparison of the satellite
- and reference data is useful to see the influence of the averaging kernel and a priori difference to the true profile. We found that using the cone co-location criterion improves the co-location between the satellite and ground-based station by observing similar airmass. This is crucial for certain stations, which are located closer to emission sources or high latitude ones. Currently, we found seven TCCON and NDACC stations where the bias changed by more than 2 ppb between the circular and cone co-location settings. The cone criterion also significantly reduces the number of co-locations for some sites thereby making the
- 785 statistics less reliable for those sites. The L2 algorithm teams are continuously working on improving the operational products by optimising their code with respect to the observed biases seen with respect to the reference data sets. These improvements will be implemented in future versions of the S5P data.

We found that the systematic difference between the S5P XCO and a priori aligned TCCON data is on average 9.22 ± 3.45 %. Due to the uncertainty of the scaling slope of XCO in TCCON to tie the TCCON XCO measurements to WMO in situ scale, we

- 790 have also used the unscaled TCCON XCO data (without application of the empirical scaling factor) for S5P XCO validation. We found that the systematic difference between the S5P XCO and a priori aligned TCCON unscaled XCO data is on average 2.45±3.38 %. Both results are within the mission requirements for bias (systematic error) of 15 %. We found that the difference of the relative bias using the TCCON unscaled XCO and the TCCON standard XCO data is on average -6.77±0.57 %. We estimated the contribution of the a priori alignment uncertainty in the validation and found that the magnitude of change
- between the a priori aligned and direct comparison is larger in the Southern Hemisphere than in the Northern Hemisphere except for sites located in polluted regions. The a priori alignment uncertainty contribution is significant at several sites as it is larger than the estimated TCCON XCO error. We observe a mean difference of 0.43 ± 4.44 % across all TCCON stations with highest values of -17.43 % for Xianghe (due to very high a priori profile difference). We found that the systematic difference between the S5P CO column and the NDACC CO column data (excluding two stations which were obvious outliers) is on
- average $5.69\pm3.07\%$ (NDACC CO direct comparison), $3.14\pm4.19\%$ (NDACC CO smoothed by using S5P a priori as the common prior), and $6.5\pm3.54\%$ (NDACC CO profile with S5P a priori substituted and additionally smoothed with S5P column averaging kernel). The effect of the smoothing depends on the station location with a mean difference of $0.86\pm2.79\%$ across all NDACC stations and a maximum value of -6.89% in relation to the direct comparison. The effect of smoothing by doing only a priori substitution in relation to the direct comparison gives a mean difference of $-2.49\pm2.96\%$ across all NDACC
- stations and a maximum value of -11.26 %. The comparison with a priori alignment and taking smoothing effects into account is recommended as the preferred method. Most TCCON and NDACC stations show a correlation above 0.9 indicating that the temporal variations in CO column captured by the ground-based instruments are reproduced very similarly by S5P. The

few exceptions are due to the limited data sets available for the comparison. We also found that the S5P random error for the TCCON and NDACC validation results is well below 10% except for few stations where a co-location mismatch occurs

- 810 during certain periods with high values of CO events occurring due to plumes passing over/nearby the stations. A seasonal dependency of the relative bias is seen. We observe a high bias during the high CO event and low bias during the low CO event. We observed a mean difference of 0.09 ± 0.55 % with a maximum difference of 2.56 % for TCCON validation results using the cone co-location criterion compared to the circular co-location criterion. The results of the cone selection criterion at the NDACC stations show higher values than for the TCCON stations. We observe a mean difference of -0.07 ± 0.47 %
- 815 with a maximum difference of -1.35%. The high difference is observed mostly for high latitude stations, where the cone co-location criterion following the line-of-sight of the ground-based FTIR is the best choice in finding co-located satellite pixels for validation. Furthermore, we observed that the validation results of the clear-sky and cloud cases of S5P pixels are in general comparable to the validation results including all pixels passing the filter criteria. The clear-sky or cloud cases are however useful for certain applications. We observe that the relative bias increases with increasing SZA of the measurement.
- 820 We estimated this increase to be 10% over the complete range of measurement SZAs.

Based on the validation results of the S5P operational methane and carbon monoxide data sets against the reference groundbased TCCON and NDACC data sets, we conclude that the S5P methane and carbon monoxide data are of high quality and fulfil the requirements for systematic and random uncertainties.

Appendix A: Reducing a priori and averaging kernel contribution in the validation

- 825 The S5P and ground-based FTIR instruments have different instrument sensitivities and use different a priori profiles to retrieve the best representation of the true atmospheric state from the recorded spectra. The S5P uses a priori derived from the TM5 model, whereas the TCCON uses a daily a priori profile generated by a stand-alone program provided by Toon and Wunch (2017) and NDACC uses a single a priori profile from climatology of the Whole Atmosphere Community Climate Model Version 6 (WACCM V6; ftp://nitrogen.acom.ucar.edu/user/jamesw/IRWG/2013/WACCM/V6/). In order to make the quanti-
- 830 tative comparison, the influence of the a priori contribution to the smoothing equation needs to be compensated/corrected by adjusting the retrieval results to a common a priori (Rodgers and Connor, 2003). The S5P prior is used as the common prior. It is re-gridded to the FTIR grid using a mass conservation algorithm (Langerock et al., 2015). For the case where the satellite pixel elevation is above the ground-based site altitude, the S5P prior profile is extrapolated (i.e., a simple extension, the lowest vmr is taken as the vmr at the lowest ground-based grid) to the altitude of the ground-based instrument. The re-gridded S5P prior $x_{a,S5P}$ is substituted in the FTIR retrieval.
- 835

$$x_{FTIR_mod_prior} = x_{FTIR} + (I - A_{FTIR})(x_{a_S5P} - x_{a_FTIR}),$$
(A1)

where x_{FTIR} is the original vmr profile, $x_{a \ FTIR}$ is the a priori profile used for the original FTIR retrieval (x_{FTIR}), $x_{FTIR mod prior}$ is the corrected FTIR-retrieved profile, A_{FTIR} is the FTIR averaging kernel matrix and I is the unity matrix. This step reduces the total smoothing uncertainty on the column differences by eliminating the uncertainty on the FTIR a priori. Although Eq. A1 is only valid for NDACC profiles, it can be modified to be applied for TCCON column data as well. In that case, the prior profiles should be transformed to partial column profiles and divided by the total column of FTIR dry air.

For NDACC profiles, to further reduce the smoothing uncertainty contribution introduced by the averaging kernel, we smooth the corrected FTIR-retrieved profile ($x_{FTIR_mod_prior}$) with the S5P column averaging kernel (cA_{S5P}). This requires the re-gridding of the corrected FTIR-retrieved profile to the S5P column averaging kernel grid before applying the smoothing equation:

$$c_{FTIR_smoothed} = c_{a_S5P} + cA_{S5P} [(x_{FTIR_mod_prior} - x_{a_S5P}) \times n_{dry,air}],$$
(A2)

where c_{a_S5P} is the column values derived from the S5P a priori profile, $c_{FTIR_smoothed}$ is the smoothed FTIR column associated with a co-located S5P pixel. The n_{dryair} in Eq. A2 is the partial column profile calculated from the pressure difference (ΔP) between the layer interfaces and the hydrostatic equation:

$$850 \quad \Delta P = m_{wet,air} \times n_{wet,air} \times g \tag{A3}$$

For CH₄, the partial column of dry air is available in the S5P Level 2 files. For CO, we derive it using the pressure on the boundaries as described in Eq. A3. In the above Eq. A3, $n_{wet,air}$ is approximated by $n_{dry,air}$ and the molar mass of wet air is approximated by the molar mass of dry air as there is no H_2O profile available in the S5P prior. We found that this approximation has only a small influence, e.g., the bias change at Paramaribo, a tropical site, is about 0.2% when compared

to the case of using NCEP H_2O profile. If the satellite pixel elevation is below the FTIR site altitude, the re-gridding of the corrected FTIR-retrieved profile is done such that the FTIR profile is extended with the S5P a priori profile. This extension of the a priori profile cancels on the right of Eq. A3 and the FTIR smoothed column coincides with the S5P a priori partial column for the region where the grids mismatch.

Appendix B: S5P pixel altitude correction

845

An altitude correction is done for each S5P pixel in order to take into account the altitude difference between the S5P pixels and the ground-based station. The correction can be significant for co-location with mountain stations where the satellite pixels can be picked up from locations around the station, which are at lower or higher altitudes than stations. The scaling factor (f)is calculated from the satellite a priori profile using the following equation:

$$f = \frac{c_{S5P}(FTIR \ altitude \to toa)}{c_{S5P}(S5P \ pixel \ altitude \to toa)},\tag{B1}$$

865 where the numerator is the partial column from the FTIR station altitude to the top of the atmosphere (toa) and the denominator is the total column from the pixel altitude to the top of the atmosphere. The scaling factor is less than 1 for cases where the satellite pixels are located below the altitude of the FTIR station. In certain cases, where the S5P pixels are above the FTIR station, the scaling factor goes above 1. The scaling factor is applied to the satellite data such that the co-located pairs are on the same FTIR station altitude. Equation B1 is valid for satellite pixels < station altitude and we use the S5P prior profile. 870 However, in the other case where satellite pixels > station altitude we extrapolate the satellite prior to compensate the small altitude differences.

The S5P products are adapted to the altitude of the station by either cutting off the scaled mixing ratio profiles at the station altitude (for FTIR station at high altitude locations) or by extending the profile assuming a constant elongation of the mixing ratio up to the station altitude (for case where S5P pixel altitude is above the FTIR station). This method of S5P pixel altitude correction is applied when the satellite and ground-based columns are not calculated between the same boundaries, e.g., S5P

875 correction is applied when the satellite and ground-based columns are not calculated between the same boundaries, e.g., S5F vs. TCCON, and S5P vs. NDACC without extra satellite smoothing.

Data availability. The S5P CH₄ and CO data used in this study are made available to the S5P Mission Performance Centre via the ESA expert Hub. The data since the release of the CO and CH₄ products are available at the Copernicus Open Access Hub (https://scihub.copernicus.eu, last access: 1 December 2020). The FTIR TCCON data are available via the TCCON Data Archive, hosted by CaltechDATA (https://tccondata.org,
last access: 1 June 2021). The FTIR TCCON data without the scaling to the WMO scale were obtained from the site PIs. The FTIR NDACC data can be provided in the public NDACC repository (ftp://ftp.cpc.ncep.noaa.gov/ndacc/station/, last access: 1 June 2021) depending on the site PI's decision.

Author contributions. MKS and BL designed the study and produced the validation analysis and results. MKS wrote the first draft of the paper with support of BL. JL and AL have a joint responsibility for the S5P CH_4 prototype algorithm and operational processor. JL and TB have a joint responsibility for the S5P CO prototype algorithm and operational processor. All authors contributed in the generation of the data used in this study. All authors read the paper and provided comments.

Competing interests. The authors declare that they have no conflict of interest.

885

Acknowledgements. Part of this study has been carried out in the framework of the Copernicus Sentinel-5 Precursor Mission Performance Centre (S5P, MPC) contracted by the European Space Agency (ESA/ESRIN, Contract No. 4000117151/16/I-LG) and supported by the
Belgian Federal Science Policy Office (BELSPO), the Royal Belgian Institute for Space Aeronomy (BIRA-IASB). Part of this study has also been supported by the S5P Validation Team (S5PVT) AO project TCCON4S5P (ID #28603, PI Mahesh Kumar Sha, BIRA-IASB) with national funding from the BELSPO through the ESA ProDEx projects TROVA and TROVA-E2 (PEA 4000116692). This work contains modified Copernicus Sentinel-5 Precursor satellite data (2018–2020) post-processed by BIRA–IASB.

The authors wish to thank the instrument operators and scientists doing the retrieval for the delivery of the ground-based reference data, which were used for the validation study in this paper. We thank Olivier Rasson and José Granville for their support with the creation of S5P overpass level-2 files and download of data used in this study at BIRA-IASB.

The Eureka TCCON and NDACC measurements were made at the Polar Environment Atmospheric Research Laboratory by the Canadian Network for the Detection of Atmospheric Change, primarily supported by the Natural Sciences and Engineering Research Council of Canada

(NSERC), Environment and Climate Change Canada (ECCC), and the Canadian Space Agency. The Toronto NDACC measurements were

- 900 made at the University of Toronto Atmospheric Observatory, supported by NSERC and ECCC. East Trout Lake is funded through grants from the CFI, ORF, NSERC, and the CSA. Funding for the Edwards TCCON stations is provided by NASA's Earth Science Division. Funding for the Darwin, Lamont, and Pasadena TCCON sites is provided by NASA's OCO-2/-3 projects (grants NNN12AA01C and 80NM0018D0004), while the Park Falls site plus TCCON (the network) is funded through NASA Carbon Cycle Science Program (grant NNX17AE15G). Measurements in Ny-Ålesund and Paramaribo are supported by the Alfred Wegener Institute Potsdam. We thank the AWIPEV station
- 905 personnel in Ny-Ålesund, Cornelis Becker and Sukarni Mitro (Meteorologische Dienst Suriname) in Paramaribo and the Senate of Bremen for support. The observations at Ny-Ålesund are supported within the DFG, German Research Foundation Project-ID 268020496 - TRR 172 project (AC)3, Arctic Amplification, subproject E02. The Sodankylä site received financial support from the European Space Agency under the grant agreement number ESA-IPL-POE-LG-cl-LE-2015-1129. The Paris TCCON site has received funding from Sorbonne Université, the French research center CNRS, the French space agency CNES, and Région Île-de-France. TCCON sites at Tsukuba, Rikubetsu and
- 910 Burgos are supported in part by the GOSAT series project. Burgos is supported in part by the Energy Development Corp. Philippines. The TCCON Nicosia site has received additional support from the European Union's Horizon 2020 research and innovation programme under grant agreement No. 856612 and the Cyprus Government, and by the University of Bremen. The Ascension Island TCCON station has been supported by the European Space Agency (ESA) under grant 4000120088/17/I-EF and by the German Bundesministerium für Wirtschaft und Energie (BMWi) under grants 50EE1711C and 50EE1711E. We thank the ESA Ariane Tracking Station at North East Bay, Ascension
- 915 Island, for hosting and local support. The TCCON and NDACC sites at Réunion Island is operated by the Royal Belgian Institute for Space Aeronomy and local activities supported by LACy/UMR8105 and by OSU-R/UMS3365 – Université de La Réunion. The TCCON Réunion Island site is financially supported since 2014 by the EU project ICOS-Inwire and the ministerial decree for ICOS (FR/35/IC1 to FR/35/IC5). The CH₄ and CO measurements at Porto Velho have been supported by the cooperation agreement BAAF and the BRAIN-pioneer project IKARE, respectively, both funded by BELSPO. We thank C. A. Bauer Aquino (IFRO) and L. Gatti (INPE) for their technical support for the

```
920 measurements at Porto Velho.
```

The NDACC data used in this publication were obtained as part of the Network for the Detection of Atmospheric Composition Change (NDACC) and are publicly available (see http://www.ndacc.org). Rapid delivery data on NDACC is supported for selected sites by the CAMS27 project (https://cams27.aeronomie.be). The National Center for Atmospheric Research is sponsored by the National Science Foundation. The NCAR FTS observation programs at Thule, GR, Boulder, CO and Mauna Loa, HI are supported under contract by the National

- 925 Aeronautics and Space Administration (NASA). The Thule work is also supported by the NSF Office of Polar Programs (OPP). We wish to thank the Danish Meteorological Institute for support at the Thule site and NOAA for support of the MLO site. The NDACC and TC-CON stations Ascension Island, Bremen, Garmisch, Izaña, Ny-Ålesund, Paramaribo and Karlsruhe have been supported by the German Bundesministerium für Wirtschaft und Energie (BMWi) via DLR under grants 50EE1711A to E. The FTIR sites Garmisch, Izaña, Karlsruhe, Kiruna, and Zugspitze have been supported by the Helmholtz Society via the research program ATMO. The NDACC/TCCON Izaña data ben-
- 930 efit from the financial support from the Ministerio de Economía y Competitividad from Spain for the project INMENSE (CGL2016-80688-P). We thank the International Foundation High Altitude Research Stations Jungfraujoch and Gornergrat (HFSJG, Bern) for supporting the facilities needed to perform the FTIR observations at Jungfraujoch. The University of Liège contribution has been supported primarily by the Fonds de la Recherche Scientifique FNRS under grant J.0147.18, by the GAW-CH program of MeteoSwiss as well as by the CAMS project. Emmanuel Mahieu is a senior research associate of the F.R.S.-FNRS. The RUOA-network "Red Universitaria de Observatorios Atmosféricos
- 935 de la Universidad Nacional Autónoma de México" are acknowledged for the support of the Altzomoni site. Omar Lopez and Delibes Flores Roman are acknowledged for technical support. Alejandro Bezanilla, Noemie Taquet are acknowledged for helping with measurements and

data processing. For the Altzomoni site UNAM-DGAPA grants IN111418 & IN107417 and CONACYT 290589 are acknowledged. The St.Petersburg site of the NDACC is operated by Saint Petersburg State University with the financial support provided by the Russian Foundation for Basic Research (grant no. 18-05-00011). The scientific equipment is maintained by the Geomodel research center of SPbU. The

940 NDACC site at Rikubetsu is operated as parts of the joint research program of the Institute for Space-Earth Environmental Research (ISEE), Nagoya University and supported in part by the GOSAT series project. The Wollongong NDACC site is funded by the Australia Research Council grant DP160101598. Measurements at Lauder and Arrival Heights are core-funded by NIWA through New Zealand's Ministry of Business, Innovation and Employment Strategic Science Investment Fund. D. S. thanks Antarctica New Zealand for providing support for the measurements at Arrival Heights. 950

- Bader, W., Bovy, B., Conway, S., Strong, K., Smale, D., Turner, A. J., Blumenstock, T., Boone, C., Collaud Coen, M., Coulon, A., Garcia, O., Griffith, D. W. T., Hase, F., Hausmann, P., Jones, N., Krummel, P., Murata, I., Morino, I., Nakajima, H., O'Doherty, S., Paton-Walsh, C., Robinson, J., Sandrin, R., Schneider, M., Servais, C., Sussmann, R., and Mahieu, E.: The recent increase of atmospheric methane from 10 years of ground-based NDACC FTIR observations since 2005, Atmospheric Chemistry and Physics, 17, 2255–2277, https://doi.org/10.5194/acp-17-2255-2017. https://acp.copernicus.org/articles/17/2255/2017/. 2017.
- Batchelor, R. L., Strong, K., Lindenmaier, R., Mittermeier, R. L., Fast, H., Drummond, J. R., and Fogal, P. F.: A New Bruker IFS 125HR FTIR Spectrometer for the Polar Environment Atmospheric Research Laboratory at Eureka, Nunavut, Canada: Measurements and Comparison with the Existing Bomem DA8 Spectrometer, Journal of Atmospheric and Oceanic Technology, 26, 1328 - 1340, https://doi.org/10.1175/2009JTECHA1215.1. https://iournals.ametsoc.org/view/iournals/atot/26/7/2009itecha1215_1.xml, 01 Jul. 2009.
- 955 Baylon, J. L., Stremme, W., Grutter, M., Hase, F., and Blumenstock, T.: Background CO₂ levels and error analysis from ground-based solar absorption IR measurements in central Mexico, Atmospheric Measurement Techniques, 10, 2425-2434, https://doi.org/10.5194/amt-10-2425-2017, https://amt.copernicus.org/articles/10/2425/2017/, 2017.
 - Bergamaschi, P., Frankenberg, C., Meirink, J. F., Krol, M., Dentener, F., Wagner, T., Platt, U., Kaplan, J. O., Körner, S., Heimann, M., Dlugokencky, E. J., and Goede, A.: Satellite chartography of atmospheric methane from SCIAMACHY on board ENVISAT: 2. Evaluation
- 960 based on inverse model simulations, Journal of Geophysical Research: Atmospheres, 112, https://doi.org/10.1029/2006JD007268, https: //agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2006JD007268, 2007.
 - Bezanilla, A., Krüger, A., Stremme, W., and de la Mora, M. G.: Solar absorption infrared spectroscopic measurements over Mexico City: Methane enhancements, Atmósfera, 27, https://www.revistascca.unam.mx/atm/index.php/atm/article/view/43392, 2014.
- Blumenstock, T., Hase, F., Schneider, M., García, O. E., and Sepúlveda, E.: TCCON data from Izana (ES), Release GGG2014.R1, 965 https://doi.org/10.14291/TCCON.GGG2014.IZANA01.R1, https://data.caltech.edu/records/302, 2017.
 - Borsdorff, T., Tol, P., Williams, J. E., de Laat, J., aan de Brugh, J., Nédélec, P., Aben, I., and Landgraf, J.: Carbon monoxide total columns from SCIAMACHY 2.3 µm atmospheric reflectance measurements: towards a full-mission data product (2003–2012), Atmospheric Measurement Techniques, 9, 227-248, https://doi.org/10.5194/amt-9-227-2016, https://amt.copernicus.org/articles/9/227/2016/, 2016.
- Borsdorff, T., aan de Brugh, J., Hu, H., Hasekamp, O., Sussmann, R., Rettinger, M., Hase, F., Gross, J., Schneider, M., Garcia, O., Stremme, 970 W., Grutter, M., Feist, D. G., Arnold, S. G., De Mazière, M., Kumar Sha, M., Pollard, D. F., Kiel, M., Roehl, C., Wennberg, P. O., Toon, G. C., and Landgraf, J.: Mapping carbon monoxide pollution from space down to city scales with daily global coverage. Atmospheric
 - Measurement Techniques, 11, 5507–5518, https://doi.org/10.5194/amt-11-5507-2018, https://www.atmos-meas-tech.net/11/5507/2018/, 2018a.
- Borsdorff, T., Aan de Brugh, J., Hu, H., Aben, I., Hasekamp, O., and Landgraf, J.: Measuring Carbon Monoxide With 975 TROPOMI: First Results and a Comparison With ECMWF-IFS Analysis Data, Geophysical Research Letters, 45, 2826–2832, https://doi.org/https://doi.org/10.1002/2018GL077045, https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1002/2018GL077045, 2018b.
 - Borsdorff, T., aan de Brugh, J., Schneider, A., Lorente, A., Birk, M., Wagner, G., Kivi, R., Hase, F., Feist, D. G., Sussmann, R., Rettinger, M., Wunch, D., Warneke, T., and Landgraf, J.: Improving the TROPOMI CO data product: update of the spectroscopic database and destriping of single orbits, Atmospheric Measurement Techniques, 12, 5443-5455, https://doi.org/10.5194/amt-12-5443-2019, https://
- 980 amt.copernicus.org/articles/12/5443/2019/, 2019.

- Borsdorff, T., García Revnoso, A., Maldonado, G., Mar-Morales, B., Stremme, W., Grutter, M., and Landgraf, J.: Monitoring CO emissions of the metropolis Mexico City using TROPOMI CO observations, Atmospheric Chemistry and Physics, 20, 15761-15774, https://doi.org/10.5194/acp-20-15761-2020, https://acp.copernicus.org/articles/20/15761/2020/, 2020.
- Buchholz, R. R., Deeter, M. N., Worden, H. M., Gille, J., Edwards, D. P., Hannigan, J. W., Jones, N. B., Paton-Walsh, C., Griffith, D. W. T., 985 Smale, D., Robinson, J., Strong, K., Conway, S., Sussmann, R., Hase, F., Blumenstock, T., Mahieu, E., and Langerock, B.: Validation of MOPITT carbon monoxide using ground-based Fourier transform infrared spectrometer data from NDACC, Atmospheric Measurement Techniques, 10, 1927–1956, https://doi.org/10.5194/amt-10-1927-2017, https://www.atmos-meas-tech.net/10/1927/2017/, 2017.
 - De Mazière, M., Sha, M. K., Desmet, F., Hermans, C., Scolas, F., Kumps, N., Metzger, J.-M., Duflot, V., and Cammas, J.-P.: TCCON data from Réunion Island (RE), Release GGG2014.R1, https://doi.org/10.14291/TCCON.GGG2014.REUNION01.R1, https://data.caltech. edu/records/322, 2017.
- 990
 - Deutscher, N. M., Griffith, D. W. T., Bryant, G. W., Wennberg, P. O., Toon, G. C., Washenfelder, R. A., Keppel-Aleks, G., Wunch, D., Yavin, Y., Allen, N. T., Blavier, J.-F., Jiménez, R., Daube, B. C., Bright, A. V., Matross, D. M., Wofsy, S. C., and Park, S.: Total column CO₂ measurements at Darwin, Australia – site description and calibration against in situ aircraft profiles, Atmospheric Measurement Techniques, 3, 947–958, https://doi.org/10.5194/amt-3-947-2010, https://amt.copernicus.org/articles/3/947/2010/, 2010.
- 995 Deutscher, N. M., Notholt, J., Messerschmidt, J., Weinzierl, C., Warneke, T., Petri, C., Grupe, P., and Katrynski, K.: TCCON data from Bialystok (PL), Release GGG2014.R2, https://doi.org/10.14291/tccon.ggg2014.bialystok01.R2, https://data.caltech.edu/records/1300, 2019.
 - Duflot, V., Dils, B., Baray, J. L., De Mazière, M., Attié, J. L., Vanhaelewyn, G., Senten, C., Vigouroux, C., Clain, G., and Delmas, R.: Analysis of the origin of the distribution of CO in the subtropical southern Indian Ocean in 2007, Journal of Geophysical Research: Atmospheres, 115, https://doi.org/10.1029/2010JD013994, https://agupubs.onlinelibrary.wiley.com/doi/abs/10.
- 1000 1029/2010JD013994, 2010.
 - ESA-EOPG-CSCOP-PL: Sentinel-5 Precursor Calibration and Validation Plan for the Operational Phase, https://sentinel.esa.int/documents/ 247904/2474724/Sentinel-5P-Calibration-and-Validation-Plan.pdf, 2017-11-06.
 - Feist, D. G., Arnold, S. G., John, N., and Geibel, M. C.: TCCON data from Ascension Island (SH), Release GGG2014.R0, https://doi.org/10.14291/TCCON.GGG2014.ASCENSION01.R0/1149285, https://data.caltech.edu/records/210, 2014.
- Griffith. D. W., Deutscher, N. M., Velazco, V. A., Wennberg, P. O., Yavin, Y., Keppel-Aleks, G., Washenfelder, R. A., 1005 Toon, G. C., Blavier, J.-F., Paton-Walsh, C., Jones, N. B., Kettlewell, G. C., Connor, B. J., Macatangay, R. C., Roehl, C., Ryczek, M., Glowacki, J., Culgan, T., and Bryant, G. W.: TCCON data from Darwin (AU), Release GGG2014.R0, https://doi.org/10.14291/TCCON.GGG2014.DARWIN01.R0/1149290, https://data.caltech.edu/records/269, 2014a.
- Griffith, D. W., Velazco, V. A., Deutscher, N. M., Paton-Walsh, C., Jones, N. B., Wilson, S. R., Macatangay, R. C., Ket-1010 tlewell, G. C., Buchholz, R. R., and Riggenbach, M. O.: TCCON data from Wollongong (AU), Release GGG2014.R0, https://doi.org/10.14291/TCCON.GGG2014.WOLLONGONG01.R0/1149291, https://data.caltech.edu/records/291, 2014b.
 - Hannigan, J. W., Coffey, M. T., and Goldman, A.: Semiautonomous FTS Observation System for Remote Sensing of Stratospheric and Tropospheric Gases, Journal of Atmospheric and Oceanic Technology, 26, 1814 - 1828, https://doi.org/10.1175/2009JTECHA1230.1, https://journals.ametsoc.org/view/journals/atot/26/9/2009jtecha1230_1.xml, 01 Sep. 2009.
- 1015 Hase, F., Hannigan, J., Coffey, M., Goldman, A., Höpfner, M., Jones, N., Rinsland, C., and Wood, S.: Intercomparison of retrieval codes used for the analysis of high-resolution, ground-based FTIR measurements, Journal of Quantitative Spectroscopy and Radiative Transfer, 87, 25 - 52, https://doi.org/https://doi.org/10.1016/j.jqsrt.2003.12.008, http://www.sciencedirect.com/science/article/pii/S0022407303003765, 2004.

Hase, F., Blumenstock, T., Dohe, S., Groß, J., and Kiel, M.: TCCON data from Karlsruhe (DE), Release GGG2014.R1,
https://doi.org/10.14291/TCCON.GGG2014.KARLSRUHE01.R1/1182416, https://data.caltech.edu/records/278, 2015.

- Hasekamp, O., Lorente, A., Hu, H., Butz, A., aan de Brugh, J., and Landgraf, J.: Algorithm Theoretical Baseline Document for Sentinel-5 Precursor Methane Retrieval, https://sentinel.esa.int/documents/247904/2476257/Sentinel-5P-TROPOMI-ATBD-Methane-retrieval, 2019.
 - Hedelius, J. K., He, T.-L., Jones, D. B. A., Baier, B. C., Buchholz, R. R., De Mazière, M., Deutscher, N. M., Dubey, M. K., Feist, D. G., Griffith, D. W. T., Hase, F., Iraci, L. T., Jeseck, P., Kiel, M., Kivi, R., Liu, C., Morino, I., Notholt, J., Oh, Y.-S., Ohyama, H., Pollard,
- 1025 D. F., Rettinger, M., Roche, S., Roehl, C. M., Schneider, M., Shiomi, K., Strong, K., Sussmann, R., Sweeney, C., Té, Y., Uchino, O., Velazco, V. A., Wang, W., Warneke, T., Wennberg, P. O., Worden, H. M., and Wunch, D.: Evaluation of MOPITT Version 7 joint TIR– NIR X_{CO} retrievals with TCCON, Atmospheric Measurement Techniques, 12, 5547–5572, https://doi.org/10.5194/amt-12-5547-2019, https://www.atmos-meas-tech.net/12/5547/2019/, 2019.
 - Hochstaffl, P., Schreier, F., Lichtenberg, G., and Gimeno García, S.: Validation of Carbon Monoxide Total Column Re-
- 1030 trievals from SCIAMACHY Observations with NDACC/TCCON Ground-Based Measurements, Remote Sensing, 10 (2), https://doi.org/https://doi.org/10.3390/rs10020223, 2018.
 - Hu, H., Hasekamp, O., Butz, A., Galli, A., Landgraf, J., Aan de Brugh, J., Borsdorff, T., Scheepmaker, R., and Aben, I.: The operational methane retrieval algorithm for TROPOMI, Atmospheric Measurement Techniques, 9, 5423–5440, https://doi.org/10.5194/amt-9-5423-2016, https://amt.copernicus.org/articles/9/5423/2016/, 2016.
- 1035 Iraci, L. T., Podolske, J. R., Hillyard, P. W., Roehl, C., Wennberg, P. O., Blavier, J.-F., Landeros, J., Allen, N., Wunch, D., Zavaleta, J., Quigley, E., Osterman, G. B., Albertson, R., Dunwoody, K., and Boyden, H.: TCCON data from Edwards (US), Release GGG2014.R1, https://doi.org/10.14291/TCCON.GGG2014.EDWARDS01.R1/1255068, https://data.caltech.edu/records/270, 2016.
 - Iwasaki, C., Imasu, R., Bril, A., Yokota, T., Yoshida, Y., Morino, I., Oshchepkov, S., Wunch, D., Griffith, D. W. T., Deutscher, N. M., Kivi, R., Pollard, D., Roehl, C. M., Velazco, V. A., Sussmann, R., Warneke, T., and Notholt, J.: Validation of GOSAT SWIR
- 1040 XCO₂ and XCH₄ Retrieved by PPDF-S Method and Comparison with Full Physics Method, SOLA, 13, 168– 173, https://doi.org/10.2151/sola.2017-031, 2017.
 - Kawakami, S., Ohyama, H., Arai, K., Okumura, H., Taura, C., Fukamachi, T., and Sakashita, M.: TCCON data from Saga (JP), Release GGG2014.R0, https://doi.org/10.14291/TCCON.GGG2014.SAGA01.R0/1149283, https://data.caltech.edu/records/288, 2014.
 - Kiel, M., Hase, F., Blumenstock, T., and Kirner, O.: Comparison of XCO abundances from the Total Carbon Column Observing Network
- and the Network for the Detection of Atmospheric Composition Change measured in Karlsruhe, Atmospheric Measurement Techniques,
 9, 2223–2239, https://doi.org/10.5194/amt-9-2223-2016, https://amt.copernicus.org/articles/9/2223/2016/, 2016.
 - Kirschke, S., Bousquet, P., Ciais, P., Saunois, M., Canadell, J. G., Dlugokencky, E. J., Bergamaschi, P., Bergmann, D., Blake, D. R., Bruhwiler, L., Cameron-Smith, P., Castaldi, S., Chevallier, F., Feng, L., Fraser, A., Heimann, M., Hodson, E. L., Houweling, S., Josse, B., Fraser, P. J., Krummel, P. B., Lamarque, J.-F., Langenfelds, R. L., Le Quéré, C., Naik, V., O'Doherty, S., Palmer, P. I., Pison, I., Plummer, D., Poulter,
- 1050 B., Prinn, R. G., Rigby, M., Ringeval, B., Santini, M., Schmidt, M., Shindell, D. T., Simpson, I. J., Spahni, R., Steele, L. P., Strode, S. A., Sudo, K., Szopa, S., van der Werf, G. R., Voulgarakis, A., van Weele, M., Weiss, R. F., Williams, J. E., and Zeng, G.: Three decades of global methane sources and sinks, Nat. Geosci., https://doi.org/doi:10.1038/ngeo1955, 2013.
 - Kivi, R. and Heikkinen, P.: Fourier transform spectrometer measurements of column CO₂ at Sodankylä, Finland, Geoscientific Instrumentation, Methods and Data Systems, 5, 271–279, https://doi.org/10.5194/gi-5-271-2016, https://gi.copernicus.org/articles/5/271/2016/, 2016.
- 1055 E.: Kivi, R., Heikkinen, P., and Kyrö, TCCON data from Sodankylä (FI), Release GGG2014.R0, https://doi.org/10.14291/TCCON.GGG2014.SODANKYLA01.R0/1149280, https://data.caltech.edu/records/289, 2014.

Kulawik, S. S., Wunch, D., O'Dell, C., Frankenberg, C., Reuter, M., Oda, T., Chevallier, F., Sherlock, V., Buchwitz, M., Osterman, G., Miller, C., Wennberg, P., Griffith, D. W. T., Morino, I., Dubey, M., Deutscher, N. M., Notholt, J., Hase, F., Warneke, T., Sussmann, R., Robinson, J., Strong, K., Schneider, M., De Maziere, M., Shiomi, K., Feist, D. G., Iraci, L. T., and Wolf, J.: Consistent evaluation of ACOS-

- 1060 GOSAT, BESD-SCIAMACHY, CarbonTracker, and MACC through comparisons to TCCON, Atmospheric Measurement Techniques, https://doi.org/doi:10.5194/amt-9-683-2016, 2016.
 - Lambert, J.-C., Compernolle, S., Eichmann, K.-U., de Graaf, M., Hubert, D., Keppens, A., Kleipool, Q., Langerock, B., Sha, M. K., Verhoelst, T., Wagner, T., Ahn, C., Argyrouli, A., Balis, D., Chan, K., De Smedt, I., Eskes, H., Fjæraa, A., Garane, K., Gleason, J., Goutail, F., Granville, J., Hedelt, P. Heue, K.-P., Jaross, G., Koukouli, M., Landgraf, J., Lutz, R., Nanda, S., Niemejer, S., Pazmiño,
- 1065 A., Pinardi, G., Pommereau, J.-P., Richter, A., Rozemeijer, N., Sneep, M., SteinZweers, D., Theys, N., Tilstra, G., Torres, O., Valks, P., van Geffen, J., Vigouroux, C., Wang, P., and Weber, M.: Quarterly Validation Report of the CopernicusSentinel-5 PrecursorOperational Data Products#08:April2018–August2020., https://mpc-vdaf.tropomi.eu/ProjectDir/reports/pdf/S5P-MPC-IASB-ROCVR-08.01. 01-20200921_FINAL.pdf, 2020.
 - Landgraf, J., aan de Brugh, J., Scheepmaker, R., Borsdorff, T., Hu, H., Houweling, S., Butz, A., Aben, I., and Hasekamp, O.: Carbon
- 1070 monoxide total column retrievals from TROPOMI shortwave infrared measurements, Atmospheric Measurement Techniques, 9, 4955–
 4975, https://doi.org/10.5194/amt-9-4955-2016, https://www.atmos-meas-tech.net/9/4955/2016/, 2016.
 - Landgraf, J., aan de Brugh, J., Scheepmaker, R., Borsdorff, T., Houweling, S., and Hasekamp, O.: Algorithm Theoretical Baseline Document for Sentinel-5 Precursor: Carbon Monoxide Total Column Retrieval, https://sentinel.esa.int/documents/247904/2476257/ Sentinel-5P-TROPOMI-ATBD-Carbon-Monoxide-Total-Column-Retrieval, 2018.
- 1075 Langerock, B., De Mazière, M., Hendrick, F., Vigouroux, C., Desmet, F., Dils, B., and Niemeijer, S.: Description of algorithms for co-locating and comparing gridded model data with remote-sensing observations, Geoscientific Model Development, 8, 911–921, https://doi.org/10.5194/gmd-8-911-2015, https://gmd.copernicus.org/articles/8/911/2015/, 2015.
 - Lorente, A., Borsdorff, T., Butz, A., Hasekamp, O., aan de Brugh, J., Schneider, A., Wu, L., Hase, F., Kivi, R., Wunch, D., Pollard, D. F., Shiomi, K., Deutscher, N. M., Velazco, V. A., Roehl, C. M., Wennberg, P. O., Warneke, T., and Landgraf, J.: Methane retrieved from
- 1080 TROPOMI: improvement of the data product and validation of the first 2 years of measurements, Atmospheric Measurement Techniques, 14, 665–684, https://doi.org/10.5194/amt-14-665-2021, https://amt.copernicus.org/articles/14/665/2021/, 2021.
 - Makarova, M. V., Kirner, O., Timofeev, Y. M., Poberovskii, A. V., Imkhasin, K. K., Osipov, S. I., and Makarov, B. K.: Annual cycle and long-term trend of the methane total column in the atmosphere over the St. Petersburg region, Izvestiya, Atmospheric and Oceanic Physics, 51, 431–438, https://doi.org/10.1134/S0001433815040088, 2015.
- 1085 Morino, I., Matsuzaki, T., and Horikawa, M.: TCCON data from Tsukuba (JP), 125HR, Release GGG2014.R2, https://doi.org/10.14291/TCCON.GGG2014.TSUKUBA02.R2, https://data.caltech.edu/records/958, 2018a.
 - Morino, I., Velazco, V. A., Hori, A., Uchino, O., and Griffith, D. W.: TCCON data from Burgos, Ilocos Norte (PH), Release GGG2014.R0, https://doi.org/10.14291/TCCON.GGG2014.BURGOS01.R0, https://data.caltech.edu/records/1090, 2018b.
- Morino, I., Yokozeki, N., Matsuzaki, T., and Horikawa, M.: TCCON data from Rikubetsu (JP), Release GGG2014.R2, https://doi.org/10.14291/TCCON.GGG2014.RIKUBETSU01.R2, https://data.caltech.edu/records/957, 2018c.
 - Notholt, J., Petri, C., Warneke, T., Deutscher, N. M., Palm, M., Buschmann, M., Weinzierl, C., Macatangay, R. C., and Grupe, P.: TCCON data from Bremen (DE), Release GGG2014.R0, https://doi.org/10.14291/TCCON.GGG2014.BREMEN01.R0/1149275, https://data.caltech.edu/records/268, 2014a.

Notholt, J., Warneke, T., Petri, C., Deutscher, N. M., Weinzierl, C., Palm, M., and Buschmann, M.: TCCON data from Ny Ålesund, Spitsber-

- 1095 gen (NO), Release GGG2014.R0, https://doi.org/10.14291/TCCON.GGG2014.NYALESUND01.R0/1149278, https://data.caltech.edu/ records/301, 2014b.
 - Novelli, P. C., Masarie, K. A., and Lang, P. M.: Distributions and recent changes of carbon monoxide in the lower troposphere, Journal of Geophysical Research: Atmospheres, 103, 19015–19033, https://doi.org/10.1029/98JD01366, https://doi.org/10.1029/98JD01366, 1998.
 O'Dell, C. W., Eldering, A., Wennberg, P. O., Crisp, D., Gunson, M. R., Fisher, B., Frankenberg, C., Kiel, M., Lindqvist, H., Mandrake,
- L., Merrelli, A., Natraj, V., Nelson, R. R., Osterman, G. B., Payne, V. H., Taylor, T. E., Wunch, D., Drouin, B. J., Oyafuso, F., Chang, A., McDuffie, J., Smyth, M., Baker, D. F., Basu, S., Chevallier, F., Crowell, S. M. R., Feng, L., Palmer, P. I., Dubey, M., García, O. E., Griffith, D. W. T., Hase, F., Iraci, L. T., Kivi, R., Morino, I., Notholt, J., Ohyama, H., Petri, C., Roehl, C. M., Sha, M. K., Strong, K., Sussmann, R., Te, Y., Uchino, O., and Velazco, V. A.: Improved retrievals of carbon dioxide from Orbiting Carbon Observatory-2 with the version 8 ACOS algorithm, Atmospheric Measurement Techniques, 11, 6539–6576, https://doi.org/10.5194/amt-11-6539-2018,
- 1105 https://www.atmos-meas-tech.net/11/6539/2018/, 2018.
 - Olsen, K. S., Strong, K., Walker, K. A., Boone, C. D., Raspollini, P., Plieninger, J., Bader, W., Conway, S., Grutter, M., Hannigan, J. W., Hase, F., Jones, N., de Mazière, M., Notholt, J., Schneider, M., Smale, D., Sussmann, R., and Saitoh, N.: Comparison of the GOSAT TANSO-FTS TIR CH₄ volume mixing ratio vertical profiles with those measured by ACE-FTS, ESA MIPAS, IMK-IAA MIPAS, and 16 NDACC stations, Atmospheric Measurement Techniques, 10, 3697–3718, https://doi.org/10.5194/amt-10-3697-2017, https://amt.copernicus.org/
- 1110 articles/10/3697/2017/, 2017.
 - Ortega, I., Buchholz, R. R., Hall, E. G., Hurst, D. F., Jordan, A. F., and Hannigan, J. W.: Tropospheric water vapor profiles obtained with FTIR: comparison with balloon-borne frost point hygrometers and influence on trace gas retrievals, Atmospheric Measurement Techniques, 12, 873–890, https://doi.org/10.5194/amt-12-873-2019, https://www.atmos-meas-tech.net/12/873/2019/, 2019.
- Ostler, A., Sussmann, R., Rettinger, M., Deutscher, N. M., Dohe, S., Hase, F., Jones, N., Palm, M., and Sinnhuber, B.-M.: Multistation intercomparison of column-averaged methane from NDACC and TCCON: impact of dynamical variability, Atmospheric Measurement
- Techniques, 7, 4081–4101, https://doi.org/10.5194/amt-7-4081-2014, https://www.atmos-meas-tech.net/7/4081/2014/, 2014.
 - Petri, C., Rousogenous, C., Warneke, T., Vrekoussis, M., Sciare, J., and Notholt, J.: TCCON data from Nicosia (Cyprus), Release GGG2014.R0, https://doi.org/10.14291/tccon.ggg2014.nicosia01.R0, 2019.
 - Plaza-Medina, E. F., Stremme, W., Bezanilla, A., Grutter, M., Schneider, M., Hase, F., and Blumenstock, T.: Ground-based remote sensing
- of O₃ by high- and medium-resolution FTIR spectrometers over the Mexico City basin, Atmospheric Measurement Techniques, 10, 2703–2725, https://doi.org/10.5194/amt-10-2703-2017, https://amt.copernicus.org/articles/10/2703/2017/, 2017.
 - Pollard, D. F., Robinson, J., and Shiona, H.: TCCON data from Lauder (NZ), Release GGG2014.R0, https://doi.org/10.14291/TCCON.GGG2014.LAUDER03.R0, https://data.caltech.edu/records/1220, 2019.
 - Pougatchev, N. S., Connor, B. J., and Rinsland, C. P.: Infrared measurements of the ozone vertical distribution above Kitt Peak, Journal of
- 1125 Geophysical Research: Atmospheres, 100, 16 689–16 697, https://doi.org/10.1029/95JD01296, https://agupubs.onlinelibrary.wiley.com/ doi/abs/10.1029/95JD01296, 1995.
 - Rodgers, C. D. and Connor, B. J.: Intercomparison of remote sounding instruments, Journal of Geophysical Research: Atmospheres, 108, https://doi.org/10.1029/2002JD002299, https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2002JD002299, 2003.
- Schneising, O., Buchwitz, M., Reuter, M., Bovensmann, H., Burrows, J. P., Borsdorff, T., Deutscher, N. M., Feist, D. G., Griffith, D. W. T.,
 Hase, F., Hermans, C., Iraci, L. T., Kivi, R., Landgraf, J., Morino, I., Notholt, J., Petri, C., Pollard, D. F., Roche, S., Shiomi, K., Strong, K.,
 Sussmann, R., Velazco, V. A., Warneke, T., and Wunch, D.: A scientific algorithm to simultaneously retrieve carbon monoxide and methane

from TROPOMI onboard Sentinel-5 Precursor, Atmospheric Measurement Techniques, 12, 6771–6802, https://doi.org/10.5194/amt-12-6771-2019, https://amt.copernicus.org/articles/12/6771/2019/, 2019.

Schneising, O., Buchwitz, M., Reuter, M., Bovensmann, H., and Burrows, J. P.: Severe Californian wildfires in November 2018 observed from

- 1135 space: the carbon monoxide perspective, Atmospheric Chemistry and Physics, 20, 3317–3332, https://doi.org/10.5194/acp-20-3317-2020, https://acp.copernicus.org/articles/20/3317/2020/, 2020a.
 - Schneising, O., Buchwitz, M., Reuter, M., Vanselow, S., Bovensmann, H., and Burrows, J. P.: Remote sensing of methane leakage from natural gas and petroleum systems revisited, Atmospheric Chemistry and Physics, 20, 9169–9182, https://doi.org/10.5194/acp-20-9169-2020, https://acp.copernicus.org/articles/20/9169/2020/, 2020b.
- 1140 Sepúlveda, E., Schneider, M., Hase, F., García, O. E., Gomez-Pelaez, A., Dohe, S., Blumenstock, T., and Guerra, J. C.: Long-term validation of tropospheric column-averaged CH₄ mole fractions obtained by mid-infrared ground-based FTIR spectrometry, Atmospheric Measurement Techniques, 5, 1425–1441, https://doi.org/10.5194/amt-5-1425-2012, https://amt.copernicus.org/articles/5/1425/2012/, 2012.
 - Sepúlveda, E., Schneider, M., Hase, F., Barthlott, S., Dubravica, D., García, O. E., Gomez-Pelaez, A., González, Y., Guerra, J. C., Gisi, M., Kohlhepp, R., Dohe, S., Blumenstock, T., Strong, K., Weaver, D., Palm, M., Sadeghi, A., Deutscher, N. M., Warneke, T., Notholt, J., Jones,
- 1145 N., Griffith, D. W. T., Smale, D., Brailsford, G. W., Robinson, J., Meinhardt, F., Steinbacher, M., Aalto, T., and Worthy, D.: Tropospheric CH₄ signals as observed by NDACC FTIR at globally distributed sites and comparison to GAW surface in situ measurements, Atmospheric Measurement Techniques, 7, 2337–2360, https://doi.org/10.5194/amt-7-2337-2014, https://amt.copernicus.org/articles/7/2337/ 2014/, 2014.
 - Sha, M. K., Langerock, B., De Mazière, M., Dils, B., Feist, D. G., Sussmann, R., Hase, F., Schneider, M., Blumenstock, T., Notholt, J.,
- Warneke, T., Petri, C., Kivi, R., Te, Y., Wennberg, P. O., Wunch, D., Iraci, L. T., Strong, K., Griffith, D. W. T., Deutscher, N. M., Velazco, V. A., Morino, I., Ohyama, H., Uchino, O., Shiomi, K., Goo, T. Y., Pollard, D. F., Borsdorff, T., Hu, H., Hasekamp, O. P., Landgraf, J., Roehl, C. M., Kiel, M., Toon, G., and TCCON team: First results of the Sentinel-5 Precursor CO and CH4 validation using TCCON data (ESA AO project TCCON4S5P) and NDACC CO data, in: Annual Joint NDACCIRWG and TCCON Meeting, Mexico City, Mexico, 11 15 June 2018, https://sites.google.com/site/irwgtccon2018/, 2018a.
- 1155 Sha, M. K., Langerock, B., De Mazière, M., Dils, B., Feist, D. G., Sussmann, R., Hase, F., Schneider, M., Blumenstock, T., Notholt, J., Warneke, T., Petri, C., Kivi, R., Te, Y., Wennberg, P. O., Wunch, D., Iraci, L. T., Strong, K., Griffith, D. W. T., Deutscher, N. M., Velazco, V. A., Morino, I., Ohyama, H., Uchino, O., Shiomi, K., Goo, T. Y., Pollard, D. F., Borsdorff, T., Hu, H., Hasekamp, O. P., Landgraf, J., Roehl, C. M., Kiel, M., Toon, G., TCCON team, and NDACC team: VDAF CO validation results & ESA AO # 28603 project TCCON4S5P first CO validation results using TCCON and NDACC data, in: S5P First products release workshop, ESA-Esrin, Frascati (Rome), Italy,
- 25 26 June, https://nikal.eventsair.com/QuickEventWebsitePortal/sentinel-5p-first-product-release-workshop/sentinel-5p, 2018b.
 Sha, M. K., De Mazière, M., Notholt, J., Blumenstock, T., Chen, H., Dehn, A., Griffith, D. W. T., Hase, F., Heikkinen, P., Hermans, C.,
- Hoffmann, A., Huebner, M., Jones, N., Kivi, R., Langerock, B., Petri, C., Scolas, F., Tu, Q., and Weidmann, D.: Intercomparison of lowand high-resolution infrared spectrometers for ground-based solar remote sensing measurements of total column concentrations of CO₂, CH₄, and CO, Atmospheric Measurement Techniques, 13, 4791–4839, https://doi.org/10.5194/amt-13-4791-2020, https://amt.copernicus.
 org/articles/13/4791/2020/, 2020.
- Sherlock, V., Connor, B., Robinson, J., Shiona, H., Smale, D., and Pollard, D. F.: TCCON data from Lauder (NZ), 125HR, Release GGG2014.R0, https://doi.org/10.14291/TCCON.GGG2014.LAUDER02.R0/1149298, https://data.caltech.edu/records/281, 2014.
 - Stocker, T. F., Qin, D., Plattner, G.-K., Alexander, L., Allen, S., Bindoff, N., Bréon, F.-M., Church, J., Cubasch, U., Emori, S., Forster, P., Friedlingstein, P., Gillett, N., Gregory, J., Hartmann, D., Jansen, E., Kirtman, B., Knutti, R., Kumar, K. K., Lemke, P., Marotzke, J.,
- 1170 Masson-Delmotte, V., Meehl, G., Mokhov, I., S.Piao, Ramaswamy, V., Randall, D., Rhein, M., Rojas, M., Sabine, C., Shindell, D., Talley, L., Vaughan, D., and Xie, S.-P.: Technical Summary, in: Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Stocker, T.F., D. Qin, G.-K. Plattner, M. Tignor, S.K. Allen, J. Boschung, A. Nauels, Y. Xia, V. Bex and P.M. Midgley (eds.)], Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA, 2013.
- 1175 Stremme, W., Grutter, M., Rivera, C., Bezanilla, A., Garcia, A. R., Ortega, I., George, M., Clerbaux, C., Coheur, P.-F., Hurtmans, D., Hannigan, J. W., and Coffey, M. T.: Top-down estimation of carbon monoxide emissions from the Mexico Megacity based on FTIR measurements from ground and space, Atmospheric Chemistry and Physics, 13, 1357–1376, https://doi.org/10.5194/acp-13-1357-2013, https://acp.copernicus.org/articles/13/1357/2013/, 2013.
- Strong, K., Roche, S., Franklin, J. E., Mendonca, J., Lutsch, E., Weaver, D., Fogal, P. F., Drummond, J. R., Batchelor, R., and Lindenmaier,
 R.: TCCON data from Eureka (CA), Release GGG2014.R3, https://doi.org/10.14291/TCCON.GGG2014.EUREKA01.R3, https://data.
 caltech.edu/records/1171, 2019.
 - Sussmann, R. and Rettinger, M.: TCCON data from Garmisch (DE), Release GGG2014.R2, https://doi.org/10.14291/TCCON.GGG2014.GARMISCH01.R2, https://data.caltech.edu/records/956, 2018a.
- Sussmann, R. and Rettinger, M.: TCCON data from Zugspitze (DE), Release GGG2014.R1, https://doi.org/10.14291/TCCON.GGG2014.ZUGSPITZE01.R1, https://data.caltech.edu/records/923, 2018b.
 - Té, Y., Jeseck, P., and Janssen, C.: TCCON data from Paris (FR), Release GGG2014.R0, https://doi.org/10.14291/TCCON.GGG2014.PARIS01.R0/1149279, https://data.caltech.edu/records/284, 2014.
 - Té, Y., Jeseck, P., Franco, B., Mahieu, E., Jones, N., Paton-Walsh, C., Griffith, D. W. T., Buchholz, R. R., Hadji-Lazaro, J., Hurtmans, D., and Janssen, C.: Seasonal variability of surface and column carbon monoxide over the megacity Paris, high-altitude Jungfraujoch and Southern
- Hemispheric Wollongong stations, Atmospheric Chemistry and Physics, 16, 10911–10925, https://doi.org/10.5194/acp-16-10911-2016, https://acp.copernicus.org/articles/16/10911/2016/, 2016.
 - GGG2014 Toon. G. C. and Wunch, D.: Α stand-alone а priori profile generation tool for release. https://doi.org/10.14291/TCCON.GGG2014.PRIORS.R0/1221661, 2017.
- Veefkind, J., Aben, I., McMullan, K., Förster, H., de Vries, J., Otter, G., Claas, J., Eskes, H., de Haan, J., Kleipool, Q., van
 Weele, M., Hasekamp, O., Hoogeveen, R., Landgraf, J., Snel, R., Tol, P., Ingmann, P., Voors, R., Kruizinga, B., Vink, R., Visser, H., and Levelt, P.: TROPOMI on the ESA Sentinel-5 Precursor: A GMES mission for global observations of the atmospheric composition for climate, air quality and ozone layer applications, Remote Sensing of Environment, 120, 70 83, https://doi.org/https://doi.org/10.1016/j.rse.2011.09.027, http://www.sciencedirect.com/science/article/pii/S0034425712000661, the Sentinel Missions New Opportunities for Science, 2012.
- 1200 Velazco, V., Morino, I., Uchino, O., Hori, A., Kiel, M., Bukosa, B., Deutscher, N., Sakai, T., Nagai, T., Bagtasa, G., Izumi, T., Yoshida, Y., and Griffith, D.: TCCON Philippines: First Measurement Results, Satellite Data and Model Comparisons in Southeast Asia, Remote Sensing, 9, 1228, https://doi.org/10.3390/rs9121228, 2017.
 - Vellalassery, A., Pillai, D., Marshall, J., Gerbig, C., Buchwitz, M., Schneising, O., and Ravi, A.: Using TROPOspheric Monitoring Instrument (TROPOMI) measurements and Weather Research and Forecasting (WRF) CO modelling to understand the contribution of meteorology
- 1205 and emissions to an extreme air pollution event in India, Atmospheric Chemistry and Physics, 21, 5393–5414, https://doi.org/10.5194/acp-21-5393-2021, https://acp.copernicus.org/articles/21/5393/2021/, 2021.

- Warneke, T., Messerschmidt, J., Notholt, J., Weinzierl, C., Deutscher, N. M., Petri, C., and Grupe, P.: TCCON data from Orléans (FR), Release GGG2014.R1, https://doi.org/10.14291/TCCON.GGG2014.ORLEANS01.R1, https://data.caltech.edu/records/1301, 2019.
- Wennberg, P. O., Wunch, D., Roehl, C. M., Blavier, J.-F., Toon, G. C., and Allen, N. T.: TCCON data from Caltech (US), Release
 GGG2014.R1, https://doi.org/10.14291/TCCON.GGG2014.PASADENA01.R1/1182415, https://data.caltech.edu/records/285, 2015.
 - Wennberg, P. O., Roehl, C. M., Blavier, J.-F., Wunch, D., and Allen, N. T.: TCCON data from Jet Propulsion Laboratory (US), 2011, Release GGG2014.R1, https://doi.org/10.14291/TCCON.GGG2014.JPL02.R1/1330096, https://data.caltech.edu/records/277, 2016a.
 - Wennberg, P. O., Wunch, D., Roehl, C. M., Blavier, J.-F., Toon, G. C., and Allen, N. T.: TCCON data from Lamont (US), Release GGG2014.R1, https://doi.org/10.14291/TCCON.GGG2014.LAMONT01.R1/1255070, https://data.caltech.edu/records/279, 2016b.
- 1215 Wennberg, P. O., Roehl, C. M., Wunch, D., Toon, G. C., Blavier, J.-F., Washenfelder, R., Keppel-Aleks, G., Allen, N. T., and Ayers, J.: TCCON data from Park Falls (US), Release GGG2014.R1, https://doi.org/10.14291/TCCON.GGG2014.PARKFALLS01.R1, https://data. caltech.edu/records/295, 2017.
 - Wiacek, A., Taylor, J. R., Strong, K., Saari, R., Kerzenmacher, T. E., Jones, N. B., and Griffith, D. W. T.: Ground-Based Solar Absorption FTIR Spectroscopy: Characterization of Retrievals and First Results from a Novel Optical Design Instrument at a New
- 1220 NDACC Complementary Station, Journal of Atmospheric and Oceanic Technology, 24, 432 448, https://doi.org/10.1175/JTECH1962.1, https://journals.ametsoc.org/view/journals/atot/24/3/jtech1962_1.xml, 01 Mar. 2007.
 - Wunch, D., Toon, G. C., Blavier, J.-F. L., Washenfelder, R. A., Notholt, J., Connor, B. J., Griffith, D. W. T., Sherlock, V., and Wennberg, P. O.: The Total Carbon Column Observing Network, 369, 2087–2112, https://doi.org/10.1098/rsta.2010.0240, 2011.
 - Wunch, D., Toon, G. C., Sherlock, V., Deutscher, N. M., Liu, C., Feist, D. G., and Wennberg, P. O.: The Total Carbon Column Observing
- 1225 Network's GGG2014 Data Version, Tech. rep., Carbon Dioxide Information Analysis Center, Oak Ridge National Laboratory, Oak Ridge, Tennessee, U.S.A., https://doi.org/10.14291/tccon.ggg2014.documentation.R0/1221662, 2015.
 - Wunch, D., Wennberg, P. O., Osterman, G., Fisher, B., Naylor, B., Roehl, C. M., O'Dell, C., Mandrake, L., Viatte, C., Griffith, D. W., Deutscher, N. M., Velazco, V. A., Notholt, J., Warneke, T., Petri, C., De Maziere, M., Sha, M. K., Sussmann, R., Rettinger, M., Pollard, D., Robinson, J., Morino, I., Uchino, O., Hase, F., Blumenstock, T., Kiel, M., Feist, D. G., Arnold, S. G., Strong, K., Mendonca, J.,
- Kivi, R., Heikkinen, P., Iraci, L., Podolske, J., Hillyard, P. W., Kawakami, S., Dubey, M. K., Parker, H. A., Sepulveda, E., Rodriguez, O. E. G., Te, Y., Jeseck, P., Gunson, M. R., Crisp, D., and Eldering, A.: Comparisons of the Orbiting Carbon Observatory-2 (OCO-2) X_{CO2} measurements with TCCON, Atmospheric Measurement Techniques Discussions, 2016, 1–45, https://doi.org/10.5194/amt-2016-227, http://www.atmos-meas-tech-discuss.net/amt-2016-227/, 2016.
- Wunch, D., Mendonca, J., Colebatch, O., Allen, N. T., Blavier, J.-F., Roche, S., Hedelius, J., Neufeld, G., Springett,
 S., Worthy, D., Kessler, R., and Strong, K.: TCCON data from East Trout Lake, SK (CA), Release GGG2014.R1, https://doi.org/10.14291/TCCON.GGG2014.EASTTROUTLAKE01.R1, https://data.caltech.edu/records/362, 2018.
 - Yang, Y., Zhou, M., Langerock, B., Sha, M., Hermans, C., Wang, T., Ji, D., Vigouroux, C., Kumps, N., Wang, G., De Mazière, M., , and Wang, P.: Ground-based FTIR CO2, CH4 and CO measurements at Xianghe, China, Royal Belgian Institute for Space Aeronomy, https://doi.org/https://doi.org/10.18758/71021049, 2019.
- 1240 Yurganov, L. N., Blumenstock, T., Grechko, E. I., Hase, F., Hyer, E. J., Kasischke, E. S., Koike, M., Kondo, Y., Kramer, I., Leung, F.-Y., Mahieu, E., Mellqvist, J., Notholt, J., Novelli, P. C., Rinsland, C. P., Scheel, H. E., Schulz, A., Strandberg, A., Sussmann, R., Tanimoto, H., Velazco, V., Zander, R., and Zhao, Y.: A quantitative assessment of the 1998 carbon monoxide emission anomaly in the Northern Hemisphere based on total column and surface concentration measurements, Journal of Geophysical Research: Atmospheres, 109, https://doi.org/https://doi.org/10.1029/2004JD004559, https://agupubs.onlinelibrary.wiley.com/doi/abs/10.1029/2004JD004559, 2004.

- 1245 Zeng, G., Wood, S. W., Morgenstern, O., Jones, N. B., Robinson, J., and Smale, D.: Trends and variations in CO, C₂H₆, and HCN in the Southern Hemisphere point to the declining anthropogenic emissions of CO and C₂H₆, Atmospheric Chemistry and Physics, 12, 7543–7555, https://doi.org/10.5194/acp-12-7543-2012, https://acp.copernicus.org/articles/12/7543/2012/, 2012.
 - Zhou, M., Langerock, B., Vigouroux, C., Sha, M. K., Ramonet, M., Delmotte, M., Mahieu, E., Bader, W., Hermans, C., Kumps, N., Metzger, J.-M., Duflot, V., Wang, Z., Palm, M., and De Mazière, M.: Atmospheric CO and CH₄ time series and seasonal variations on Reunion
- 1250 Island from ground-based in situ and FTIR (NDACC and TCCON) measurements, Atmospheric Chemistry and Physics, 18, 13881– 13901, https://doi.org/10.5194/acp-18-13881-2018, https://acp.copernicus.org/articles/18/13881/2018/, 2018.
 - Zhou, M., Langerock, B., Vigouroux, C., Sha, M. K., Hermans, C., Metzger, J.-M., Chen, H., Ramonet, M., Kivi, R., Heikkinen, P., Smale, D., Pollard, D. F., Jones, N., Velazco, V. A., García, O. E., Schneider, M., Palm, M., Warneke, T., and De Mazière, M.: TC-CON and NDACC X_{CO} measurements: difference, discussion and application, Atmospheric Measurement Techniques, 12, 5979–5995, https://doi.org/10.5194/amt-12-5979-2019, https://amt.copernicus.org/articles/12/5979/2019/, 2019.
- 1255







Figure 2. Relative biases between co-located S5P (standard XCH₄ product - top panel; bias-corrected XCH₄ product - bottom panel) and TCCON XCH₄ data with a priori alignment are plotted as a function of the surface albedo retrieved by S5P at 25 TCCON stations within the period between November 2017 and September 2020. Spatial co-location with radius of 100 km and time of ± 1 hour around the satellite overpass was used.



Figure 3. Mosaic plots showing relative biases between co-located S5P (standard XCH₄ product - top panel; bias-corrected XCH₄ product - bottom panel) and TCCON XCH₄ data with a priori alignment at 25 TCCON stations within the period between November 2017 and September 2020. Spatial co-location with radius of 100 km and time of ± 1 hour around the satellite overpass was used. The time resolution of the data shown here is weekly. The stations are sorted with decreasing latitude.

2019-06-01

2019-09-01

2019-12-01

2020-03-01

2020-06-01

2020-09-01

2019-03-01

45.0

4.5

LAUDER

2017-12-01

2018-03-01

2018-06-01

2018-09-01

2018-12-01



Figure 4. XCH₄ time series for all TCCON data (grey), S5P bias-corrected data (light blue), S5P data co-located with TCCON data (blue) and co-located TCCON data with a priori alignment (black) at each site ordered with decreasing latitude. Spatial co-location with radius of 100 km and time of ± 1 hour around the satellite overpass was used.



Figure 5. same as Fig. 4



Figure 6. Relative difference [(satellite - ground-based)/ground-based] of XCH₄ time series for all co-located S5P bias-corrected data and TCCON data with a priori alignment as the reference data at each site ordered with decreasing latitude as in Fig. 4. Spatial co-location with radius of 100 km and time of ± 1 hour around the satellite overpass was used.



Figure 7. same as Fig. 6



Figure 8. Taylor diagram for daily mean differences between S5P bias-corrected XCH₄ and TCCON XCH₄ data with a priori alignment at the 25 TCCON stations within the period between November 2017 and September 2020. Spatial co-location with radius of 100 km and time of ± 1 hour around the satellite overpass was used. The stations are sorted with decreasing latitude.







Figure 10. Relative biases between co-located S5P (standard XCH₄ product - top panel; bias-corrected XCH₄ product - bottom panel) and NDACC XCH₄ data smoothed with S5P a priori and additionally smoothed with the S5P column averaging kernel are plotted as a function of the surface albedo retrieved by S5P at 20 NDACC stations within the period between November 2017 and September 2020. Spatial co-location with radius of 100 km and time of \pm 3 hour around the satellite overpass was used.



Figure 11. Mosaic plots showing relative biases between co-located S5P (standard XCH₄ product - top panel; bias-corrected XCH₄ product - bottom panel) and NDACC XCH₄ data smoothed with S5P a priori and additionally smoothed with the S5P column averaging kernel at 20 NDACC stations within the period between November 2017 and September 2020. Spatial co-location with radius of 100 km and time of ± 3 hour around the satellite overpass was used. The time resolution of the data shown here is weekly. The stations are sorted with decreasing latitude.



Figure 12. XCH₄ time series for all NDACC data (grey), S5P bias-corrected data (light cyan), S5P data co-located with NDACC data (cyan) and co-located NDACC data smoothed with S5P a priori and additionally smoothed with the S5P column averaging kernel (black) at each site ordered with decreasing latitude. Spatial co-location with radius of 100 km and time of ± 3 hour around the satellite overpass was used.





 ${}_{2017}^{-12} {}_{2018}^{01} {}_{0218}^{0.01} {}_{2018}^{0.01} {}_{2018}^{0.01} {}_{2018}^{0.01} {}_{2019}^{0.01} {}_{2019}^{0.01} {}_{02019}^{0.01} {}_{2019}^{0.01} {}_{2019}^{0.01} {}_{2019}^{0.01} {}_{2019}^{0.01} {}_{2020}^{0.01} {}_{020}^{0.01}$



Figure 13. same as Fig. 12



Figure 14. Relative difference [(satellite - ground-based)/ground-based] of XCH₄ time series for all co-located S5P bias-corrected data and NDACC data smoothed with S5P a priori and additionally smoothed with the S5P column averaging kernel as the reference data at each site ordered with decreasing latitude as in Fig. 12. Spatial co-location with radius of 100 km and time of ± 3 hour around the satellite overpass was used. 53



 $_{2017}^{-12.01} + 0.3.01 \\ _{2018}^{-0.01} + 0.2018 \\ - 0.2018 \\ - 0.2018 \\ - 0.2018 \\ - 0.2018 \\ - 0.2019$

 $\begin{array}{c} {}_{2017-12.01} \\ {}_{2018-0.2018-0.2018-0.2018-0.2018-1.2019-0.2019-0.2019-0.2019-0.2019-0.2019-0.2020-0.2$

Figure 15. same as Fig. 14



Figure 16. Taylor diagram for daily mean differences between S5P bias-corrected XCH₄ and NDACC XCH₄ data smoothed with S5P a priori and additionally smoothed with the S5P column averaging kernel at the 20 NDACC stations within the period between November 2017 and September 2020. Spatial co-location with radius of 100 km and time of ± 3 hour around the satellite overpass was used. The stations are sorted with decreasing latitude.



Figure 17. Relative biases between co-located S5P bias-corrected XCH₄ and TCCON XCH₄ data with a priori alignment are plotted as a function of the S5P measurement solar zenith angles retrieved at a few TCCON stations within the period between November 2017 and September 2020. Spatial co-location with radius of 100 km and time of ± 1 hour around the satellite overpass was used. The colours represent the different months from January (1) till December (12) of a year.







Figure 19. XCO time series for all unscaled TCCON data (grey), all S5P data (light red), S5P data co-located with TCCON data (red) and co-located unscaled TCCON data with a priori alignment (black) at each site ordered with decreasing latitude. Spatial co-location with radius of 50 km and time of ± 1 hour around the satellite overpass was used.



Figure 20. same as Fig. 19



Figure 21. Relative difference [(satellite - ground-based)/ground-based] of XCO time series for all co-located S5P data and unscaled TCCON data with a priori alignment as the reference data at each site ordered with decreasing latitude as in Fig. 19. Spatial co-location with radius of 50 km and time of ± 1 hour around the satellite overpass was used.



Figure 22. same as Fig. 21



S5P xCO OFFL ALL sm 1hr and FTIR CO dry air mol fraction (xCO) relative differences (SAT-GB)/GB (weekly mean, surf-toa)

Figure 23. Mosaic plots showing relative biases between co-located S5P and TCCON XCO data with a priori alignment (standard - top panel; unscaled - bottom panel) at 28 TCCON stations within the period between November 2017 and September 2020. Spatial co-location with radius of 50 km and time of ± 1 hour around the satellite overpass was used. The time resolution of the data shown here is weekly. The stations are sorted with decreasing latitude.



Figure 24. Taylor diagram for daily mean differences between S5P and TCCON unscaled XCO data with a priori alignment at 28 TCCON stations within the period between November 2017 and September 2020. Spatial co-location with radius of 50 km and time of ± 1 hour around the satellite overpass was used. The stations are sorted with decreasing latitude.



Figure 25. S5P CO column number density plotted around NDACC station at Altzomoni for one sample day. Top-left panel shows all available S5P pixels containing CO data in the overpass file. Top-right panel shows the co-located S5P pixels with 50 km radius selection criterion. Bottom panel shows the co-located S5P pixels with the cone co-location criterion with 1° opening angle of the cone at the highest altitude. The yellow line in the plots represent the line-of-sight of the ground-based FTIR at the time of the satellite overpass over the site.







Figure 27. CO column time series for all NDACC data (grey), all S5P data (light red), S5P data co-located with NDACC data (red) and co-located NDACC data smoothed with S5P a priori and additionally smoothed with the S5P column averaging kernel (black) at each site ordered with decreasing latitude. Spatial co-location with radius of 50 km and time of ± 3 hour around the satellite overpass was used.



 $_{2017}^{-12.01} + 0.03.01 \\ \times 0.018 \\ - 0.018 \\ - 0.018 \\ - 0.018 \\ - 0.018 \\ - 0.018 \\ - 0.018 \\ - 0.018 \\ - 0.019 \\ - 0.0$

Figure 28. same as Fig. 27



Figure 29. Relative difference [(satellite - ground-based)/ground-based] of CO column time series for all co-located S5P data and NDACC data smoothed with S5P a priori and additionally smoothed with the S5P column averaging kernel as the reference data at each site ordered with decreasing latitude as in Fig. 27. Spatial co-location with radius of 50 km and time of ± 3 hour around the satellite overpass was used.



Figure 30. same as Fig. 29



Figure 31. Mosaic plots showing relative biases between co-located S5P and NDACC CO column data smoothed with S5P a priori and additionally smoothed with the S5P column averaging kernel at 23 NDACC stations within the period between November 2017 and September 2020. Spatial co-location with radius of 50 km and time of ± 3 hour around the satellite overpass was used. The time resolution of the data shown here is weekly. The stations are sorted with decreasing latitude.



Figure 32. Taylor diagram for daily mean differences between S5P and NDACC CO column data smoothed with S5P a priori and additionally smoothed with the S5P column averaging kernel at 23 NDACC stations within the period between November 2017 and September 2020. Spatial co-location with radius of 50 km and time of ± 3 hour around the satellite overpass was used. The stations are sorted with decreasing latitude.










Figure 35. Relative biases between co-located S5P XCO and TCCON unscaled XCO data with a priori aligned are plotted as a function of the S5P measurement solar zenith angles retrieved at a few TCCON stations within the period between November 2017 and September 2020. Spatial co-location with radius of 50 km and time of ± 1 hour around the satellite overpass was used. The colours represent the different months from January (1) till December (12) of a year.

Product ID	Stream	Version	In operation from (orbit no., date)	In operation until (orbit no., date)
	RPRO	01.02.02	0657, 2017-11-28	5346, 2018-10-25
		01.03.01	2818, 2018-04-30	5832, 2018-11-28
		01.03.02	2463, 2018-04-04	2477, 2018-04-05
$L2_CH_4$	OFFL	01.02.02	5833, 2018-11-28	7424, 2019-03-20
		01.03.00	7425, 2019-03-20	7906, 2019-04-23
		01.03.01	7907, 2019-04-23	8814, 2019-06-26
		01.03.02	8812, 2019-06-26	current version
	RPRO	01.02.02	5236, 2018-10-17	5346, 2018-10-25
		01.03.01	2818, 2018-04-30	5832, 2018-11-28
		01.03.02	2463, 2018-04-04	2477, 2018-04-05
L2_CO	OFFL	01.02.00	5346, 2018-10-25	5832, 2018-11-28
		01.02.02	5833, 2018-11-28	7424, 2019-03-20
		01.03.00	7425, 2019-03-20	7906, 2019-04-23
		01.03.01	7907, 2019-04-23	8814, 2019-06-26
		01.03.02	8815, 2019-06-26	current version

Table 1. S5P operational CH₄ RPRO+OFFL data versions and CO RPRO+OFFL data versions used in the present work.

Table 2. List of FTIR stations that are associated to TCCON and contributed to the present work by providing public and rapid delivery data. The stations marked with a star (*) are not yet associated to TCCON but perform observations and data analysis fully compatible with TCCON guidelines. Active dates correspond to the dates for which the measurements were provided from the satellite launch till the present work.

Station	Latitude	Longitude	Altitude (km a.s.l.)	Active dates	Data reference
EUREKA	80.05° N	86.42° W	0.61	Nov 2017 - Present	Strong et al. (2019)
NY-ÅLESUND	78.90° N	11.90° E	0.02	Nov 2017 - Present	Notholt et al. (2014b)
SODANKYLÄ	67.37° N	26.63° E	0.19	Nov 2017 - Present	Kivi et al. (2014); Kivi and Heikkinen (2016)
EAST TROUT LAKE	54.35° N	104.99° W	0.50	Nov 2017 - Present	Wunch et al. (2018)
BIAŁYSTOK	53.23° N	23.05° E	0.18	Nov 2017 - Oct 2018	Deutscher et al. (2019)
BREMEN	53.10° N	8.85° E	0.03	Nov 2017 - Present	Notholt et al. (2014a)
KARLSRUHE	49.10° N	8.44° E	0.12	Nov 2017 - Present	Hase et al. (2015)
PARIS	48.85° N	2.36° E	0.06	Nov 2017 - Present	Té et al. (2014)
ORLÉANS	47.97° N	2.11° E	0.13	Nov 2017 - Present	Warneke et al. (2019)
GARMISCH	47.48° N	11.06° E	0.74	Nov 2017 - Present	Sussmann and Rettinger (2018a)
ZUGSPITZE	47.42° N	10.98° E	2.96	Nov 2017 - Present	Sussmann and Rettinger (2018b)
PARK FALLS	45.95° N	90.27° W	0.44	Nov 2017 - Present	Wennberg et al. (2017)
RIKUBETSU	43.46° N	143.77° E	0.38	Nov 2017 - Present	Morino et al. (2018c)
XIANGHE*	39.75° N	116.96° E	0.05	Nov 2017 - Present	Yang et al. (2019)
LAMONT	36.60° N	97.49° W	0.32	Nov 2017 - Present	Wennberg et al. (2016b)
TSUKUBA	36.05° N	140.12° E	0.03	Nov 2017 - Present	Morino et al. (2018a)
NICOSIA*	35.14° N	33.38° E	0.19	Aug 2019 - Present	Petri et al. (2019)
EDWARDS	34.96° N	117.88° W	0.70	May 2018 - Present	Iraci et al. (2016)
JPL	34.20° N	118.18° W	0.39	Nov 2017 - May 2018	Wennberg et al. (2016a)
PASADENA	34.14° N	118.13° W	0.23	Nov 2017 - Present	Wennberg et al. (2015)
SAGA	33.24° N	130.29° E	0.01	Nov 2017 - Present	Kawakami et al. (2014)
IZAÑA	28.30° N	16.50° W	2.37	Nov 2017 - Present	Blumenstock et al. (2017)
BURGOS	18.53° N	120.65° E	0.04	Nov 2017 - Present	Morino et al. (2018b); Velazco et al. (2017)
ASCENSION	7.92° S	14.33° W	0.01	Nov 2017 - Present	Feist et al. (2014)
DARWIN	12.46° S	130.93° E	0.04	Nov 2017 - Present	Griffith et al. (2014a)
RÉUNION	20.90° S	55.49° E	0.09	Nov 2017 - Present	De Mazière et al. (2017)
WOLLONGONG	34.41° S	150.88° E	0.03	Nov 2017 - Present	Griffith et al. (2014b)
LAUDER	45.04° S	169.68° E	0.61	Nov 2017 - Present	Sherlock et al. (2014); Pollard et al. (2019)

Station	Latitude	Longitude	Altitude (km a.s.l.)	Active dates	Teams
EUREKA	80.05° N	86.42° W	0.61	Nov 2017 - Present	U. of Toronto; Batchelor et al. (01 Jul. 2009)
NY-ÅLESUND	78.90° N	11.90° E	0.02	Nov 2017 - Present	U. of Bremen
THULE	76.52° N	68.77° W	0.22	Nov 2017 - Present	NCAR; Hannigan et al. (01 Sep. 2009)
KIRUNA	67.84° N	20.40° E	0.42	Nov 2017 - Present	KIT-ASF; IRF Kiruna
SODANKYLÄ*	67.37° N	26.63° E	0.19	Nov 2017 - Present	FMI; BIRA-IASB
HARESTUA	60.20° N	10.80° E	0.60	Nov 2017 - Present	Chalmers
ST.PETERSBURG	59.88° N	29.83° E	0.02	Nov 2017 - Present	SPbU; Makarova et al. (2015)
BREMEN	53.10° N	8.85° E	0.03	Nov 2017 - Present	U. of Bremen
KARLSRUHE*	49.10° N	8.44° E	0.12	Nov 2017 - Present	KIT-ASF
GARMISCH*	47.48° N	11.06° E	0.74	Nov 2017 - Present	KIT-IFU
ZUGSPITZE	47.42° N	10.98° E	2.96	Nov 2017 - Present	KIT-IFU
JUNGFRAUJOCH	46.55° N	7.98° E	3.58	Nov 2017 - Present	U. of Liège
TORONTO	43.60° N	79.36° W	0.17	Nov 2017 - Present	U. of Toronto; Wiacek et al. (01 Mar. 2007)
RIKUBETSU	43.46° N	143.77° E	0.38	Nov 2017 - Present	Nagoya U.; NIES
BOULDER	40.04° N	105.24° W	1.61	Nov 2017 - Present	NCAR; Ortega et al. (2019)
IZAÑA	28.30° N	16.50° W	2.37	Nov 2017 - Present	AEMET; KIT-ASF
MAUNA LOA	19.54° N	155.57° W	3.40	Nov 2017 - Present	NCAR
ALTZOMONI	19.12° N	98.66° W	3.98	Nov 2017 - Present	UNAM
PARAMARIBO	5.81° N	55.21° W	0.03	Nov 2017 - Present	U. of Bremen
PORTO.VELHO*	8.77° S	296.13° E	0.09	Nov 2017 - Present	BIRA-IASB
LA.RÉUNION.MAÏDO	21.08° S	55.38° E	2.16	Nov 2017 - Present	BIRA-IASB
WOLLONGONG	34.41° S	150.88° E	0.03	Nov 2017 - Present	U. of Wollongong
LAUDER	45.04° S	169.68° E	0.37	Nov 2017 - Present	NIWA
ARRIVAL HEIGHTS	77.82° S	166.65° E	0.20	Nov 2017 - Present	NIWA

Table 3. List of FTIR stations that are associated to NDACC-IRWG and contributed to the present work by providing public and rapid delivery data. The stations marked with a star (*) are not yet associated to NDACC but perform observations and data analysis fully compatible with NDACC guidelines. The location of the stations and the teams involved are indicated for the respective stations.

Table 4. S5P XCH ₄ validation results against TCCON XCH ₄ data at 25 stations for the period between November 2017 and September 2020. Spatial co-location with radius of 100 km	
or cone with 1° opening angle along the FTIR line-of-sight and time co-location of ± 1 hour around the satellite overpass was used. TCCON station (column 1) are sorted according to	
the decreasing latitude (column 2). The column with title 'No'' represents the number of co-located measurements, column title 'Std' represents the standard deviation of the time series	
of the ground-based data relative to the standard deviation of the time series of the S5P data, column title 'Corr' represents the correlation coefficient between the S5P and the reference	
ground-based data, column title 'Rel diff bias' represents the relative difference ((SAT-GB)/GB) bias in percent and column title 'Rel diff std' represents the standard deviation of the	
relative bias in percent.	

		S5	P bc X	CH ₄ smot	oth 100 km 1	hr	S5P	XCH ₄ s.	mooth 100 k	m 1 hr	S	5P bc XC	$H_4 100 \text{ km}$	1 hr	S5P l	oc XCH	4 smooth o	cone 100 km	ı 1 hr
Site	Lat	No.	Std	Corr	Rel diff	Rel diff	Std	Corr	Rel diff	Rel diff	Std	Corr	Rel diff	Rel diff	No.	Std	Corr	Rel diff	Rel diff
					bias (%)	std (%)			bias (%)	std (%)			bias (%)	std (%)				bias (%)	std (%)
EUREKA	80	1384	-	0.62	0.8	0.77	0.8	0.79	0.22	0.67	-	0.66	0.87	0.72	648	0.0	0.74	0.83	0.66
NY-ÅLESUND	78.9	113	-	0.97	0.63	0.13	1	0.95	-0.23	0.17	0.9	0.95	0.71	0.17	11	2.8	-0.56	0.89	0.09
SODANKYLÄ	67.4	5915	0.9	0.37	-0.84	0.99	0.9	0.36	-1.87	1.01	0.9	0.4	-0.7	0.96	4681	0.9	0.42	-0.92	0.93
EAST TROUT LAKE	54.3	12302	0.8	0.64	-0.9	0.93	0.8	0.62	-1.85	0.95	0.9	0.67	-0.72	0.91	10358	0.9	0.65	-0.98	0.88
BIAŁYSTOK	53.2	1821	0.7	0.46	-0.46	0.56	0.8	0.37	-0.99	0.57	0.8	0.49	-0.21	0.56	1623	0.8	0.41	-0.35	0.54
BREMEN	53.1	1150	0.9	0.81	-0.15	0.49	1	0.79	-0.68	0.49	1	0.82	0.03	0.5	1081	0.9	0.8	-0.17	0.5
KARLSRUHE	49.1	4592	0.8	0.79	-0.36	0.48	0.8	0.82	-1.04	0.49	0.8	0.79	-0.13	0.48	4114	0.8	0.81	-0.35	0.45
PARIS	48.8	4999	-	0.67	-0.03	0.49	0.8	0.55	-0.67	0.63	-	0.69	0.18	0.48	4370	0.9	0.67	0.01	0.53
ORLÉANS	48	5984	0.8	0.79	-0.34	0.52	0.6	0.79	-0.94	0.65	0.8	0.79	-0.15	0.52	5712	0.8	0.79	-0.44	0.52
GARMISCH	47.5	3149	0.9	0.77	0.19	0.49	0.8	0.77	-0.61	0.51	0.9	0.79	0.41	0.48	1520	0.9	0.83	0.1	0.43
ZUGSPITZE	47.4	578	1.3	0.59	0.03	0.73	1.4	0.62	-0.78	0.69	1.4	0.61	0.23	0.75	290	1.5	0.48	0.09	0.69
PARK FALLS	45.9	7201	0.8	0.7	-1.46	0.75	0.8	0.67	-2.35	0.77	0.8	0.71	-1.27	0.74	6671	0.8	0.7	-1.43	0.77
RIKUBETSU	43.5	2466	0.8	0.83	-0.1	0.59	0.9	0.8	-0.81	0.62	0.9	0.85	0.05	0.55	2206	0.8	0.83	-0.1	0.6
XIANGHE	39.8	4395	1	0.87	-0.45	0.61	1.2	0.81	-0.52	0.68	-	0.87	-0.23	0.61	3706	0.9	0.88	-0.47	0.6
LAMONT	36.6	9608	0.9	0.84	-0.36	0.53	0.9	0.84	-0.58	0.53	0.9	0.84	-0.22	0.52	8053	0.9	0.85	-0.25	0.49
TSUKUBA	36	4655	0.9	0.84	-0.02	0.43	1	0.84	-0.46	0.43	-	0.86	0.1	0.42	4494	0.9	0.86	0.03	0.43
NICOSIA	35.1	3414	0.9	0.85	0.62	0.41	0.8	0.83	0.89	0.52	-	0.85	0.65	0.41	3325	0.9	0.85	0.6	0.41
EDWARDS	35	24350	0.9	0.85	-0.07	0.45	-	0.83	0.82	0.47	0.9	0.85	-0.01	0.44	22987	0.9	0.86	-0.09	0.44
JPL	34.2	3233	1.1	0.33	-1.34	0.63	1	0.4	-0.54	0.65	1.1	0.34	-1.2	0.62	1954	0.9	0.27	-1.14	0.67
PASADENA	34.1	20646	0.9	0.79	-0.73	0.57	0.9	0.74	-0.03	0.61	0.9	0.81	-0.67	0.55	15868	0.9	0.85	-0.55	0.52
SAGA	33.2	2539	0.9	0.79	0.09	0.53	0.9	0.79	-0.6	0.51	0.9	0.81	0.21	0.5	1943	0.9	0.83	0.27	0.49
BURGOS	18.5	417	-	0.76	0.23	0.36	0.9	0.77	-0.75	0.37	0.9	0.77	0.28	0.34	269	0.8	0.76	0.26	0.33
DARWIN	-12.5	3598	0.5	0.23	-0.32	0.71	0.5	0.3	-0.53	0.71	0.5	0.27	-0.24	0.69	3267	0.5	0.24	-0.33	0.7
MOLLONGONG	-34.4	4765	0.9	0.55	-0.88	0.69	0.8	0.56	-1.57	0.69	0.8	0.54	-0.81	0.69	4677	0.9	0.56	-0.86	0.67
LAUDER	-45	6708	-	0.8	-0.18	0.5	0.9	0.84	-0.6	0.47	0.9	0.8	-0.12	0.51	1771	0.0	0.92	-0.48	0.37
Mean all stations			0.9	0.7	-0.26	0.57	0.9	0.7	-0.68	0.59	0.9	0.71	-0.12	0.57		-	0.65	-0.23	0.55
STD all stations			0.2	0.19	0.56	0.18	0.2	0.18	0.74	0.17	0.2	0.18	0.55	0.17		0.4	0.32	0.58	0.18

Site Lat No. Sid Corr Rel diff Rel diff Sid Corr Ner Sid Corr Sid Sid <t< th=""><th>d Corr 5 0.37 9 0.92 9 0.14 9 0.14 1 0.45 9 0.52 9 0.56 0 0.79 9 0.56</th><th>Rel diff bias (%) 0.95 4.73 4.52 -0.15 0.15 0.86 0.31 1.44</th><th>Rel diff std (%) 2.57 0.98 1.09 1.43 1.07 1.57</th><th>Std 0</th><th>1</th><th></th><th>шсп</th><th>S</th><th>5P bc X(</th><th>CH4 100 km</th><th>3 hr</th><th>S5P</th><th>be XCH</th><th>4 smooth</th><th>cone 100 kn</th><th>a 3 hr</th></t<>	d Corr 5 0.37 9 0.92 9 0.14 9 0.14 1 0.45 9 0.52 9 0.56 0 0.79 9 0.56	Rel diff bias (%) 0.95 4.73 4.52 -0.15 0.15 0.86 0.31 1.44	Rel diff std (%) 2.57 0.98 1.09 1.43 1.07 1.57	Std 0	1		шсп	S	5P bc X(CH4 100 km	3 hr	S5P	be XCH	4 smooth	cone 100 kn	a 3 hr
EUREKAbias (%) bias (%)std (%) std (%)std (%) bias (%)std (%) std (%)std (%) biasbias (%) biasstd (%) biasbiasbiasbiasbiasEUREKA80.12.641.02.90.924.730.9830.8233THULE76.52.621.20.74.521.091.10.16-1.SODANKYLÅ67.41691.10.450.151.071.10.42-0.HARESTUA60.21612.20.310.861.572.20.29-0.STPETERSBURG59.964.71.30.490.310.911.10.42-0.BREMEN53.11881.90.521.441.341.90.50.BREMEN53.11881.90.550.330.472.20.290.BREMEN53.11881.90.550.790.330.470.00.BREMEN53.11881.90.550.790.330.470.0BREMEN53.11881.90.560.790.330.470.0CGSPITZE47.41.341.90.50.50.50.640.0 <th> 5 0.37 9 0.92 9 0.14 9 0.14 1 0.45 9 0.79 9 0.79 0.56 1 0.7 </th> <th>bias (%) 0.95 4.73 4.52 -0.15 0.15 0.15 0.15 0.31 1.44</th> <th>std (%) 2.57 2.57 0.98 1.09 1.07 1.07 1.57 </th> <th>3.4</th> <th>1 10</th> <th>Rel diff</th> <th>Rel diff</th> <th>Std</th> <th>Corr</th> <th>Rel diff</th> <th>Rel diff</th> <th>No.</th> <th>Std</th> <th>Corr</th> <th>Rel diff</th> <th>Rel diff</th>	 5 0.37 9 0.92 9 0.14 9 0.14 1 0.45 9 0.79 9 0.79 0.56 1 0.7 	bias (%) 0.95 4.73 4.52 -0.15 0.15 0.15 0.15 0.31 1.44	std (%) 2.57 2.57 0.98 1.09 1.07 1.07 1.57	3.4	1 10	Rel diff	Rel diff	Std	Corr	Rel diff	Rel diff	No.	Std	Corr	Rel diff	Rel diff
EUREKA 80.1 264 2.5 0.37 0.95 2.57 3.4 0.34 0. NY-ÅLESUND 78.9 10 2.9 0.92 4.73 0.98 3 0.82 3.7 THULE 76.5 262 1.2 0.7 4.52 1.09 1.2 0.67 3. KIRUNA 67.8 367 0.9 0.14 -0.15 1.43 1 0.16 -1. SODANKYLÄ 67.4 1694 1.1 0.45 0.15 1.07 1.1 0.42 0.1 SODANKYLÄ 67.4 1694 1.1 0.45 0.31 0.86 1.3 0.49 0.31 0.47 0.1 SODANKYLÄ 67.4 169 2.2 0.31 0.36 1.1 0.47 0.3 0.47 0.1 SODANKYLÄ 67.4 188 1.9 0.52 1.44 1.3 0.39 0.6 1.1 0.47 0.6 0.77 0.6 <th> 5 0.37 9 0.92 9 0.14 9 0.14 9 0.14 9 0.56 0.79 0.56 </th> <th>0.95 4.73 4.52 -0.15 0.15 0.31 0.31</th> <th>2.57 0.98 1.09 1.43 1.07 1.57</th> <th>3.4 (</th> <th>4</th> <th>ias (%)</th> <th>std (%)</th> <th></th> <th></th> <th>bias (%)</th> <th>std (%)</th> <th></th> <th></th> <th></th> <th>bias (%)</th> <th>std (%)</th>	 5 0.37 9 0.92 9 0.14 9 0.14 9 0.14 9 0.56 0.79 0.56 	0.95 4.73 4.52 -0.15 0.15 0.31 0.31	2.57 0.98 1.09 1.43 1.07 1.57	3.4 (4	ias (%)	std (%)			bias (%)	std (%)				bias (%)	std (%)
NY-ÅLESUND 78.9 10 2.9 0.92 4.73 0.98 3 0.82 3: 3 THULE 76.5 262 1.2 0.7 4.52 1.09 1.2 0.67 3: 3 THULE 76.5 262 1.2 0.7 4.52 1.09 1.2 0.67 3: 3 KIRUNA 67.4 1694 1.1 0.45 0.15 1.07 1.1 0.42 0.0 ASDANKYLÄ 67.4 1694 1.1 0.45 0.15 1.07 1.1 0.42 0.0 0.1 0.41 0.42 0.0 0.1 0.41 0.41 0.42 0.0 0.1 0.41	9 0.92 2 0.7 9 0.14 1 0.45 2 0.31 9 0.56 9 0.79 0.56	4.73 4.52 -0.15 0.15 0.86 0.31 1.44	0.98 1.09 1.43 1.07 1.57).34	0.16	2.56	2.5	0.35	0.41	2.71	67	1.3	0.74	0.88	0.75
THULE 76.5 262 1.2 0.7 4.52 1.09 1.2 0.67 $3.$ KIRUNA 67.8 367 0.9 0.14 -0.15 1.43 1 0.16 $-1.$ SODANKYLÄ 67.4 1694 1.1 0.45 0.15 1.07 1.1 0.42 $-0.$ HARESTUA 60.2 161 2.2 0.31 0.86 1.57 2.2 0.29 $-0.$ HARESTUA 60.2 161 2.2 0.31 0.86 1.57 2.2 0.29 ST.PETERSBUG 59.9 647 1.3 0.49 0.31 0.9 1.2 0.42 0.14 BREMEN 53.1 188 1.9 0.52 1.44 1.34 1.9 0.2 0.2 BREMEN 53.1 188 1.9 0.52 1.44 1.34 1.9 0.2 0.2 BREMEN 53.1 188 1.9 0.52 1.44 1.34 1.2 0.29 0.0 BREMEN 53.1 1.88 1.9 0.52 1.44 1.34 1.2 0.29 0.2 BREMEN 53.1 1.88 1.9 0.56 0.77 0.2 0.29 0.77 CARLISRUHE 49.1 53 1.9 0.77 0.76 0.9 0.77 0.2 CARMISCH 47.5 700 1.9 0.77 0.76 0.9 0.64 -0.2 UNORGRIZCH 43.5 31 1.9 <td>2 0.7 9 0.14 1 0.45 2 0.31 3 0.49 9 0.79 0.56 0.79</td> <td>4.52 -0.15 0.15 0.86 0.31 1.44</td> <td>1.09 1.43 1.07 1.57</td> <td>3 (</td> <td>).82</td> <td>3.79</td> <td>1.03</td> <td>2.9</td> <td>0.94</td> <td>2.53</td> <td>0.9</td> <td></td> <td></td> <td></td> <td></td> <td></td>	2 0.7 9 0.14 1 0.45 2 0.31 3 0.49 9 0.79 0.56 0.79	4.52 -0.15 0.15 0.86 0.31 1.44	1.09 1.43 1.07 1.57	3 ().82	3.79	1.03	2.9	0.94	2.53	0.9					
KIRUNA 67.8 367 0.9 0.14 -0.15 1.43 1 0.16 $-1.$ SODANKYLÄ 67.4 1694 1.1 0.45 0.15 1.07 1.1 0.42 -0.6 HARESTUA 60.2 161 2.2 0.31 0.86 1.57 2.2 0.29 -0.6 ST.PETERSBURG 59.9 647 1.3 0.49 0.31 0.9 1.1 0.42 0.2 0.2 BREMEN 53.1 188 1.9 0.52 1.44 1.34 1.9 0.2 0.7 0.2 BREMEN 53.1 188 1.9 0.52 1.44 1.34 1.9 0.5 0.7 BREMEN 53.1 188 1.9 0.52 1.44 1.34 1.9 0.7 0.2 ZUGSPITZE 47.4 485 1.1 0.75 0.76 0.76 0.7	9 0.14 1 0.45 2 0.31 3 0.49 9 0.52 9 0.79 0.56	-0.15 0.15 0.86 0.31 1.44	1.43 1.07 1.57	1.2 0	.67	3.47	1.12	1.2	0.68	2.66	1.12	65	0.9	0.61	4.98	1.14
SODANKYLÄ 674 1694 1.1 0.45 0.15 1.07 1.1 0.42 -0. HARESTUA 60.2 161 2.2 0.31 0.86 1.57 2.2 0.29 -0. ST.PETERSBURG 59.9 647 1.3 0.49 0.31 0.96 1.3 0.39 -0. BREMEN 53.1 188 1.9 0.52 1.44 1.34 1.9 0.5 0. BREMEN 53.1 188 1.9 0.52 1.44 1.34 1.9 0.5 0. ZUGSPITZE 47.4 485 1.1 0.7 0.65 0.7 0.	1 0.45 2 0.31 3 0.49 9 0.52 9 0.79 0.56 0.56	0.15 0.86 0.31 1.44	1.07	1 0	0.16	-1.12	1.37	0.8	0.3	-0.55	1.23	256	1	0.2	-0.4	1.38
HARESTUA 60.2 161 2.2 0.31 0.86 1.57 2.2 0.29 $-0.$ STPETERSBURG 59.9 647 1.3 0.49 0.31 0.9 1.3 0.39 $-0.$ BREMEN 53.1 188 1.9 0.52 1.44 1.34 1.9 0.5 0.1 KARLSRUHE 49.1 591 0.9 0.79 0.33 0.47 0.8 0.77 $-0.$ KARLSRUHE 49.1 591 0.9 0.79 0.33 0.47 0.8 0.77 $-0.$ GARMISCH 47.5 700 1 0.56 0.79 0.75 1.1 0.56 $-0.$ ZUGSPITZE 47.4 485 1.1 0.77 0.65 0.76 0.7 0.6 JUNGFRAUJOCH 46.6 178 0.9 0.66 -0.23 0.76 0.9 0.64 $-0.$ JUNGFRAUJOCH 43.6 178 0.9 0.66 -0.23 0.76 0.9 0.64 $-0.$ JUNGFRAUJOCH 43.6 337 3.5 -0.13 1.72 3.95 3.6 -0.08 1.1 JUNGFRAUJOCH 43.6 3.7 3.7 0.26 1.72 0.92 0.64 -0.06 JUNGFRAUJOCH 43.5 3.1 1.9 0.26 1.72 0.92 0.64 -0.06 RIKUBETSU 43.5 3.1 1.9 0.26 1.72 0.92 1.23 0.945 2.2 <td< td=""><td>2 0.31 3 0.49 9 0.52 9 0.79 0.56</td><td>0.86 0.31 1.44</td><td>1.57</td><td>1.1 0</td><td>0.42</td><td>-0.88</td><td>1.09</td><td>1.2</td><td>0.49</td><td>-0.17</td><td>1.11</td><td>1303</td><td>1.1</td><td>0.48</td><td>0.01</td><td>0.96</td></td<>	2 0.31 3 0.49 9 0.52 9 0.79 0.56	0.86 0.31 1.44	1.57	1.1 0	0.42	-0.88	1.09	1.2	0.49	-0.17	1.11	1303	1.1	0.48	0.01	0.96
ST.PETERSBURG 59.9 647 1.3 0.49 0.31 0.9 1.3 0.39 $-0.$ BREMEN 53.1 188 1.9 0.52 1.44 1.34 1.9 0.5 0.0 KARLSRUHE 49.1 53.1 188 1.9 0.52 1.44 1.34 1.9 0.5 0.0 KARLSRUHE 49.1 591 0.9 0.52 1.44 1.34 1.9 0.5 0.0 GARMISCH 47.5 700 1 0.56 0.77 0.0 0.77 0.0 ZUGSPITZE 47.4 485 1.1 0.77 0.65 0.77 0.0 ZUGSPITZE 47.4 485 1.1 0.77 0.8 0.77 0.0 ZUGSPITZE 47.4 1.88 0.9 0.66 0.77 0.0 ZUGSPITZE 47.5 3.37 3.5 0.13 1.72	3 0.49 9 0.52 9 0.79 0.56 1 0.7	0.31 1.44		2.2 (0.29	-0.27	1.57	2.3	0.31	0.84	1.67	68	1.2	0.22	0.86	0.99
BREMEN 53.1 188 1.9 0.52 1.44 1.34 1.9 0.5 0.3 KARLSRUHE 49.1 591 0.9 0.79 0.33 0.47 0.8 0.77 -0. GARMISCH 47.5 700 1 0.56 0.79 0.33 0.47 0.8 0.77 -0. ZUGSPITZE 47.4 485 1.1 0.77 0.65 0.65 1.1 0.69 0.77 -0. ZUGSPITZE 47.4 485 1.1 0.77 0.65 0.65 1.1 0.61 0.75 1 0.51 -0. ZUGSPITZE 47.4 485 1.1 0.77 0.65 0.65 0.65 0.65 0.65 0.65 0.66 0.77 -0. ZUGSPITZE 47.5 337 3.5 -0.13 1.72 3.95 3.6 0.08 1.1 0.66 0.66 0.66 0.64 0.6 RINUBETSU 43.5	9 0.52 9 0.79 0.56 1 0.7	1.44	0.9	1.3 ().39	-0.64	0.98	1.4	0.51	-0.21	0.94	529	1.3	0.5	0.43	0.89
KARLSRUHE 49.1 591 0.9 0.79 0.33 0.47 0.8 0.77 -0. GARMISCH 47.5 700 1 0.56 0.79 0.75 1 0.51 -0. ZUGSPITZE 47.4 485 1.1 0.76 0.65 1.1 0.69 -0. JUNGFRAUJOCH 46.6 178 0.9 0.66 -0.23 0.76 0.9 -0. JUNGFRAUJOCH 46.6 178 0.9 0.66 -0.23 0.76 0.9 -0. JUNGFRAUJOCH 46.6 178 0.9 0.66 -0.23 0.77 0. -0. JUNGFRAUJOCH 45.5 3.3 3.5 -0.13 1.72 3.55 0.69 0.64 -0. INNONITO 43.5 3.1 1.9 -0.02 1.77 2.22 0.06 0. BOULDERCO 40 337 3.5 -0.02 1.17 2.22 0.06 0. <	9 0.79 0.56 1 0.7		1.34	1.9	0.5	0.95	1.35	2.1	0.56	1.16	1.42	182	1.1	0.68	1.59	0.82
GARMISCH 47.5 700 1 0.56 0.75 1 0.51 -0. ZUGSPITZE 47.4 485 1.1 0.7 0.65 0.65 1.1 0.69 -0. JUNGFRAUJOCH 46.6 178 0.9 0.66 -0.23 0.76 0.9 0.64 -0. JUNGFRAUJOCH 46.6 178 0.9 0.66 -0.23 0.76 0.9 0.64 -0. TORONTO 43.5 31 1.9 -0.02 1.72 3.95 3.6 0.08 1. RIKUBETSU 43.5 31 1.9 -0.02 1.72 2.29 0.06 0. BOULDER.CO 40 393 1.4 0.26 1.66 0.9 1.3 0.45 2.2 0.06 0.1 ALTZOMONI 19.1 325 0.9 0.45 2.44 0.68 1 0.23 2.2 PORTO.VELHO -8.8 53 1.4 0.06	0.56	0.33	0.47	0.8 (77.0	-0.35	0.51	0.9	0.79	-0.37	0.46	485	0.9	0.77	0.42	0.5
ZUGSPITZE 47.4 485 1.1 0.7 0.65 0.65 1.1 0.69 -0. JUNGFRAUJOCH 46.6 178 0.9 0.66 -0.23 0.76 0.9 0.64 -0. TORONTO 43.6 178 0.9 0.66 -0.23 0.76 0.9 0.64 -0. RIKUBETSU 43.5 31 1.9 -0.02 1.72 3.95 3.26 -0.08 1. RIKUBETSU 43.5 31 1.9 -0.02 1.77 2.29 2.2 0.06 0. BOULDER.CO 40 393 1.4 0.26 1.66 0.9 1.3 0.45 2.2 ALTZOMONI 19.1 325 0.9 0.45 2.44 0.68 1 0.23 2.2 2.0 0.06 -1 PORTO.VELHO -8.8 53 1.4 0.05 -0.76 0.0 0.65 -1 1 2.2 2.2 2.0	1 0.7	0.79	0.75	1	0.51	-0.02	0.8	1.1	0.55	0.53	0.81	141	-	0.71	0.76	0.59
JUNGFRAUJOCH 46.6 178 0.9 0.66 -0.23 0.76 0.9 0.64 -0. TORONTO 43.6 337 3.5 -0.13 1.72 3.95 3.6 -0.08 1. RIKUBETSU 43.5 31 1.9 -0.02 1.72 3.95 3.6 -0.08 1. BUULDER.CO 40 393 1.4 0.26 1.66 0.9 1.3 0.45 2.1 ALTZOMONI 19.1 325 0.9 0.45 2.44 0.68 1 0.23 2.1 PORTO.VELHO -8.8 53 1.4 0.06 -0.76 0.81 1 0.23 2.1 PORTO.VELHO -8.8 53 1.4 0.06 -0.76 0.81 1.3 0.06 -0.1 WOLLDONGONG -34.4 716 0.9 0.62 -0.04 0.76 0.6 -0.6 LAUDER 45 561 1.5 0.61 0.08		0.65	0.65	1.1 0	.69	-0.13	0.65	1.3	0.68	0.58	0.77	86	-	0.63	0.69	0.61
TORONTO 43.6 337 3.5 -0.13 1.72 3.95 3.6 -0.08 1. RIKUBETSU 43.5 31 1.9 -0.02 1.7 2.29 2.2 0.06 0.1 BOULDER.CO 40 393 1.4 0.26 1.66 0.9 1.3 0.45 2.1 ALTZOMONI 19.1 325 0.9 0.45 2.44 0.68 1 0.23 2.1 PORTO.VELHO -8.8 53 1.4 0.06 -0.76 0.81 1.3 0.06 -1. WOLLDONGONG -34.4 716 0.9 0.62 -0.04 0.78 0.9 0.6 -0. LAUDER -45 561 1.5 0.61 0.08 0.99 1.4 0.68 -0.	9 0.06	-0.23	0.76	0.9 ().64	-0.89	0.76	1.1	0.64	-0.24	0.89	31	-	0.73	-0.12	0.55
RIKUBETSU 43.5 31 1.9 -0.02 1.7 2.29 2.2 0.06 0.1 BOULDER.CO 40 393 1.4 0.26 1.66 0.9 1.3 0.45 2.1 ALTZOMONI 19.1 325 0.9 0.45 2.44 0.68 1 0.23 2.1 PORTO.VELHO -8.8 53 1.4 0.06 -0.76 0.81 1.3 0.06 -1. WOLLONGONG -34.4 716 0.9 0.62 -0.04 0.78 0.9 0.6 -0. LAUDER -45 561 1.5 0.61 0.08 0.99 1.4 0.68 -0	5 -0.13	1.72	3.95	3.6 -(0.08	1.14	3.87	3.6	-0.12	0.67	3.98	235	2.5	-0.18	1.78	3.4
BOULDER.CO 40 393 1.4 0.26 1.66 0.9 1.3 0.45 2.4 ALTZOMONI 19.1 325 0.9 0.45 2.44 0.68 1 0.23 2.3 PORTO.VELHO -8.8 53 1.4 0.06 -0.76 0.81 1.3 0.06 -1. WOLLONGONG -34.4 716 0.9 0.62 -0.04 0.78 0.9 0.6 -0. LAUDER -45 561 1.5 0.61 0.08 0.99 1.4 0.68 -0	9 -0.02	1.7	2.29	2.2 (0.06	0.96	2.17	1.8	0.09	1.56	2.03	24	1.8	0.02	2.1	2.38
ALTZOMONI 19.1 325 0.9 0.45 2.44 0.68 1 0.23 2.3 PORTO.VELHO -8.8 53 1.4 0.06 -0.76 0.81 1.3 0.06 -1. WOLLONGONG -34.4 716 0.9 0.62 -0.04 0.78 0.9 0.6 -0.1 LAUDER -45 561 1.5 0.61 0.08 0.99 1.4 0.68 -0	4 0.26	1.66	0.9	1.3 (0.45	2.09	0.79	1.1	0.34	0.83	0.74	289	1.4	0.42	1.76	0.86
PORTO.VELHO -8.8 53 1.4 0.06 -0.76 0.81 1.3 0.06 -1. WOLLDNGONG -34.4 716 0.9 0.62 -0.04 0.78 0.9 0.6 -0. LAUDER -45 561 1.5 0.61 0.08 0.99 1.4 0.68 -0	9 0.45	2.44	0.68	1	0.23	2.59	0.76	-	0.52	2.52	0.68	188	1.1	0.48	1.95	0.64
WOLLONGONG -34.4 716 0.9 0.62 -0.04 0.78 0.9 0.6 -0. LAUDER -45 561 1.5 0.61 0.08 0.99 1.4 0.68 -0	4 0.06	-0.76	0.81	1.3 (0.06	-1.98	0.82	1.5	0.16	-1.34	0.8	51	1.5	-0.05	-0.82	0.77
LAUDER -45 561 1.5 0.61 0.08 0.99 1.4 0.68 -0	9 0.62	-0.04	0.78	0.9	0.6	-0.76	0.81	1	0.57	-0.75	0.84	603	0.9	0.65	0	0.74
	5 0.61	0.08	0.99	1.4 0	.68	-0.33	0.9	1.3	0.61	-0.41	0.89	41	1.4	0.71	0.38	0.75
ARRIVAL.HEIGHTS -77.8 45 0.7 0.64 2.65 0.78 0.7 0.62 1.	7 0.64	2.65	0.78	0.7 ().62	1.66	0.8	1	0.81	2.26	0.62	5	0.3	-	2.22	0.19
Mean all stations* $1.3 0.47 0.57 1.04 1.4 0.45 -0.$	3 0.47	0.57	1.04	1.4 0	.45	-0.11	1.05	1.4	0.49	0.19	1.06		1.1	0.52	0.56	0.79
Std all stations* 0.5 0.21 0.83 0.52 0.7 0.21 1.	5 0.21	0.83	0.52	0.7 ().21	1.19	0.51	0.5	0.17	0.94	0.55		0.2	0.24	0.79	0.22

Table 5. S5P XCH₄ validation results against NDACC XCH₄ data at 20 stations for the period between November 2017 and September 2020. Spatial co-location with radius of 100 km or cone with 1° opening angle along the FTIR line-of-sight and time co-location of ± 3 hour around the satellite overpass was used. NDACC station (column 1) are sorted according to the decreasing latitude (column 2). The column with title 'No'' represents the number of co-located measurements, column title 'Std' represents the standard deviation of the time series of the ground-based data relative to the standard deviation of the time series of the S5P data, column title 'Corr' represents the correlation coefficient between the S5P and the reference

79

unexpected bias (Thule).

Table 6. S5P XCO validation results against TCCON XCO data at 28 stations for the period between November 2017 and September 2020. Spatial co-location with radius of 50 km or cone with 1° opening angle along the FTIR line-of-sight and time co-location of ± 1 hour around the satellite overpass was used. TCCON station (column 1) are sorted according to the decreasing latitude (column 2). The column with title 'No' represents the number of co-located measurements, column title 'Std' represents the standard deviation of the time series of the ground-based data relative to the standard deviation of the time series of the S5P data, column title 'Corr' represents the correlation coefficient between the S5P and the reference ground-based data, column title 'Rel diff bias' represents the relative difference ((SAT-GB)/GB) bias in percent and column title 'Rel diff std' represents the standard deviation of the relative bias in percent.

50 km 1 hr	Rel diff	std (%)	4.23	5.14	5.24	5.33	3.25	4.83	2.99	3.66	3.04	4.75	7.91	6.48	3.97	6.96	4.15	4.73	3.66	4.72	5.43	7.9	3.98	4.87	4.09	4.66	6.16	4.72	18	4.09	5.32	
ooth cone 5	Rel diff	bias (%)	5.95	7.91	5.78	5.85	2.94	3.47	0	2.04	4.52	3.35	5.71	4.25	4.05	-5.81	-1.02	3.21	3.39	3.11	-2.81	-2.99	0.42	5.1	0.22	-0.62	-0.63	3.49	2.06	2.93	2.35	
MCO sm	Corr		0.95	0.96	0.95	0.94	0.97	0.92	0.97	0.93	0.97	0.91	0.81	0.94	0.96	0.95	0.94	0.95	0.91	0.94	0.87	0.84	0.98	0.89	0.97	0.63	0.92	0.96	0.82	0.97	0.92	
oant N	Std		0.8	0.9	0.9	-	0.9	-	-	-	0.9	-	1.2	-	1.1	0.9	-	1	0.9	0.9	1.3	0.8	1	-	0.9	1.2	1	0.9	0.8	1	-	
JUUL	No		10608	8569	18057	29856	4687	1651	8508	11960	11043	5091	1093	16643	5163	11290	17575	10434	10651	34510	4875	30707	16575	8715	18580	406	9292	4045	10472	29622		
hr 1	Rel diff	std (%)	4.55	4.54	4.86	4.74	3.22	4.55	3.87	3.53	3.04	5.16	6.86	5.58	3.63	7.36	3.65	5.43	3.47	4.82	5.1	8.23	5.05	4.88	5.72	4.47	6.7	4.58	17.14	4.2	5.32	
VCO 501	Rel diff	bias (%)	3.62	5.78	3.42	3.75	1.94	2.99	5.74	1.06	2.8	5.65	1.03	2.51	3.06	11.41	-2.38	3.24	2.56	2.39	-1.28	0.21	2.37	1.49	0.65	-3.07	1.19	-1.73	-2.05	-1.99	2.01	
			0.95	0.97	0.96	0.94	0.97	0.92	0.95	0.93	0.97	0.9	0.82	0.94	0.96	0.95	0.95	0.93	0.92	0.93	0.89	0.83	0.96	0.87	0.94	0.63	0.9	0.95	0.82	0.97	0.92	
Ē	Std	100	0.8	0.8	0.9	0.9	0.9	-	0.9	-	0.9	0.9	1.1	-	-	0.9	-	0.9	-	0.9	1.2	0.8	0.9	-	0.9	1.2	0.9	0.9	0.8	1	0.0	
0 km 1 hr	Rel diff	std (%)	4.15	4.45	4.7	5.16	3.39	4.54	3.14	3.64	2.91	4.45	7.58	6.82	3.72	7.35	4.1	4.5	3.79	4.93	5.18	8.14	4.21	5.17	4.31	5.01	6.49	4.87	19.37	3.87	5.36	
CO emooth 6	Rel diff	bias (%)	12.44	13.1	11.88	12.53	9.89	9.87	6.61	90.6	11.46	10.87	15.16	11.3	10.73	-0.73	5.91	9.94	10.75	10.39	3.36	3.29	7.5	12.87	7.47	6.83	6.3	11.22	8.8	9.3	9.22	
N etd Y	Corr		0.95	0.97	0.96	0.94	0.97	0.93	0.97	0.94	0.97	0.93	0.81	0.95	0.97	0.95	0.95	0.96	0.92	0.94	0.89	0.85	0.98	0.9	0.97	0.63	0.92	0.96	0.82	0.98	0.93	
	Std V		0.8	0.8	0.9	0.9	0.9	0.9	0.9	1	0.9	0.9	1.1	0.9	-	0.8	0.9	0.9	0.8	0.9	1.2	0.7	1	0.9	0.8	1.1	-	0.9	0.8	0.9	0.0	
4 1 1 1	штш Веј diff	std (%)	4.2	4.4	4.92	5.18	3.24	4.65	2.94	3.59	2.97	4.58	7.69	6.48	3.77	6.9	4.22	4.54	3.66	4.71	5.02	7.69	4	4.91	4.07	4.66	6.13	4.71	17.93	3.99	5.21	
mooth 50 b	Rel diff	bias (%)	6.55	7.54	5.79	5.92	2.88	3.51	0.01	2.27	4.53	4.16	8.27	4.31	3.89	-6.02	-0.94	3.05	3.44	3.1	-3.3	-3.41	0.43	5.14	0.24	-0.62	-0.66	3.46	2.02	2.9	2.45	
			0.95	0.97	0.95	0.94	0.97	0.92	0.97	0.93	0.97	0.92	0.8	0.94	0.97	0.95	0.94	0.95	0.91	0.94	0.89	0.85	0.98	0.89	0.97	0.63	0.92	0.96	0.82	0.97	0.92	
NOC	Std	nic	0.8	0.9	0.9	-	0.9	-	-	-	0.9	-	1.2	-	1.1	0.9	-	1	0.9	0.9	1.3	0.8	1	-	0.9	1.2	-	0.9	0.8	1	-	
	No		11114	9495	19439	31198	4698	1671	8593	12139	11147	8325	1887	16796	5164	11349	17655	10467	10668	34554	4951	30819	16583	8715	18581	406	9292	4045	10479	29781		
	Iat	Lat	80	78.9	67.4	54.3	53.2	53.1	49.1	48.8	48	47.5	47.4	45.9	43.5	39.8	36.6	36	35.1	35	34.2	34.1	33.2	28.3	18.5	-7.9	-12.5	-20.9	-34.4	-45		
	Site		EUREKA	NY-ÅLESUND	SODANKYLÄ	EAST TROUT LAKE	BIAŁYSTOK	BREMEN	KARLSRUHE	PARIS	ORLÉANS	GARMISCH	ZUGSPITZE	PARK FALLS	RIKUBETSU	XIANGHE	LAMONT	TSUKUBA	NICOSIA	EDWARDS	JPL	PASADENA	SAGA	IZAÑA	BURGOS	ASCENSION	DARWIN	RÉUNION	MOLLONGONG	LAUDER	Mean all stations	

((SAT-GB)/GB) bias in perce	ent and cc	olumn title	e 'Rel di	ff std' rej	presents the st	andard devi	ation of	f the rela	tive bias in pe	ercent.									
			NDACC	CO smo	oth 50 km 3	hr		NDAC	C CO 50 km	3 hr		NDACC	CO ap 50 kn	ı 3 hr	Z	DACC C	O smooth	t cone 50 km	3 hr
Site	Lat	No.	Std	Corr	Rel diff	Rel diff	Std	Corr	Rel diff	Rel diff	Std	Corr	Rel diff	Rel diff	No.	Std	Corr	Rel diff	Rel diff
					bias (%)	std (%)			bias (%)	std (%)			bias (%)	std (%)				bias (%)	std (%)
EUREKA	80.1	714	0.8	0.95	12.96	4.56	0.8	0.96	11.64	3.72	0.8	0.96	11.65	3.79	636	0.8	0.95	12.18	4.83
NY-ÅLESUND	78.9	90	0.9	0.97	11.43	4.26	0.8	0.98	8.12	3.73	0.9	0.98	1.78	3.37	89	0.9	0.96	11.77	4.33
THULE	76.5	2667	0.9	0.95	9.44	4.79	0.9	0.96	7.67	3.67	0.9	0.95	4.28	4.04	2516	0.9	0.95	9.02	5.06
KIRUNA	67.8	622	0.8	0.95	3.38	4.7	0.9	0.97	4.32	3.63	0.9	0.97	2.36	3.63	604	0.8	0.95	3.47	S
HARESTUA	60.2	216	0.9	0.97	6.7	3.73	0.9	0.97	4.8	3.22	0.9	0.97	4.58	3.22	192	0.9	0.96	7.23	4.26
ST.PETERSBURG	59.9	843	0.9	0.96	6.67	3.88	0.9	0.95	5.08	3.74	0.9	0.95	4	3.75	842	0.9	0.95	6.53	3.95
BREMEN	53.1	257	0.9	0.95	4.88	3.85	0.9	0.95	5.93	3.81	0.9	0.95	4.1	3.78	251	0.9	0.95	4.93	3.8
KARLSRUHE	49.1	933	1	0.96	-0.55	3.24	0.9	0.96	6.34	3.47	-	0.97	-0.14	3.07	913	-	0.96	-0.41	3.46
GARMISCH	47.5	2181	1	0.93	3.99	4.79	1	0.91	5.93	5.26	1.1	0.92	-5.33	4.32	2058	-	0.91	3.83	5.46
ZUGSPITZE	47.4	1773	-	0.91	6.16	5.16	1.1	0.9	7.62	5.49	1.1	0.9	<i>T.</i> 77	5.38	1675	-	0.9	6.91	5.8
JUNGFRAUJOCH	46.6	384	1	0.94	8.09	4.43	1	0.92	11.34	4.87	1	0.91	10.92	4.97	377	1	0.92	8.79	4.93
TORONTO	43.6	935	1	0.9	11.82	7.11	0.9	0.92	9.73	5.24	0.9	0.92	7.5	5.15	894	1	0.89	11.99	7.48
RIKUBETSU	43.5	43	-	0.97	7.65	3.57	-	0.96	3.78	3.21	1	0.95	2.51	3.33	43	1	0.96	7.36	3.53
BOULDER.CO	40	610	0.8	0.73	8.32	11.89	0.9	0.76	3.62	9.87	1	0.76	-2.91	9.72	601	0.8	0.77	7.68	11.5
IZAÑA	28.3	657	0.9	0.92	2.5	4.22	-	0.9	3.64	4.34	1	0.9	3.59	4.33	656	0.9	0.91	2.48	4.35
MAUNA LOA	19.5	155	0.9	0.97	2.65	3.22	-	0.96	3.17	3.41	1	0.97	2.93	3.38	153	0.9	0.96	2.49	3.64
ALTZOMONI	19.1	364	0.7	0.62	20.6	10.73	0.8	0.56	20.32	11.2	0.8	0.56	20.35	11.2	358	0.6	0.63	20.26	11.79
PARAMARIBO	5.8	59	1.2	0.82	0.88	9	1.1	0.77	-1.12	5.64	1.1	0.76	-2.24	5.84	57	1.2	0.81	0.98	6.05
PORTO.VELHO	-8.8	168	1	0.97	60.9	8.13	-	0.96	1.02	8.42	1.1	0.96	0.95	8.36	164	-	0.96	6.33	8.5
LA.RÉUNION.MAÏDO	-21.1	1117	0.9	0.98	6.44	4.62	-	0.98	5.96	4.54	1	0.98	5.78	4.58	1055	0.9	0.97	69.9	5.53
MOLLONGONG	-34.4	1403	0.5	0.7	9.17	23.01	0.7	0.7	5.04	21.41	0.7	0.7	0.82	20.92	1382	0.5	0.71	8.84	21.88
LAUDER	-45	1805	0.9	0.96	7.82	4.54	0.9	0.97	5.93	4.27	0.9	0.97	1.08	3.99	1776	0.9	0.96	7.89	4.74
ARRIVAL.HEIGHTS	-77.8	174	0.8	0.95	18.3	5.63	0.8	0.96	15.68	4.93	0.9	0.95	11.99	5.34	148	0.8	0.93	19.65	7.11
Mean all stations			0.9	0.91	7.62	6.09	0.9	0.91	6.76	5.7	0.9	0.9	4.27	5.63		0.9	0.91	7.69	6.39
Std all stations	_		0.13	0.1	5.04	4.31	0.1	0.11	4.65	4.02	0.1	0.11	5.62	3.94		0.14	0.09	5.09	4.07

Table 7. S5P CO column validation results against NDACC CO column data at 23 stations for the period between November 2017 and September 2020. Spatial co-location with radius of 50 km or cone with 1° opening angle along the FTIR line-of-sight and time co-location of ± 3 hour around the satellite overpass was used. NDACC station (column 1) are sorted according to the decreasing latitude (column 2). The column with title 'No' represents the number of co-located measurements, column title 'Std' represents the standard deviation of the time series of the ground-based data relative to the standard deviation **Table 8.** Validation of S5P XCO ALL, CLSKY and CLOUD data with TCCON XCO data at 28 stations for the period between November 2017 and September 2020. Spatial co-location with radius of 50 km and time co-location of ± 1 hour around the satellite overpass was used. TCCON station (column 1) are sorted according to the decreasing latitude (column 2). The column with title 'No.' represents the number of co-located measurements, column title 'Std' represents the standard deviation of the time series of the ground-based data relative to the standard deviation of the time series of the S5P data, column title 'Corr' represents the correlation coefficient between the S5P and the reference ground-based data, column title 'Rel diff bias' represents the relative difference ((SAT-GB)/GB) bias in percent and column title 'Rel diff std' represents the standard deviation of the relative bias in percent.

		TCCO	N unsc	XCO sm	ooth 50 km 1	hr ALL	TCCON	unsc X	CO smoo	oth 50 km 1 l	r CLSKY	TCCON	unse X	CO smo	oth 50 km 1 h	r CLOUD
Sites	Lat	No.	Std	Corr	Rel diff	Rel diff	No.	Std	Corr	Rel diff	Rel diff	No.	Std	Corr	Rel diff	Rel diff
					bias (%)	std (%)				bias (%)	std (%)				bias (%)	std (%)
EUREKA	80	11114	0.8	0.95	6.55	4.2	9738	0.8	0.94	7.31	4.34	6124	0.9	0.92	4.91	3.9
NY-ÅLESUND	78.9	9495	0.9	0.97	7.54	4.4	7854	0.9	0.96	7.75	4.93	4637	0.9	0.97	6.12	4.43
SODANKYLÄ	67.4	19439	0.9	0.95	5.79	4.92	13242	0.9	0.94	6.44	5.56	8072	0.9	0.96	4.32	4.11
EAST TROUT LAKE	54.3	31198	1	0.94	5.92	5.18	18415	1	0.92	7.03	5.11	16283	0.9	0.91	4.79	6.26
BIAŁYSTOK	53.2	4698	0.9	0.97	2.88	3.24	1122	1	0.98	2.04	2.02	4110	0.9	0.97	3.04	3.41
BREMEN	53.1	1671	1	0.92	3.51	4.65	1028	1	0.96	4.82	3.65	1128	1	0.9	3.18	5.17
KARLSRUHE	49.1	8593	1	0.97	0.01	2.94	4089	0.9	0.96	0.22	3.22	6472	1	0.96	0.04	3.21
PARIS	48.8	12139	1	0.93	2.27	3.59	5703	1.1	0.92	3.11	3.8	8627	1	0.93	1.77	3.64
ORLÉANS	48	11147	0.9	0.97	4.53	2.97	5155	0.9	0.95	4.98	3.44	8710	0.9	0.97	4.45	3.08
GARMISCH	47.5	8325	1	0.92	4.16	4.58	3158	0.9	0.87	4.84	5.84	3609	0.9	0.93	3.07	4.36
ZUGSPITZE	47.4	1887	1.2	0.8	8.27	7.69	741	1.2	0.7	5.86	9.38	861	1.1	0.84	5.13	7.12
PARK FALLS	45.9	16796	1	0.94	4.31	6.48	9226	1	0.93	5.21	7.85	11031	0.9	0.94	3.35	4.1
RIKUBETSU	43.5	5164	1.1	0.97	3.89	3.77	2775	1	0.94	5.7	4.25	4398	1	0.97	3.15	3.53
XIANGHE	39.8	11349	0.9	0.95	-6.02	6.9	4610	0.9	0.93	-4.05	8.52	7910	0.9	0.95	-6.79	6.79
LAMONT	36.6	17655	1	0.94	-0.94	4.22	6513	0.9	0.88	-0.4	4.79	15225	1	0.94	-1.15	4.27
TSUKUBA	36	10467	1	0.95	3.05	4.54	5651	0.9	0.95	5.04	4.2	7887	1	0.95	2.03	4.5
NICOSIA	35.1	10668	0.9	0.91	3.44	3.66	8320	0.8	0.87	4.38	4.78	9626	0.9	0.91	3.3	3.63
EDWARDS	35	34554	0.9	0.94	3.1	4.71	17050	0.9	0.94	2.32	4.74	32387	0.9	0.93	3.5	5.09
JPL	34.2	4951	1.3	0.89	-3.3	5.02	2626	1.2	0.84	-2.9	5.75	3897	1.2	0.86	-4.32	5.14
PASADENA	34.1	30819	0.8	0.85	-3.41	7.69	19464	1	0.84	-3.71	6.72	26508	0.5	0.58	-3.2	19.09
SAGA	33.2	16583	1	0.98	0.43	4	7581	1	0.95	1.36	4.89	14660	1	0.97	0.63	4.27
IZAÑA	28.3	8715	1	0.89	5.14	4.91	3633	1	0.82	5.82	6.64	7997	1	0.89	5.02	4.96
BURGOS	18.5	18581	0.9	0.97	0.24	4.07	8442	0.9	0.94	0.69	4.67	17951	0.9	0.97	0.09	4.17
ASCENSION	-7.9	406	1.2	0.63	-0.62	4.66	126	2	0.88	-2.83	4.34	383	1.1	0.59	-1.54	4.99
DARWIN	-12.5	9292	1	0.92	-0.66	6.13	3960	1.1	0.91	0.08	7.24	8361	1	0.93	-0.76	5.83
RÉUNION	-20.9	4045	0.9	0.96	3.46	4.71	1055	0.9	0.9	3.13	7.53	4045	0.9	0.95	3.59	4.9
WOLLONGONG	-34.4	10479	0.8	0.82	2.02	17.93	7428	1	0.89	1.41	9.14	7430	0.8	0.77	2.52	22.37
LAUDER	-45	29781	1	0.97	2.9	3.99	19658	1	0.96	3.46	4.32	22787	1	0.97	2.71	4.25
Mean all stations			1	0.92	2.45	5.21		1	0.91	2.83	5.42		0.9	0.9	1.89	5.73
Std all stations			0.1	0.07	3.38	2.79		0.2	0.06	3.43	1.84		0.1	0.1	3.11	4.37

Table 9. Validation of S5P CO column ALL, CLSKY and CLOUD data with NDACC CO column data at 23 stations for the period between November 2017 and September 2020. Spatial co-location with radius of 50 km and time co-location of \pm 3 hour around the satellite overpass was used. NDACC station (column 1) are sorted according to the decreasing latitude (column 2). The column with title 'No.' represents the number of co-located measurements, column title 'Std' represents the standard deviation of the time series of the ground-based data relative to the standard deviation of the time series of the S5P data, column title 'Corr' represents the correlation coefficient between the S5P and the reference ground-based data, column title 'Rel diff bias' represents the relative difference ((SAT-GB)/GB) bias in percent and column title 'Rel diff std' represents the standard deviation of the relative bias in percent.

		NI	DACC C	CO smoot	h 50 km 3 hi	ALL	ND	ACC CC) smooth	50 km 3 hr (CLSKY	NDA	ACC CC) smooth	50 km 1 hr 0	CLOUD
Sites	Lat	No.	Std	Corr	Rel diff	Rel diff	No.	Std	Corr	Rel diff	Rel diff	No.	Std	Corr	Rel diff	Rel diff
					bias (%)	std (%)				bias (%)	std (%)				bias (%)	std (%)
EUREKA	80.1	714	0.8	0.95	12.96	4.56	597	0.8	0.95	12.23	4.08	300	0.8	0.96	11.72	4.64
NY-ÅLESUND	78.9	90	0.9	0.97	11.43	4.26	89	0.8	0.96	10.7	4.65	71	0.9	0.97	11.72	4.1
THULE	76.5	2667	0.9	0.95	9.44	4.79	2388	0.9	0.95	9.43	4.67	1609	0.9	0.94	7.81	5.14
KIRUNA	67.8	622	0.8	0.95	3.38	4.7	534	0.8	0.94	4.06	4.82	441	0.8	0.94	2.39	4.94
HARESTUA	60.2	216	0.9	0.97	6.7	3.73	159	0.9	0.95	6.48	4.45	126	1	0.97	6.81	3.69
ST.PETERSBURG	59.9	843	0.9	0.96	6.67	3.88	741	0.9	0.94	6.41	4.31	654	0.9	0.94	6.76	4.52
BREMEN	53.1	257	0.9	0.95	4.88	3.85	171	0.9	0.93	5.57	4.45	168	0.9	0.96	5.12	3.79
KARLSRUHE	49.1	933	1	0.96	-0.55	3.24	506	1	0.96	0.19	3.27	795	1	0.96	-0.8	3.42
GARMISCH	47.5	2181	1	0.93	3.99	4.79	1215	1	0.92	4.08	5.01	1773	1	0.94	3.93	5
ZUGSPITZE	47.4	1773	1	0.91	6.16	5.16	1197	1	0.9	5.93	5.72	1326	1	0.94	6.38	5.31
JUNGFRAUJOCH	46.6	384	1	0.94	8.09	4.43	306	1	0.92	7	4.92	310	0.9	0.95	9.01	4.79
TORONTO	43.6	935	1	0.9	11.82	7.11	400	1	0.9	10.27	6.32	868	1	0.89	12.47	7.74
RIKUBETSU	43.5	43	1	0.97	7.65	3.57	31	1	0.95	6.98	3.63	37	1.1	0.96	8.82	4.61
BOULDER.CO	40	610	0.8	0.73	8.32	11.89	323	0.9	0.47	9.78	15.72	563	0.7	0.66	9.14	16.32
IZAÑA	28.3	657	0.9	0.92	2.5	4.22	276	1	0.85	2.67	5.78	613	0.9	0.93	2.53	4.14
MAUNA LOA	19.5	155	0.9	0.97	2.65	3.22	81	0.9	0.88	2.1	6.11	145	0.9	0.96	2.51	3.64
ALTZOMONI	19.1	364	0.7	0.62	20.6	10.73	212	0.7	0.73	22.05	13.44	338	0.8	0.65	19.51	10.3
PARAMARIBO	5.8	59	1.2	0.82	0.88	6						59	1.2	0.82	1.01	5.98
PORTO.VELHO	-8.8	168	1	0.97	6.09	8.13	64	1	0.93	3.64	10.5	161	1	0.96	7.82	9.52
LA.RÉUNION.MAÏDO	-21.1	1117	0.9	0.98	6.44	4.62	265	0.9	0.96	7.92	6.61	1011	0.9	0.98	6.57	4.64
WOLLONGONG	-34.4	1403	0.5	0.7	9.17	23.01	1007	0.9	0.89	6.49	8.01	1050	0.5	0.7	10.7	26.87
LAUDER	-45	1805	0.9	0.96	7.82	4.54	1132	0.9	0.96	7.91	4.58	1501	0.9	0.97	7.82	4.81
ARRIVAL.HEIGHTS	-77.8	174	0.8	0.95	18.3	5.63	164	0.8	0.95	17.59	5.2	111	0.9	0.94	18.24	5.95
Mean all stations			0.9	0.91	7.62	6.09		0.9	0.9	7.7	6.19		0.9	0.91	7.74	6.69
Std all stations			0.1	0.1	5.04	4.31		0.1	0.11	4.96	3.15		0.1	0.1	4.97	5.26